CLAYTON UTZ

Execution Version

Facility agreement

LM Investment Management Limited in its capacity as responsible entity for the LM First Mortgage Income Fund Borrower

Deutsche Bank AG, Sydney Branch Financier

Clayton Utz Lawyers Level 28, Riparian Plaza 71 Eagle Street Brisbane QLD 4000 Australia GPO Box 55 Brisbane QLD 4001 T +61 7 3292 7000 F +61 7 3221 9669

www.claytonutz.com Our reference 17306/17011/80107994

Legal\301465383.15

Contents

1.	Definitions and Interpretation			
	1.1 1.2 1.3 1.4 1.5	Definitions Interpretation. Capacity of Responsible Entity Limitation of Responsible Entity's Liabilities Custodian assets and obligations	19 20	
2.	Facility		21	
	2.1 2.2 2.3 2.4 2.5 2.6	Facility	21 22 22 22	
3.	Conditio	ns precedent		
	3.1	Conditions precedent to the Advance	22	
4.	Advance		22	
	4.1	Notice	22	
	4.2	Contents of Utilisation Notice		
	4.3 4.4	Requirements of Utilisation Notice	ZJ 23	
	4.5	Maximum number of Advance		
5.	Cancellation and reduction of Commitment			
	5.1	Commitment	23	
	5.2	Voluntary cancellation or reduction	23	
	5.3	Reduction consequent on repayment or prepayment		
	5.4	Limitations		
6.		ent and prepayments		
	6.1	Voluntary prepayment.	24	
	6.2 6.3	Mandatory prepayments Disposal of LM Mortgaged Properties	24	
	6.4	Order of application	25	
	6.5	Date for prepayment	25	
	6.6	General provisions relating to repayment and prepayment	26	
7.	Interest.		26	
	7.1	Interest Periods	26	
	7.2	Calculation of interest		
	7.3 7.4	Payment of interest	26 28	
8.		on overdue amounts		
0.		Payment of interest		
	8.1 8.2	Accrual of interest	27	
9.	Fees			
	9.1	Establishment fee	28	
	9.2	Commitment fee		
10.	Payment	ts	28	

i.

	10.1 10.2	Time and place	
	10.2	Currency for payments	
	10.4	Payments in gross	
	10.5	Payment on a Business Day	
11.			
11.		Ţ,	
	11.1	No deduction for Taxes	
	11.2	Payment net of Taxes	
	11.3	Tax saving	30
12.	lliegality	***************************************	31
13.	Increase	ed cost	31
	13.1	Obligation to indemnify	
	13.1	Exceptions	31
14.	Mitigatio)n	
	14.1	Mitigation	32
	14.2	Costs and expenses	32
4=	Denzee	ntations and warranties	20
15.	Represe	· · · · · · · · · · · · · · · · · · ·	
	15.1	General representations and warranties	32
	15.2	Information representations and warranties	34
	15.3	Corporate representations and warrantles	34
	15.4	Responsible Entity representations and warranties	35
	15.5	Specific Portfolio Representations	36
	15.6	Representations and warranties repeated	38
16.	Underta	kings	
	16.1	Duration and benefit	39
	16.2	Information and Custody Audits	ļ39
	16.3	Restrictions on Disposals	42
	16.4	Liabilities	42
	16.5	Financial Accommodation	42
	16.6	Environmental matters	ļ43
	16.7	Insurance	ļ .44
	16.8	Authorisations	ļ46
	16.9	Security Property	46
	16.10	Cash management and application	47
	16.11	Operation of Managed Cash Account	47
	16.12	Receivership of Customers	<u>!</u> 49
	16.13	Custodian Arrangements	49
	16.14	General undertakings	48
	16.15	LM Mortgage Trust undertakings	51
	16.16	LM Loan Document undertakings	
	16.17	Financier's reputation	54
	16.18	Customer Payments	05
	16.19	LM Mortgaged Property Taxes	1
17.	Default	and termination	
	17.1	Events of Default	.j5t
	17.2	Financier's rights on Event of Default	
18.	Indemn	ities	
40		1 1441p4894919191914144114494491191911914114494949194914494949444444	i
19.	26f-CII		

20.	Costs, stamp duties and GST6			
	20.1	Costs	6	
	20.2	Stamp duties	6	
	20.3	GST	61	
21.	Assignments and confidentiality			
	21.1	Successors and assigns		
	21.2	Assignments by the Borrower	62	
	21,3	Assignments by Financier	62	
	21.4	No deductions and withholdings or increased costs in certain circumstances:	-	
	21.5	Confidentiality		
22.	Governi	ng law	64	
	22.1	Governing law	64	
23.	Arbitrati	on	64	
	23.1	Notice of Dispute		
	23.1			
	23.2 23.3	Negotiations	64	
	23.4	Reference to arbitration		
		General principles		
	23.5 23.6	Expedited proceedings		
	23.7	Consolidation		
	23.8	Joinder		
	23.9	Award final and binding		
		Effect on Powers		
24.	Miscella	neous	6t	
	24.1	Certificate of Financier	66	
	24.2	Notices		
	24.3	Continuing obligation		
	24.4	Settlement conditional		
	24.5	Further acts and documents		
	24.6	Attorney	69	
	24.7	Severance		
	24.8	Powers cumulative		
	24.9	Waiver	70	
	24.10	Consents	70	
	24.11	Indemnities	70	
	24.12	Time of the essence	70	
	24.13	Moratorium legislation		
	24.14	Binding on each signatory	71	
	24.15	Counterparts		
	24.16	Entire agreement	71	
25.	No repre	sentation by or reliance	71	
Calbanded	•	umentary conditions precedent		
		•		
Schedul	e 2 - Forn	n of Utilisation Notice and Selection Notice	74	
Schodul	-3-IM1	oan Schedule	7 <i>!</i>	
Schedul	e 4 - Rele	ase Prices	7€	
Schedul	e 5 - New	Lending Criteria	77	

Schedule 6 Financier Authorised Officers	78
Schedule 7 - Exceptions to LM Documentation Assumptions	79
Schedule 8 - Verification Certificate	81
Schedule 9 - Trust Bank Accounts	83
Schedule 10 - Managed Cash Accounts	

Facility agreement dated

Parties

LM Investment Management Limited ACN 077 208 461 in its capacity as Responsible Entity for the LM First Mortgage Income Fund (Borrower)

Deutsche Bank AG, Sydney Branch (Financier)

Background

At the request of the Borrower, the Financier has agreed to provide financial accommodation to the Borrower in accordance with the Finance Documents.

Operative provisions

1. Definitions and interpretation

1.1 Definitions

In this agreement:

Acceptable Bank means a financial institution who is an authorised deposit taking institution under the Banking Act 1959 carrying on business in Australia which has assigned to it a short term credit rating equal to or higher than "A" by Standard & Poors.

Accepted Accounting Practices means the accounting practices and standards generally accepted in Australia from time to time.

Accounts means a balance sheet, an income statement, a statement of cash flow and, if for a Financial Year and required by law, a statement of changes in equity for the year, together with any statements, reports (including any directors', Responsible Entity and auditors' reports) and notes attached to or intended to be read with any of them.

Advance means any loan made available under the Facility or, where the context requires, the principal amount of that loan or its funding.

Adversely Affected LM Loans means, at any time, those LM Loans (other than those permitted under clause 16.5) in respect of which any representation and warranty in clause 15.5 is incorrect.

Affiliate means:

- (a) with respect to the Borrower, any person who controls (whether directly or indirectly) or is controlled by (whether directly or indirectly) the Borrower; and
- (b) any other trust or managed investment scheme of which LM Investment is the trustee or responsible entity.

Alternative Cover Amount means at any time the aggregate of:

- (a) an amount equal to interest on the Advance calculated in accordance with clause 7.1 and 7.2 in respect of a period of 12 months following the date of calculation; and
- (b) the Projected Principal Amortisation Amount for the applicable Calculation Period less any amount prepaid by the Borrower in respect of the Projected Principal Amortisation Amount for that Calculation Period.

Legal\301465383.15

Alternative Minimum Cash Balance means the greater of:

- (a) \$6 million Cash or Cash Equivalent Investments; or
- (b) Cash or Cash Equivalent Investments equal to the Alternative Cover Amount.

Approved Auditors means PricewaterhouseCoopers, KPMG, Emst & Young or Deloitte Touche Tohmatsu.

Auditors means Ernst & Young.

Authorisation means:

- (a) any authorisation, permit, consent, approval, resolution, licence, exemption, permission, recording, filing or registration required by any Government Authority or any law; and
- (b) in relation to any act, matter or thing which will be prohibited or restricted either in whole or in part by any law if a Government Authority intervenes or acts in any way within a specified period after lodgement, filing, registration, notification or any other event, the expiry of that period without intervention or action.

Authorised Officer means:

- (a) in relation to the Borrower, a person appointed as an authorised officer of the Borrower for the purposes of the Finance Documents by a resolution of the board of directors of the Borrower and in respect of whom the Financier has received a certificate signed by a director of the Borrower:
 - (i) setting out that person's name, position and signature; and
 - (ii) confirming the appointment,

provided the Financier has not received notice of revocation of that appointment; and

(b) in relation to the Financier, any person who is listed in Schedule 6 (or any updated Schedule 6 provided by the Financier to the Borrower from time to time) as being an Authorised Officer of the Financier.

Availability Period means the period from and including the date of this agreement to the close of business on the third Business Day following execution of this agreement, or any later date agreed in writing between the Financier and the Borrower on or after the date of this agreement.

BBSW for any period means the average bid rate displayed at or about 10.30am (Sydney time) on page BBSW on the Reuters Monitor System on the first day of that period for a tenor equal to (or approximately equal to) that period.

Biannual Interest Payment means in respect of any Biannual Interest Payment Date, the amount of interest accrued on the Advance from the proceeding Biannual Interest Payment Date less any interest paid by the Borrower to the Financier for that period.

Biannual Interest Payment Date means 31 December 2010 and the last day of each and every successive 6 month period following 31 December 2010, provided the last Biannual Interest Payment Date shall end on the Termination Date.

Bill has the meaning given to "bill of exchange" in the Bills of Exchange Act 1909 (Cth), but does not include a cheque or payment order and any reference to the drawing, acceptance, endorsement or other dealing of or with a Bill refers to a drawing, acceptance, endorsement or other dealing within the meaning of that Act.

Binding Requirement means:

- (a) any law; or
- (b) any administrative guideline, directive, request or policy of any Government Authority whether or not having the force of law and, if not having the force of law, the observance of which is in accordance with the practice of responsible banks or financial institutions.

Break Costs means the loss or liability, if any, incurred by the Financier (as determined by it in its sole opinion), because of:

- (a) the liquidation or re-deployment of deposits or other funds acquired or contracted for by the Financier to fund or maintain any Advance or Unpaid Sum; or
- (b) the reversing or termination of any agreement or arrangement entered into by the Financier to hedge, fix or limit its effective cost of funding or maintaining any Advance or Unpaid Sum,

to the extent that loss or liability arises as a result of the prepayment of that Advance or Unpaid Sum on any date other than the last day of an Interest Period for that amount. The amount of Break Costs will be the amount payable in lieu of the interest which would otherwise have been paid if no prepayment had occurred.

Business Day means a day (other than a Saturday or Sunday) on which banks are open for business generally in Brisbane and Sydney.

Calculation Period means a period of 6 months from the date the Alternative Cover Amount is calculated.

Cash means, at any time, cash denominated in Australian dollars in hand or at bank and (in the latter case) credited to an account in the name of the Borrower or the Custodian with an Acceptable Bank and for so long as:

- (a) that cash is repayable on demand within 90 days after the relevant date of calculation;
- (b) repayment of that cash is not contingent on the prior discharge of any other indebtedness of the Borrower or Custodian or of any other person whatsoever or on the satisfaction of any other condition;
- (c) there is no Encumbrance over that cash except for the Encumbrance granted in favour of the Financier; and
- (d) the cash is freely and (except as mentioned in paragraph (a) above) immediately available to be applied in repayment or prepayment of the Facilities.

Cash Equivalent Investment means:

(a) bonds, debentures, stock, treasury bills, notes or any other security issued or guaranteed by the Commonwealth of Australia or any government of any State or Territory of the Commonwealth of Australia;

- (b) deposits or certificates of deposit or short term securities which have not more than 90 days to final maturity and are not convertible into any other form of security with, or guaranteed by an entity which has, a short term credit rating of at least Prime-2 from Moody's Investor Services Inc. or A from Standard & Poor's Rating Services; and
- (c) any other investment approved from time to time by the Financier (which approval may be withdrawn at any time provided that any individual investment then held by the Borrower or the Custodian will continue to be a Cash Equivalent whilst it is so held).

CBA means Commonwealth Bank of Australia ABN 48 123 123 124.

CBA Loan means all of the Borrower's Financial Liability to CBA.

Certificate of Title in relation to a LM Mortgaged Property means the Certificate of Title or other documents evidencing title to that LM Mortgaged Property (including, if applicable, the documents forming any abstract of title) or where the certificate of title or other documents have been cancelled due to the computerisation of the register, any original registration confirmation notification or statement which the Custodian has in its files.

Change means:

- (a) a change to any Binding Requirement;
- (b) the introduction of a new Binding Requirement; or
- (c) a change in the interpretation or administration of a Binding Requirement,

in any case occurring after the date of this agreement.

Charge means the document titled "Fixed & Floating Charge" dated on or about the date of this agreement granted by Borrower and Custodian in favour of the Lender.

Commitment means \$90,000,000, to the extent not cancelled or reduced in accordance with the terms of this agreement.

Compliance Plan means the compliance plan of the LM Mortgage Trust which has been lodged with ASIC and complies with section 601HA of the Corporations Act.

Constitution means, in relation to any person, the replaceable rules, constitution or combination of both (as those terms are used in section 134 of the Corporations Act) of that person (if any).

Contamination means the presence in, on or under any land, air or water of a substance (whether a solid, liquid, gas, odour, heat, sound, vibration or radiation) at a concentration above the concentration at which the substance is normally present in, on or under (respectively) land, air or water in the same locality, being a presence that presents a risk of harm to human health or any other aspect of the Environment.

Controller has the meaning given in section 9 of the Corporations Act.

Corporations Act means the Corporations Act 2001 (Cth).

Cost means any cost, expense, charge, liability or disbursement.

Credit Legislation shall mean and include each of the following enactments (including regulations and proclamations pursuant thereto) as may be applicable:

- (a) the National Consumer Credit Protection Act 2009 (Cth); the National Credit Code (Cth);
- (b) the Credit Act 1987 (Queensland); the Credit Act 1984 (New South Wales); the Credit Act 1984 (Victoria); the Credit Act 1984 (Western Australia); the Credit Ordinance 1985 (Australian Capital Territory);
- (c) Consumer Credit (New South Wales) Code;
 Consumer Credit (Victoria) Code;
 Consumer Credit (Western Australia) Code;
 Consumer Credit (Queensland) Code;
 Consumer Credit (Tasmania) Code;
 Consumer Credit (South Australia) Code;
 Consumer Credit (Australian Capital Territory) Code; and
 Consumer Credit (Northern Territory) Code;
- (d) any corresponding Act or Ordinance of another State or Territory of Australia
 which is a recognised State for the purposes of the Credit Act 1987 (Queensland)
 and/or any of the above enactments; and
- (e) any other Act or Ordinance for the regulation of the provision of credit of another State or Territory of Australia which is substantially similar in the opinion of the Mortgagee to the Credit Act 1987 (Queensland) or Consumer Credit Code.

Custodian means the Trust Company (PTAL) Limited ACN 008 412 913 (formerly named Permanent Trustee Australia Limited) or any successor approved by the Financier.

Custody Agreement means the agreement dated 4 February 1999 between the Custodian and the Borrower as amended and restated on 14 June 1999, 20 May 1999, 24 May 2000, 18 March 2002, 19 November 2002 and 1 September 2004.

Customer in relation to an LM Loan means the person or persons to whom the loan or other financial accommodation has been provided under that LM Loan including any person whom has provided LM Collateral Security in respect of the LM Loan (however arising).

Customer Information means any personal information in relation to a Customer that is provided to the Custodian or the Borrower in connection with an LM Loan.

Default Margin means initially 3% per annum or, where the Borrower has been in default of any payment obligations under this agreement for six (6) consecutive months, thereafter 5% per annum and in the case of the increased rate such rate shall apply from the Interest Period commencing immediately after such 6th month.

Deferred Release Price Percentage means 75% or such other percentage agreed to between the Borrower and the Financier from time to time following a Deferred Release Price Review Event.

Deferred Release Price Review Event occurs if on any day the aggregate credit balance of the Trust Bank Accounts is not less than an amount calculated in accordance with the definition of the Alternative Cover Amount on that day except that the Calculation Period is 12 months from the date that the Alternative Cover Amount is calculated.

Delegate means any officer, employee, agent or attorney of the Financier.

Disclosed Litigation means Supreme Court of New South Wales case number 2009/298727.

Legal\301465383.15 5

Disposal means any sale, assignment, transfer, lease or other disposal, or grant of any interest whether:

- (a) in a single transaction or in a series of transactions (whether related or not); or
- (b) voluntarily or involuntarily.

Distribution means:

- (a) any dividend, charge, interest, fee, payment or other distribution (whether cash or assets) or redemption, repurchase, defeasance, retirement or repayment on or in respect of any units in the LM Mortgage Trust; or
- (b) any interest payment, any repayment or prepayment of any amount of principal or any other payment in respect of any liability of the Borrower to any holder of a unit in the LM Mortgage Trust (or to any Related Body Corporate of any holder of a unit in the LM Mortgage Trust) at any time in connection with any Financial Liability.

dollar or \$ means the lawful currency of the Commonwealth of Australia.

Due Diligence Report means the due diligence report prepared by the Financier's solicitors, Clayton Utz, dated on or about the date of this agreement.

Encumbrance means a mortgage, charge, pledge, lien, encumbrance, security interest, title retention, preferential right, trust arrangement, contractual right of set-off or any other security agreement or arrangement in favour of any person.

Environment means components of the earth, including:

- (a) land, air and water;
- (b) any layer of the atmosphere;
- (c) any organic or inorganic matter and any living organism; and
- (d) any human made or modified structure or area,

and includes interacting natural ecosystems that include components referred to in paragraphs (a) to (d) of this definition.

Environmental Law means any law relating to:

- (a) the Environment (including any law relating to land use, planning, environmental assessment, pollution, Contamination, chemicals, waste, the use or presence of asbestos or dangerous goods or hazardous substances, building regulation, the occupation of buildings, heritage, species, flora and fauna or noise); or
- (b) any aspect of protection of the Environment.

Event of Default has the meaning given in clause 17.1.

Event of Insolvency means:

- (a) a controller (as defined in section 9 of the Corporations Act), administrator or similar officer is appointed in respect of a person or any asset of a person;
- (b) a liquidator or provisional liquidator is appointed in respect of a person;

- (c) any application (not withdrawn or dismissed within 7 days (or such longer period agreed to by the Financier) is made to a court for an order, an order is made, a meeting is convened or a resolution is passed, for the purpose of:
 - (i) appointing a person referred to in paragraph (a) or (b) of this definition;
 - (ii) winding up or deregistering a person; or
 - (iii) proposing or implementing a scheme of arrangement, other than with the prior approval of the Financier under a solvent scheme of arrangement pursuant to Part 5.1 of the Corporations Act.

unless the Financier is satisfied that the application or the convening of the meeting is frivolous or vexatious;

- (d) any application (not withdrawn or dismissed within 7 days (or such longer period agreed to by the Financier) is made to a court for an order, a meeting is convened, a resolution is passed or any negotiations are commenced, for the purpose of implementing or agreeing:
 - (i) a moratorium of any debts of a person;
 - (ii) any other assignment, composition or arrangement (formal or informal) with a person's creditors; or
 - (iii) any similar proceeding or arrangement by which the assets of a person are subjected conditionally or unconditionally to the control of that person's creditors or a trustee,

or any agreement or other arrangement of the type referred to in this paragraph (d) is ordered, declared or agreed to unless the Financier is satisfied that the application or the convening of the meeting is frivolous or vexatious;

- (e) as a result of the operation of section 459F(1) of the Corporations Act, a person is taken to have failed to comply with a statutory demand (as defined in the Corporations Act);
- (f) any writ of execution, garnishee order, mareva injunction or similar order, attachment or other process is made, levied or issued against or in relation to any asset of a person that arises from an event which the Financier in its reasonable opinion considers material;
- (g) anything analogous to anything referred to in paragraphs (a) to (f) inclusive of this definition, or which has a substantially similar effect, occurs with respect to a person under any law; or
- (h) a person is, or admits in writing that it is, or is declared to be, or is taken under any applicable law to be (for any purpose), insolvent or unable to pay its debts.

Excluded Tax means any Tax imposed by any jurisdiction on the overall net income of the Financier as a consequence of the Financier being a resident of or organised or doing business in that jurisdiction other than any Tax:

(a) calculated on or by reference to the gross amount of any payment (without the allowance of any deduction) derived under any Finance Document by the Financier; or

- (b) imposed as a result of the Financier being considered a resident of or organised or doing business in that jurisdiction solely as a result of:
 - (i) it being a party to any Finance Document; or
 - (ii) any transaction contemplated by any Finance Document.

Existing Financial Accommodation means:

- (a) the debt due to the Borrower by LM Investment Management Limited in its capacity as trustee for the LM Managed Performance Fund in relation to the assignment of three loans pursuant to deeds of assignment dated on or about the 28 August 2008 (being \$29,527,890 as at the date of this agreement);
- (b) the management fees totalling \$6,000,000 (as at the date of this agreement) which have been prepaid by the Borrower to LM Administration Pty Ltd ACN 055 691 426; and
- (c) the debt due to the Borrower by LM Investment Management Limited in its capacity as Trustee for the LM Managed Performance Fund in relation to the assignment of management fees totalling \$1,752,926.15 (as at the date of this agreement);

provided that:

- (i) any amounts repaid or prepaid will not be re-advanced;
- (ii) any amount advanced is not increased;
- (iii) the Borrower is not obligated to advance any further moneys; and
- (iv) the Borrower will upon request provide the Financier with any information and records relating to the above financial accommodation.

Facility means the term loan facility made available under this agreement, as referred to in clause 2.1.

Finance Document means:

- (a) this agreement;
- (b) any Security;
- (c) the Securities Custody Deed;
- (d) the Service Provider Side Deed; or
- (e) any other document as agreed by the Borrower and the Financier to be, for the purposes of this agreement, a Finance Document.

Financial Close means the date on which the first Advance is made.

Financial Liability of a person means any liability or indebtedness (whether present or future, actual or contingent) of that person for or in respect of:

(a) money borrowed or raised and debit balances at banks or financial institutions;

- (b) its obligations as lessee under any lease which in accordance with Accepted Accounting Practices would be treated as a finance or a capital lease;
- (c) any indemnity obligation in respect of any guarantee, indemnity, bond or letter of credit or similar instrument issued by a bank or financial institution;
- (d) any guarantee, indemnity, letter of credit or similar assurance in respect of financial loss given in connection with any Financial Liability (as referred to in any other paragraph of this definition) of another person;
- (e) any amount payable in connection with the redemption of any redeemable preference share issued by that person;
- (f) amounts raised under or in connection with any Bill acceptance, endorsement or discounting arrangement;
- (g) amounts raised under or in connection with any bond, debenture, note, loan stock or similar instrument;
- (h) receivables sold or discounted (other than to the extent they are sold on a non-recourse basis);
- (i) any Treasury Transaction;
- (j) the acquisition cost of any asset to the extent payable before or after the time of acquisition or possession where the advance or deferred payment is arranged primarily as a method of raising finance or financing the acquisition of that asset; or
- (k) amounts raised under any other transaction or series of transactions having the commercial effect of a borrowing or raising of money.

Financial Year means any 12 month period ending on 30 June.

Good Industry Practice means the exercise of that degree of skill, care, prudence, foresight and operating practice which would reasonably and ordinarily be expected from a skilled and experienced owner and operator engaged in the same type of business as the Group or any Group Member under the same or similar circumstances.

Government Authority means any government or any governmental or semi-governmental entity, authority, agency, commission, corporation or body (including those constituted or formed under any Statute), local government authority, administrative or judicial body or tribunal or stock exchange having authority in the Commonwealth of Australia.

GST means any goods and services Tax, consumption Tax, value added Tax or any similar Tax.

Hedge Provider means Customs House Currency Exchange (Australia) Pty Limited and HiFX Pty Ltd or any person approved by the Financier.

Insolvency Provision means any law relating to insolvency, sequestration, liquidation, winding up or bankruptcy (including any law relating to the avoidance of conveyances in fraud of creditors or of preferences and any law under which a liquidator or trustee may set aside or avoid transactions) and any provision of any agreement, arrangement or scheme, formal or informal, relating to the administration of any of the assets of any person.

Insurance means any insurance required to be taken out or maintained by the Borrower to comply with the provisions of the Finance Documents.

Interest Period means, in relation to an Advance, each period determined in accordance with clause 7.1 and, in relation to an Unpaid Sum, each period determined in accordance with clause 8.2.

Interest Rate means:

- (a) for an Interest Period in relation to any amount, 15% per annum; or
- (b) if the Facility is extended pursuant to clause 2.5, 18% per annum.

Joint Venture means any form of joint venture, whether a company, unincorporated entity, undertaking, association, partnership or other similar entity.

know your client procedures means the identification checks which the Financier is required to undertake under any Binding Requirement to meet its obligations to identify its customers.

Land means:

- (a) land (including tenements and hereditaments corporeal and incorporeal and every estate and interest in it whether vested or contingent, freehold or crown leasehold, the term of which lease is expressed to expire not earlier than 5 years after the maturity of the relevant LM Mortgage, and whether at law and equity) wherever situated including any fixture to land; and
- (b) any parcel in any lot, common property and land comprising a parcel within the meaning of the Body Corporate and Community Management Act 1997 (Queensland) or any equivalent legislation in other Australian jurisdiction.

LM Collateral Security means in respect of a LM Loan:

- (a) any:
 - (i) Encumbrance; or
 - (ii) guarantee, indemnity or other assurance,

which secures or otherwise provides for the repayment or payment of the LM Loan but does not include the LM Mortgage relating to the LM Loan; and

(b) any mortgage insurance policy and any LM Insurance Policy (both present and future) in respect of the LM Mortgage relating to the LM Loan or the Land secured by that LM Mortgage.

LM Documentation Assumptions means:

- (a) the obligations which are expressed to be assumed by each party to an LM Loan Document are legal, valid, binding and enforceable obligations subject to applicable insolvency and other laws affecting creditors' rights generally;
- (b) the entry into and performance by any party of, and the transactions contemplated by, any LM Loan Document do not and will not conflict with:
 - (i) any law or regulation applicable to that party;
 - (ii) the constitutional documents or trust deed of that party;
 - (iii) any agreement or instrument binding upon the Borrower;

- (c) amounts payable under each LM Loan Agreement were advanced in and are payable only in Australian dollars;
- (d) each LM Loan:
 - is secured by a first ranking mortgage over Qualifying Land which is the only first ranking mortgage over the Qualifying Land;
 - (ii) has a variable rate of interest or a fixed rate of interest;
 - (iii) at the time the LM Loan was initially drawn down the Facility to
 Security Ratio (as defined in the LM Loan Agreement relating to the LM
 Loan) was complied with; and
 - (iv) satisfied the Borrower's lending policy at the time of origination;
- (e) in respect of each LM Mortgaged Property:
 - the Customer is the legal and beneficial owner of the Land subject to the LM Mortgage; and
 - (ii) on or before the LM Loan was drawn down the LM Mortgaged Property was valued by a qualified valuer;
- each LM Loan Document does not contravene any laws, rules, regulations or Authorisations;
- (g) all payments required to be made by a Customer are payable on the dates shown in the LM Loan Documents and are payable without any set-off or other deduction by the Customer;
- (h) from Financial Close, no LM Loan Document is subject to any Encumbrance other than the Encumbrance in favour of the Financier;
- each LM Loan was originated during the ordinary course of the Borrower's business;
- (j) each LM Loan Document is governed by the laws of an Australian jurisdiction;
- (k) unless otherwise permitted by the Securities Custody Deed, the Custodian holds duly executed originals of each LM Loan Document;
- each LM Loan Document, if required to be stamped, has been stamped with all applicable duty;
- (m) each LM Loan Agreement is not able to be, and will not become able to be, cancelled or avoided by a Customer due to any misrepresentation, breach of warranty or other act or omission by the Borrower or the Custodian;
- (n) each LM Loan Document contains all of the terms and arrangements between the Borrower and the other parties to that document in respect of matters covered by that document and there are no other documents or agreements which have the effect of varying or discharging any LM Loan Document;
- (o) no circumstances existed on the date of execution of any of the LM Collateral Security which would be held by a court of competent jurisdiction to constitute an undervalue transfer or otherwise as preferential, fraudulent or uncommercial transaction; and

(p) the Borrower is not obliged to provide any further financial accommodation to a Customer under the LM Loan (other than under the terms of a new LM Loan permitted under clause 16.5),

with the exception that the LM Documentation Assumptions do not extend to those matters disclosed by the Borrower in Schedule 7.

LM Insurance Policy means any insurance policy (whether present or future) other than a mortgage insurance policy, in which the Custodian or Borrower has an interest and which is in force from time to time in respect of the Land the subject of an LM Mortgage or LM Collateral Security.

LM Loan means any present or future loan advanced by the Custodian or the Borrower to a Customer pursuant to an LM Loan Agreement other than a loan in respect of a Released LM Mortgaged Property.

LM Loan Agreement means each present or future loan agreement entered into between the Custodian and a Customer or the Borrower and a Customer as initially set out in Schedule 3 and includes any agreement, schedule, terms and conditions, application, approval or other document relating to the provision of financial accommodation to a Customer in connection with that LM Loan.

LM Loan Documents means in relation to an LM Loan:

- (a) each LM Loan Agreement relating to the LM Loan;
- (b) the original or duplicate LM Mortgage documents in relation to the LM Loan (including any document evidencing any substituted or additional LM Mortgage);
- (c) the Certificate of Title, registration confirmation statement or other indication of title (if any) in respect of the Land the subject of the LM Mortgage in relation to the LM Loan;
- (d) each LM Insurance Policy (or certificate of currency for the LM Insurance Policy) held in respect of the LM Mortgage;
- (e) each deed of priority or its equivalent in writing entered into in connection with the LM Mortgage;
- (f) any LM Collateral Security; and
- (g) any amendment, variation or replacement of or to or any of the foregoing such documents.

LM Loan Payment means any payment or series of payments made within a three month period by a Customer in respect of an LM Loan which is not related to a Disposal of a LM Mortgaged Property on account of principal or interest which exceeds an amount equal to 3 months interest on that LM Loan.

LM Loan Schedule means the schedule attached as Schedule 3 setting out details in relation to each LM Loan and the LM Mortgaged Property as updated in accordance with clause 16.2(d).

LM Mortgage in relation to an LM Loan means each registered mortgage over Land situated in any state or territory of Australia that secures the repayment of the LM Loan and the payment of interest and all other monies in respect of the LM Loan notwithstanding that the terms of the mortgage may secure other liabilities.

LM Mortgaged Property in relation to an LM Mortgage means the Land and all other property mortgaged under the LM Mortgage.

LM Mortgage Trust means the registered managed investment scheme known as the LM First Mortgage Income Fund ARSN 089 343 288 constituted by the LM Mortgage Trust Constitution.

LM Mortgage Trust Constitution means the constitution establishing the LM First Mortgage Income Fund dated 18 September 1999 and as amended and/or replaced on 6 June 2005, 7 July 2006, 31 May 2007 and 10 April 2008.

Loan Files in relation to a LM Loan means:

- (a) any correspondence between the Customer and the Borrower or the Custodian in relation to a LM Loan Document; and
- (b) all Customer information and such books, records, paper and electronic files (whether originals or copies) relating to that LM Loan (other than the LM Loan Documents) which the Borrower, the Service Provider or their agents has in its custody.

Managed Cash Account means collectively the accounts maintained by the Custodian as set out in Schedule 10 and such other accounts as are approved by the Financier from time to time and excluding any such accounts as are permanently closed by the Custodian from time to time.

Mandatory Principal Repayment means at any time:

- (a) if the aggregate credit balance of the Trust Bank Accounts is not less than the Alternative Minimum Cash Balance, an amount not less than the Deferred Release Price Percentage of the Release Price for the LM Mortgaged Property; and
- (b) in all other cases, an amount equal to the Release Price.

Marketable Securities has the meaning given to "marketable securities" in section 9 of the Corporations Act.

Material Adverse Effect means a material adverse effect on the:

- (a) ability of the Borrower to perform its obligations under any Finance Document to which it is a party; or
- (b) business, assets or financial condition of the Borrower;
- (c) the Trust Property; or
- (d) the validity or enforceability of the Financier's Powers under the Security;
- (e) the validity or enforceability of the Borrower's Power under the LM Mortgages;
- (f) the value or marketability of a LM Mortgaged Property where, at the relevant time, the balance of the Adversely Affected LM Loans exceeds (in aggregate) \$2,000,000;
- (g) the ability of the Custodian to perform its obligations under the Custody Agreement or Securities Custody Deed;

(h) the ability of the Service Provider (or any replacement successor) to perform its obligations under the Service Agreement.

Material Authorisation means in relation to the Borrower, any Authorisation required to be obtained by it in accordance with clause 16.8(a), excluding any Authorisation required under any Environmental Law.

Minimum Cash Balance means, subject to clause 16.11, \$6,000,000 in Cash or Cash Equivalent Investments or any lesser amount agreed to by the Financier from time to time in writing.

Minimum Monthly Interest Payment means an amount calculated on the basis as if the Interest Rate was 10% per annum.

Monthly Interest Payment Date means the last day of each calendar month following the first Utilisation Date but excluding each Biannual Interest Payment Date provided that the last Monthly Interest Payment Date shall be the Termination Date.

Net Proceeds means the consideration received by the Borrower in respect of the Disposal of any LM Mortgaged Property but after deduction of:

- (a) all Taxes applicable to the Disposal (including Goods and Services Tax); and
- (b) reasonable agents' fees;
- (c) all other reasonable costs, fees and expenses properly incurred by the Borrower payable to an unrelated third party in arranging and effecting the Disposal.

New Lending Criteria means the requirements set out in Schedule 5.

Obligations means all the liabilities and obligations of the Borrower to the Financier under or by reason of any Finance Document and includes any liabilities or obligations which:

- (a) are liquidated or unliquidated;
- (b) are present, prospective or contingent;
- (c) are in existence before or come into existence on or after the date of this agreement;
- (d) relate to the payment of money or the performance or omission of any act;
- (e) sound in damages only;
- (f) accrue as a result of any Event of Default; or
- (g) would exist but for an Event of Insolvency affecting any person,

and irrespective of:

- (h) whether the Borrower is liable or obligated solely, jointly or jointly and severally with another person;
- the circumstances in which the Financier comes to be owed each liability or obligation, including any assignment of any liability or obligation; or
- (j) the capacity in which the Borrower and the Financier comes to owe or to be owed that liability or obligation.

Permitted Disposal means:

- (a) any Disposal for cash consideration of obsolete plant and equipment not required for the efficient operation of the Borrower's business, on arm's length terms and at fair market value:
- (b) the expenditure of cash in the ordinary course of its ordinary business carried on in compliance with the terms of the Finance Documents;
- (c) the lending of cash in compliance with the terms of the Finance Documents; and
- (d) any Disposal of a LM Mortgaged Property provided that:
 - (i) each Disposal is for eash consideration on arm's length terms and at fair market value; and
 - (ii) the Financier's prior written consent to the Disposal is obtained (if the Net Proceeds of the Disposal of that LM Mortgaged Property is less than the applicable Release Price).

Permitted Encumbrance means:

- (a) any Encumbrance created under the Securities;
- (b) any Encumbrance (other than any banker's lien) arising by operation of law, in the ordinary course of ordinary business and which secures an amount (not constituting a Financial Liability) which is either:
 - (i) not yet due; or
 - (ii) being contested in good faith by appropriate proceedings if:
 - the Borrower has sufficient financial resources to pay the relevant amount if a legally binding determination is made that payment is required; and
 - B. the enforcement of that Encumbrance is stayed or otherwise suspended during the course of the proceedings; and
- (c) until Financial Close, Encumbrances granted in favour of CBA.

Permitted Financial Liability means:

- (a) any Financial Liability incurred under the Finance Documents;
- (b) a Treasury Transaction entered into during the ordinary course of business.

Permitted Responsible Entity Financial Liabilities means those Financial Liabilities described in clause 15.4(f).

Planning Law means a law regulating (or otherwise relating to) town planning and any plan or regulation made by a Government Authority in relation to the conduct of any activity in relation to the Land.

Potential Event of Default means any event or circumstance which, with the giving of notice, expiration of any applicable grace period, satisfaction of a condition or a determination (or any combination of these) would be an Event of Default.

Power means any right, power, authority, discretion, remedy or privilege.

Principal Shortfall shall have the meaning given to that term in clause 6.2(b)(ii).

Projected Principal Amortisation Amount means on any day during the period from Financial Close to 30 June 2012, an amount calculated as follows:

- (a) Subject to paragraph (b):
 - (i) $(\$1,500,000) \times yz = a;$ and
 - (ii) $($1,875,000) \times xy = b$,

where:

- yz is the number of months which occur during the Calculation Period and occur before 30 June 2011;
- xy is the number of months which occur during the Calculation Period and occur after 30 June 2011 and end on 30 June 2012; and
- a + b = Projected Principal Amortisation Amount.
- (b) If the Borrower does not exercise the option to extend the Facility under clause 2.5

$$a \div 3 = b$$

where:

a is the outstanding Advance as at 1 April 2012

b is the Projected Principal Amortisation Amount

Qualifying Capital Improvements means, in relation to an LM Mortgaged Property, any capital improvements approved by the Financier for that LM Mortgaged Property.

Qualifying Land means:

- (a) if the Land is freehold, the Customer is the registered owner of an estate in fee simple in the Land;
- (b) there is no unsatisfied judgment, order or writ affecting the Land;
- (c) as at the date of this agreement there are no current or threatened claims, notices or pre-proceedings that may lead to a judgment, order or writ affecting the Land;
- (d) there is no outstanding obligation on the owner of the Land to give notice to the
 administering authority under the Environmental Protection Act 1994 (Qld) (EPA)
 (or similar legislation applying in other States and Territories of Australia) of
 notifiable activity being conducted on the Land;
- (e) the Borrower is not aware of any facts or circumstances that may lead to the Land being classified as contaminated land within the meaning of the EPA;
- (f) no order has been made under Part 11 of the Property Law Act 1974 (Qld) (or similar legislation applying in other States and Territories of Australia) which would operate as a charge on the Land;

- (g) no notice has been issued by a competent authority of proceedings issued instituted in a court pursuant to any statute whereby the interest of the owner of the Land may be rendered liable to forfeiture to the Crown; and
- (h) in respect of which the Customer has obtained all approvals under any Planning Law or otherwise from all relevant Government Authorities required to carry out any works or developments on the Land including any development consent required under any Planning Law or otherwise.

Receivable means:

- (a) the receipt of any money by the Borrower which the Borrower is entitled to retain and use for the purposes of its business; and
- (b) an amount payable or paid by a Receiver to the Borrower.

Receiver has the meaning given in section 9 of the Corporations Act.

Receiver and Manager has the meaning given in section 9 of the Corporations Act.

Related Body Corporate has the meaning given in section 9 of the Corporations Act, but on the basis that "subsidiary" for the purposes of that definition has the meaning given to "Subsidiary" in this agreement.

Release Price means in relation to an LM Mortgaged Property the amount specified against that LM Mortgaged Property as the Release Price in Schedule 4 provided that the amount specified will be increased by the amount of any Qualifying Capital Improvements expended in respect of that LM Mortgaged Property.

Released LM Mortgaged Property means an LM Mortgaged Property which has been Disposed and for which the Net Proceeds of sale have been applied in accordance with the provisions of this Agreement.

Repeating Representation means each of the representations set out in clauses 15.1, 15.2, 15.3, 15.4 and 15.5 (other than those in clause 15.2(c), 15.5(g), (l) and (n) or otherwise waived by the Financier).

Responsible Entity means LM Investment Management Ltd ACN 007 208 461 or any successor approved by the Financier.

Right of Indemnity means the rights of the Responsible Entity to be fully indemnified and reimbursed out of the Trust Property for all liabilities incurred by it in its capacity as responsible entity of the LM Mortgage Trust, including those incurred by it under the Finance Documents to which it is a party.

Securities Custody Deed means the securities custody deed to be entered into between the Responsible Entity, the Custodian and the Financier in respect of the custody of the Security.

Security means the Charge or any other Encumbrance held by the Financier at any time for the due satisfaction of the Obligations, as amended from time to time.

Security Property means any property subject to a Security.

Service Provider means LM Administration Pty Ltd ACN 055 691 426 as Trustee for the LM Administration Trust.

Service Provider Side Deed means the deed entitled Service Provider Side Deed dated on or about the date of this agreement between the Service Provider, the Borrower and the Financier.

Services Agreement means the agreement between the Borrower and the Services Provider dated 1 July 2009 for the provision of services by the Services Provider to the Borrower.

Standard & Poor's means Standard & Poor's (Australia) Pty Ltd or any of its Related Bodies Corporate.

Statute means any legislation of the Parliament of the Commonwealth of Australia, of any State or Territory of the Commonwealth of Australia or of any other jurisdiction in force at any time and any rule, regulation, ordinance, by-law, statutory instrument, order or notice at any time made under that legislation and, in each case, any consolidations, amendments, reenactments and replacements.

SCO means the Borrower's Senior Compliance Officer.

Tax means any taxes, levies, imposts, deductions, charges and withholdings assessed, imposed, collected or withheld under any legislation and, in each case, all interest, fines, penalties, charges, fees or other amounts in respect of them.

Termination Date means the earlier of:

- (a) 30 June 2012 (or 30 June 2013 if the date is extended pursuant to clause 2.5) or any other date agreed in writing between the Financier and the Borrower; and
- (b) any date on which the Facility is terminated or cancelled in accordance with this agreement.

Treasury Transaction means any swap or forward exchange agreement in respect of any currency or any similar transaction.

Trust Bank Accounts means accounts on terms and conditions acceptable to the Financier maintained by the Borrower or Custodian on behalf of the Borrower with an Acceptable Bank as described in Schedule 9.

Trust Documents in respect of the LM Mortgage Trust, means:

- (a) the LM Mortgage Trust Constitution;
- (b) the Custody Agreement; and
- (c) the Compliance Plan.

Trust Property means any and all property of the LM Mortgage Trust from time to time held by:

- (a) the Borrower as responsible entity of the LM Mortgage Trust; and
- (b) the Custodian in its capacity as custodian of the LM Mortgage Trust.

Unpaid Sum means any sum due and payable by the Borrower under any Finance Document but unpaid.

Utilisation Date means the date on which an Advance is made.

Utilisation Notice means a notice given under clause 4.1 in the form set out in Part A of Schedule 2.

1.2 Interpretation

In this agreement:

- (a) headings are for convenience only and do not affect interpretation;
 and unless the context indicates a contrary intention;
- (b) person includes an individual, the estate of an individual, a corporation, a Government Authority, an association or a Joint Venture and a trust;
- (c) a reference to a party includes that party's executors, administrators, successors and permitted assigns, including persons taking by way of novation and, in the case of a trustee, includes a substituted or an additional trustee;
- (d) a reference to a document (including any Finance Document) is to that document as varied, novated, ratified, replaced or restated from time to time;
- (e) a reference to a law includes any law, principle of equity, Statute and official directive of any Government Authority and a reference to a Statute includes consolidations, amendments, re-enactments and replacements of it;
- (f) a word importing the singular includes the plural (and vice versa) and a word indicating a gender includes every other gender;
- (g) a reference to a party, clause or schedule is a reference to a party, clause or schedule to or of this agreement, and a reference to this agreement includes all schedules to it;
- if a word or phrase is given a defined meaning, any other part of speech or grammatical form of that word or phrase has a corresponding meaning;
- includes in any form is not a word of limitation;
- all accounting terms used in this agreement have the meaning given to them under Accepted Accounting Practices;
- (k) where the day on or by which any sum is payable or any act, matter or thing is to be done is a day other than a Business Day, that sum will be paid or that act, matter or thing will be done on the next Business Day in the same calendar month (if there is one) or the preceding Business Day (if there is not);
- (I) a reference to a month is, where that month is the last month to occur in any period, a reference to a period starting on the relevant date in a calendar month and ending on the numerically corresponding day in the next calendar month, except that if there is no numerically corresponding day in the next calendar month, the period will end on the last Business Day in that calendar month;
- (m) a reference to subsists or any similar expression in relation to an Event of Default or a Potential Event of Default indicates an Event of Default or Potential Event of Default which has not been remedied or waived in accordance with the terms of the Finance Documents; and
- (n) a time of day is a reference to Sydney time.

1.3 Capacity of Responsible Entity

The Responsible Entity enters into each Finance Document to which it is a party in its capacity as responsible entity or trustee of the LM Mortgage Trust except in relation to clause 15.

1.4 Limitation of Responsible Entity's Liabilities

- (a) A liability of the Responsible Entity arising under or in connection with a Finance Document is limited to the extent to which the liability can be satisfied out of the assets of the LM Mortgage Trust by the Responsible Entity exercising its rights of indemnity out of those assets. This limitation applies despite any other provision of a Finance Document (except the other provisions of this clause 1.4) and extends to all liabilities and obligations of the Responsible Entity in any way connected with any representation, warranty, conduct, omission, deed or transaction related to a Finance Document.
- (b) The Financier may not seek to recover any amounts owing to it under any Finance Documents by applying to have the Responsible Entity wound up or placed in administration under Part 5.3A of the Corporations Act, 2001 or by seeking to appoint a receiver or receiver and manager to any of its assets of the Responsible Entity other than a receiver or receiver and manager to, or in respect of, any Trust Property. However, the Financier may:
 - (i) do anything necessary to enforce its rights in connection with the Trust Property; and
 - (ii) take proceeds to obtain:
 - A. an injunction or other order to restrain any breach of any
 Finance Document by the Responsible Entity or to compel the
 Responsible Entity to perform any of its obligations at law or
 under any Finance Document; or
 - B. declaratory relief or other similar judgment or order as to the obligations of the Responsible Entity under any Finance Document.
- (c) The limitation and restrictions in clause 1.4(a) and 1.4(b) do not apply to any obligation or liability of the Responsible Entity to the extent that is not satisfied because under the LM Mortgage Trust Constitution, or other constitutional documents, or by operation of law there is a reduction in the extent of the Responsible Entity's indemnification out of the Trust Property, as a result of the Responsible Entity's fraud, negligence or wilful default or by operation of law;
- (d) The Responsible Entity is not obliged to do or refrain from doing anything under a Finance Document (including incur any liability) unless the Responsible Entity's liability is limited in the manner which is consistent with this clause 1.4 or which is otherwise satisfactory to the Responsible Entity in its absolute discretion;
- (e) A failure by the Responsible Entity to comply with or a breach by the Responsible Entity of any of its obligations under a Finance Document will not be considered to be a fraud, negligence or wilful default by the Responsible Entity if the relevant failure or breach:
 - (i) arose as a result of a breach by a person other than the Responsible Entity where the performance of the action (the non-performance of which gave

rise to such breach) is a precondition to the Responsible Entity performing the said obligation; or

- (ii) was in accordance with a lawful court order or direction or otherwise required by law;
- (f) The limitation of the Responsible Entity's liability under this clause 1.4 is to be disregarded for the purposes of determining when a liability of the Responsible Entity is due and payable, whether an Event of Default has occurred because of a failure by the Responsible Entity to pay an amount payable by it under any Finance Document, whether an Event of Default has occurred or any other reason, for exercising any rights in connection with the Trust Property and in interpreting the definition of Security Property in any Finance Document.
- (g) Paragraphs (a) to (f) of this clause 1.4 are subject to paragraph (g) of this clause 1.4.

 All of the terms, clauses and conditions of this Agreement are subject to this clause 1.4.

1.5 Custodian assets and obligations

Any reference in this agreement to:

- (a) an obligation owed by the LM Mortgage Trust includes an obligation owed by the Custodian in relation to the LM Mortgage Trust; and
- (b) any rights, assets or undertakings of the LM Mortgage Trust includes the rights, assets or undertakings of the Custodian of the LM Mortgage Trust in relation to the LM Mortgage Trust.

2. Facility

2.1 Facility

Subject to the terms of the Finance Documents and in reliance on the representations and warranties set out in clause 15 and in the other Finance Documents, the Financier grants to the Borrower a term loan facility under which the Financier will make an Advance in dollars to the Borrower during the Availability Period in an aggregate amount not greater than the Commitment.

2.2 Purpose

- (a) The Facility may only be used for the following purposes:
 - (i) first, repaying the CBA Loan in full;
 - (ii) secondly, establishing the Minimum Cash Balance;
 - (iii) thirdly, an amount up to \$1,600,000 shall be applied towards paying any Taxes, insurance or statutory charges which are due and unpaid in respect of any LM Mortgage within 30 days of Financial Close; and
 - (iv) fourthly, paying all outstanding fees, costs and expenses due from the Borrower, including under clause 9 or 20.
- (b) The Financier is not bound to monitor or verify the application of any Advance and the Financier will not have any liability to any person arising from a failure by the Borrower to use any Advance for a purpose specified in this clause 2.2.

2.3 Termination

The Facility terminates on the Termination Date.

2.4 Financier's Commitments

The Financier is not obliged to make an Advance if to do so would cause the aggregate of the Advance to exceed the Commitment.

2.5 Option to Extend Termination Date

Provided no Event of Default or Potential Event of Default has occurred and is subsisting, the Borrower will be entitled upon written notice to the Financier given not later than 120 days and not before 180 days prior to 30 June 2012 elect to extend the Termination Date to 30 June 2013.

2.6 Deferred Release Price Review

If a Deferred Release Price Review Event occurs the Financier agrees that it will consider any request by the Borrower to reduce the Deferred Release Price Percentage.

3. Conditions precedent

3.1 Conditions precedent to the Advance

The obligation of the Financier to make the Advance available to the Borrower is subject to the Financier being satisfied that on the Utilisation Date:

- (a) (Documents) the Financier has received all of the documents listed in Part A of Schedule 1 in form and substance satisfactory to it;
- (b) (Fees) all fees, costs and expenses then due from the Borrower, including under clause 9 or 20, have been paid or will be paid on the Utilisation Date;
- (c) (Representations and warranties true) the Repeating Representations are true and correct on those dates and will be true and correct immediately after the making of the Advance, in each case, with reference to the facts and circumstances existing at that time; and
- (d) (No Event of Default) no Event of Default or Potential Event of Default is subsisting or will result from the making of that Advance.

4. Advance

4.1 Notice

The Borrower may request an Advance by giving a Utilisation Notice to the Financier. Only one Utilisation Notice can be issued.

4.2 Contents of Utilisation Notice

The Utilisation Notice will be in the form of Schedule 2 and will specify:

 (a) the proposed Utilisation Date which must be a Business Day before the expiry of the Availability Period;

- (b) the amount of the Advance which must be not less than \$2,000,000 and not more than the Commitment;
- (c) the purpose for which the Advance will be applied;
- (d) payment instructions; and
- (e) any other details as the Financier reasonably requires.

4.3 Requirements of Utilisation Notice

The Utilisation Notice will be irrevocable and must be:

- (a) received by the Financier 3 Business Days before the proposed Utilisation Date; and
- (b) signed by an Authorised Officer of the Borrower.

4.4 Making of Advance

The Financier will make the Advance to or to the order of the Borrower in accordance with the provisions of the relevant Utilisation Notice and this agreement.

4.5 Maximum number of Advance

The Borrower may not deliver more than one Utilisation Notice and the Financier is only required to make one Advance subject to and in accordance with the terms of this agreement.

5. Cancellation and reduction of Commitment

5.1 Commitment

Any part of the Commitment not utilised under this agreement before the expiry of the Availability Period will be cancelled automatically on the expiry of the Availability Period.

5.2 Voluntary cancellation or reduction

- (a) The Borrower may, on giving not less than 5 Business Days' prior notice to the Financier, cancel or reduce the Commitment in whole or in part (but, if in part, by a minimum of and in multiples of \$2,500,000) without incurring any penalty or other cost, provided that the cancellation or reduction may only be effected to the extent of the amount of the applicable Commitment unutilised on the date of the cancellation or reduction.
- (b) Any notice under clause 5.2(a) will be irrevocable and must be signed by an Authorised Officer of the Borrower and must specify the date on which the cancellation or reduction is to become effective and the amount of the cancellation or reduction.

5.3 Reduction consequent on repayment or prepayment

- (a) The Commitment will be reduced by the amount of any repayment or prepayment of an Advance made under clauses 6.1, 6.2 or 6.3.
- (b) The Commitment will be reduced by the amount of any other prepayment of an Advance made under any other provision of this agreement.

5.4 Limitations

- (a) Any Commitment reduced, cancelled or otherwise extinguished under this agreement may not be reinstated.
- (b) The Commitment may only be reduced, cancelled or otherwise extinguished as expressly provided under this agreement.

6. Repayment and prepayments

6.1 Voluntary prepayment

- (a) The Borrower may prepay an Advance, in whole or in part (but, if in part, by a minimum of and in multiples of \$1,000,000), on giving not less than 5 Business Days' prior notice to the Financier.
- (b) Any notice of prepayment given by the Borrower is irrevocable and the Borrower is bound to prepay in accordance with the notice.
- (c) The Borrower may not voluntarily prepay any Advance except in accordance with this clause 6.1.

6.2 Mandatory prepayments Disposal of LM Mortgaged Properties

- (a) Subject to clause 6.2(b), if there is a Disposal of any of the LM Loans or LM Mortgaged Property or any LM Loan Payment is made, the Borrower agrees that the Net Proceeds will be applied as follows:
 - (i) firstly, payment to the Financier of any fees or Costs which are due and payable by the Borrower to the Financier under the Finance Documents;
 - (ii) secondly, in payment to the Financier of any accrued interest or Unpaid Sums:
 - (iii) thirdly, in payment to the Financier of any indemnity payments payable under the Finance Documents;
 - (iv) fourthly, in payment to the Financier of an amount equal to any Principal Shortfall (which shall be applied by the Financier in repayment of Advance);
 - (v) fifthly, an amount equal to the Mandatory Principal Repayment is applied from the Net Proceeds in or towards prepayment of the Advance in accordance with this clause 6. The prepayment will, subject to clause 6.5, be made immediately on the relevant Disposal occurring;
 - (vi) sixthly, any third party costs incurred by the Borrower in:
 - A. exercising or enforcing any LM Mortgage or Collateral Security; and
 - B. paying any land tax, rates, body corporate fees or insurance payable but unpaid by a Customer in respect of a LM Mortgaged Property; and
 - (vii) lastly, provided the Minimum Cash Balance is maintained, the balance, if any, may be paid to or retained by the Borrower.

- (b) If the Net Proceeds of an LM Mortgaged Property are less than the Mandatory Principal Repayment then:
 - (i) an amount equal to the Net Proceeds shall be paid in accordance with clause 6.2(a); and
 - (ii) the difference between the Release Price and the Net Proceeds (such difference being a Principal Shortfall) shall be paid within 90 days of the Principal Shortfall occurring in or towards repayment of the Advance from the Net Proceeds of any subsequent sale of any LM Mortgaged Property in accordance with clause 6.2(a) provided that at no time will the Principal Shortfall exceed \$3,000,000; and
 - (iii) to the extent that the Principal Shortfall is not discharged it shall be carried forward and discharged from any subsequent sale of a LM Mortgaged Property.

6.3 Mandatory prepayments - Insurance Proceeds

If:

- (a) as a result of making a claim under an insurance policy (other than in relation to any public liability, directors' and officers' liability, workers' compensation or any other third party liability) the Borrower receives or recovers any proceeds then the Borrower will, unless the Financier agrees to the contrary, apply the insurance proceeds as follows:
 - first, to the costs of restoring or replacing the property for which the insurance proceeds are received;
 - (ii) second, to compensate the Financier for any diminution of the value of the Financier's Security as reasonably determined by the Financier, and
 - (iii) the balance, if any, to the Borrower.
- (b) the Borrower receives any proceeds of insurance in respect of an LM Mortgaged Property, the Borrower will, unless the Financier agrees to the contrary, pay the insurance proceeds or an amount equal to those insurance proceeds in or towards prepayment of the Advance in accordance with this clause 6. The prepayment will, subject to clause 6.5, be made immediately on the relevant amount being received by the Borrower.

6.4 Order of application

Subject to clause 6.2 any amount paid by the Borrower to the Financier will be applied in discharge of the obligations in such order as the Financier determines notwithstanding any direction or appropriation by the Borrower.

6.5 Date for prepayment

If any prepayment is required under clause 6.2 or 6.3, the prepayment will be made on the last day of the Interest Period relating to the Advance to be prepaid and pending the prepayment must be retained in the Managed Cash Account. All interest earned in respect of the prepayment account is for the account of the Borrower.

6.6 General provisions relating to repayment and prepayment

- (a) Amounts repaid or prepaid in respect of any Advance under any provision of this agreement may not be re-utilised.
- (b) Any repayment or prepayment of any amount under this agreement will be made together with interest and fees accrued on the amount repaid or prepaid and any amount required to be paid in accordance with clause 18 but otherwise without premium or penalty.

7. Interest

7.1 Interest Periods

- (a) The Interest Period for each Advance (other than the first) will be one month's duration.
- (b) The first Interest Period in relation to any Advance is the period commencing on the first Utilisation Date for that Advance and ending on the last day of the month in which that Utilisation Date falls. Any subsequent Interest Period in relation to an Advance will commence on the last day of the immediately preceding Interest Period.
- (c) Notwithstanding any other provision of this clause 7.1, the term of each Interest Period is subject to any adjustment as the Financier in its discretion determines so that:
 - (i) the first and last days of it are Business Days;
 - (ii) no Interest Period expires later than the Termination Date.
- (d) The Financier will promptly notify the Borrower of any adjustment made by the Financier to any Interest Period determined in accordance with this clause 7.1.

7.2 Calculation of interest

- (a) Interest on each Advance accrues daily and is to be computed on the basis of the actual number of days elapsed and a year of 365 days.
- (b) The rate of interest for each Advance for each Interest Period is the Interest Rate in relation to that Interest Period for that Advance.

7.3 Payment of interest

- (a) On each Monthly Interest Payment Date the Borrower will pay to the Financier interest in relation to each Advance in an amount not less than the Minimum Monthly Interest Payment. Any interest which has accrued and is not paid shall be capitalised and form part of that Advance.
- (b) On each Biannual Interest Payment Date the Borrower will pay to the Financier interest in relation to each Advance in an amount not less than the Biannual Interest Payment.

7.4 Market disruption

(a) The Financier may determine whether a Market Disruption Event has occurred in relation to an Advance during an Interest Period. If the Financier determines that a

Legal\301465383.15 26

Market Disruption Event has occurred, it will as soon as practicable and in any event no later than the Business Day before interest is due to be paid in respect of that Interest Period, notify the Borrower of the rate that expresses as a percentage rate per annum either:

- (i) aggregated BBSW and 5% per annum;
- (ii) if BBSW is not quoted, the aggregate of the cost to the Financier of funding that Advance plus 5% per annum from whatever source or sources it may reasonably select for the relevant Interest Period.
- (b) The Financier will determine the rate notified by it under clause 7.4(a) in good faith. The rate so notified, and any notification under clause 7.4(a), will be conclusive and binding on the parties in the absence of manifest error.
- (c) In this agreement a Market Disruption Event occurs if, as a result of market circumstances not limited to the Financier (whether or not those circumstances, or their effect on the Financier's cost of funds, subsist on the date of this agreement), the cost to it of funding that Advance exceeds the cost to the Financier on the date of this agreement.

8. Interest on overdue amounts

8.1 Payment of interest

The Borrower will pay to the Financier interest on any Unpaid Sum, including any interest payable under clause 7 and this clause 8.

8.2 Accrual of interest

- (a) Default interest will accrue on all Unpaid Sums from day to day:
 - (i) from the due date up to the date of actual payment, before and (as a separate and independent obligation) after judgment; and
 - (ii) (subject to clause 8.2(b)) at the Interest Rate plus the Default Margin commencing on the date of default.
- (b) If an Unpaid Sum consists of all or any part of an Advance that has become due and payable before the last day of an Interest Period for that Advance, then in respect of that Unpaid Sum:
 - its first Interest Period will be the unexpired portion of the relevant Interest Period for that Advance;
 - (ii) during its first Interest Period, default interest will accrue at a rate equal to Default Margin above the rate which would have been payable if that Unpaid Sum had not become due and payable; and
 - (iii) after its first Interest Period, default interest will accrue at a rate calculated in accordance with clause 8.2(a)(ii).
- (c) Default interest accruing under this clause 8.2 will be immediately payable by the relevant the Borrower on demand. Default interest (if unpaid) arising on any Unpaid Sum will be compounded with that Unpaid Sum at the end of each Interest Period applicable to that Unpaid Sum but will remain immediately due and payable.

27

9. Fees

9.1 Establishment fee

- (a) The Borrower will pay to the Financier a non-refundable establishment fee in dollars computed at the rate of 1.5% on the aggregate of the Commitment. The establishment fee will be payable in full on Financial Close.
- (b) The Financier acknowledges that it has received from the Borrower, prior to the date of this agreement, a non-refundable fee in the amount of 0.5% on the aggregate of the Commitment which it shall apply towards part payment of the establishment fee referred to in clause 9.1(a) on Financial Close.

9.2 Commitment fee

- (a) The Borrower will pay to the Financier an unused commitment fee in dollars computed at the rate of 6% per annum on the daily unutilised balance of the aggregate of the Commitment during the Availability Period.
- (b) The accrued commitment fee will be payable quarterly in arrears from the date of this agreement and also on any date on which the Commitment is terminated.
- (c) The commitment fee will accrue from day to day and be calculated on the basis of a year of 365 days and for the actual number of days elapsed.

10. Payments

10.1 Time and place

All payments by the Borrower under any Finance Document are to be made to the Financier in dollars or any other currency specified in the relevant Finance Document in immediately available funds not later than 11.00 am in the place of payment on the due date to the account that the Financier from time to time designates.

10.2 No set-off, counterclaim or merger

- (a) All payments by the Borrower under any Finance Documents will be without any set-off or counterclaim and the Borrower irrevocably waives any right of set-off or counterclaim which it may have against the Financier.
- (b) If the liability of the Borrower to pay any money, the payment or repayment of which forms part of the Obligations, becomes merged in any judgment or order, the Borrower will as an independent obligation pay to the Financier interest at the rate which is the higher of that payable under the Finance Documents and that fixed by or payable under the judgment or order.

10.3 Currency for payments

- (a) Except as otherwise expressly provided in any Finance Document:
 - (i) each payment due from the Borrower in respect of Costs or Taxes will be made in the currency in which the Costs or Taxes are incurred; and
 - (ii) all other payments due from the Borrower under any Finance Document will be made in dollars.

- (b) If any amount to be applied in satisfaction of the Obligations is in a currency (Foreign Currency) other than the currency in which that amount is required to be paid under the Finance Documents (Due Currency), the Financier in its absolute discretion may accept application in that Foreign Currency.
- (c) If any application of an amount in a Foreign Currency is accepted by the Financier as contemplated by clause 10.3(b), or if any amount is recovered by the Financier under any Finance Document in a Foreign Currency (whether as a result of any judgment or order, the winding up of the Borrower or otherwise), the Financier in its absolute discretion may actually or notionally convert that amount to the Due Currency at any time or times as it sees fit and at any rate or rates as it is, or considers it would be, able to obtain in the relevant foreign currency market at the time of that conversion. The amount of the Due Currency actually or notionally received after conversion will be applied in reduction of the Obligations.
- (d) The Borrower will pay to the Financier all commissions and Costs incurred by the Financier in connection with the actual or notional conversion of any Foreign Currency into the Due Currency on the terms contemplated by clause 10.3(c).

10.4 Payments in gross

All amounts received or recovered by the Financier on account of the Obligations will be treated as payments in gross without any right on the part of the Borrower to claim the benefit of any amount received or recovered by the Financier under any Finance Document, until the Obligations have been fully and finally satisfied.

10.5 Payment on a Business Day

When any payment under a Finance Document would otherwise be due on a day which is not a Business Day, the due date for payment shall be the immediately preceding Business Day but the amount payable shall not be adjusted as a result of the change of the due date.

11. Taxes

11.1 No deduction for Taxes

All payments by the Borrower under any Finance Document, whether of principal, interest or other amounts, will be without deduction or withholding for any present or future Taxes unless the Borrower is compelled by law to deduct or withhold the same.

11.2 Payment net of Taxes

If:

- (a) the Borrower is compelled by law to make any deduction or withholding from any payment under any Finance Document on account of Taxes; or
- (b) the Financier is obliged to pay any Taxes in respect of a payment made or to be made by the Borrower under the Finance Documents,

then:

- (i) the Borrower will promptly notify the Financier if it becomes aware of any event referred to in clauses 11.2(a) to 11.2(b);
- (ii) the Borrower will on demand by the Financier pay to the Financier any additional amounts necessary to ensure that the Financier receives a net

amount equal to the full amount which it would have been entitled to receive and retain had the deduction or withholding not been made or had the payment been free and clear of Taxes or had the Financier not been obliged to pay any Taxes in respect of the payment; and

- (iii) where clause 11.2(a) applies the Borrower compelled to make the deduction or withholding will:
 - A. pay to the appropriate Government Authority any amount deducted or withheld in respect of Taxes within the time allowed and in the minimum amount required by law; and
 - B. within 20 Business Days after making the deduction or withholding, provide to the Financier evidence satisfactory to the Financier of that payment having been made.

11.3 Tax saving

- (a) If the Borrower is required under clause 11.2 to pay Taxes or to pay any additional amount in respect of Taxes and as a result the Financier (in its absolute discretion and based on its own interpretation of any relevant laws):
 - (i) receives or is granted a credit against or remission for or deduction from or in respect of any Tax payable by it; or
 - (ii) will obtain any other relief in respect of Tax,

(referred to as a saving), the Financier will, to the extent that it can do so without prejudice to the retention of the saving, pay to the Borrower an amount that the Financier in its absolute discretion concludes to be the amount or value of the saving.

- (b) If after any payment is made to the Borrower under clause 11.3(a) the Financier is not able to obtain the benefit of the saving or the saving is subsequently disallowed or cancelled, the Borrower will, within 5 Business Days of demand by the Financier, refund the payment made to it in respect of the saving or the relevant part.
- (c) Nothing in this agreement will:
 - (i) interfere with the right of the Financier to arrange its Tax and other affairs in whatever manner it thinks fit; or
 - (ii) require the Financier to:
 - A. disclose to the Borrower or any other person any details of its Tax or other affairs;
 - B. claim relief from Tax on its corporate profits, or from any similar Tax liability, in respect of the Tax referred to above; or
 - claim relief in priority to any other claims, reliefs, credits or deductions available to it.

- (a) attributable to a deduction or withholding on account of Tax required to be made by law;
- (b) compensated for by clause 11.2 or would have been but for an exception in that clause; or
- (c) attributable to the wilful breach of any law by the Financier.

14. Mitigation

14.1 Mitigation

If circumstances arise which would, or on the giving of notice would, result in the operation of clauses 11, 12 or 13:

- (a) the Financier will promptly notify the Borrower on becoming aware of the applicable circumstances (provided that a failure to provide notice in accordance with this clause 14.1(a) will not prejudice the rights of the Financier under clauses 11, 12 or 13);
- (b) the Financier will, on request by the Borrower, enter into discussions with the Borrower to determine what mitigating action might be taken by the Financier including discussion of the possibility of a change in the office through which it provides the Facilities, a change in the method of funding Advance or a transfer of the Commitment to another person; and
- (c) at the request of the Borrower, the Financier will enter into discussions with the Borrower to determine what mitigating action might be taken by the Financier with respect to the administration of this agreement,

provided that nothing in this clause 14.1 will oblige the Financier to incur any Costs or to take any action or refrain from taking any action or affect the obligations of the Borrower under clauses 11, 12 or 13.

14.2 Costs and expenses

Any costs and expenses reasonably incurred by the Financier under this clause 14 will be paid by the Borrower within 5 Business Days after receipt of a demand specifying the costs and expenses in reasonable detail.

15. Representations and warranties

15.1 General representations and warranties

The Borrower and the Responsible Entity in its personal capacity represents and warrants to the Financier that:

- (a) (Legally binding obligation) each Finance Document to which it is a party
 constitutes its valid and legally binding obligation and is enforceable against it in
 accordance with its terms, subject to any necessary stamping and registration
 requirements and laws affecting creditors' rights generally;
- (b) (Execution, delivery and performance) the execution, delivery and performance of each Finance Document to which it is a party and each transaction contemplated by those Finance Document does not violate or breach any law or any document or agreement to which it is a party or which is binding on it or any of its assets;

Legal/301465383.15 32

- (c) (Authorisation) all Authorisations required:
 - to be obtained by it in connection with the execution, delivery and performance of each Finance Document to which it is a party, the transactions contemplated by those Finance Documents and the legality, validity and enforcement of each of those Finance Documents;
 - (ii) for it or for any agents employed by the Borrower to carry on business and which are material; and
 - (iii) in the case of the Responsible Entity, for the LM Mortgage Trust to be established.

have been obtained and are valid and subsisting and it is not aware of any breach of any of the conditions of any of those Authorisations or aware of any fact or circumstance which may cause any of those Authorisations to be suspended, forfeited, cancelled or rendered void;

- (d) (No litigation) other than the Disclosed Litigation, no litigation, arbitration, criminal or administrative proceedings are current, pending or to its knowledge threatened which, if determined adversely to it, is reasonably likely to have a Material Adverse Effect;
- (e) (No Event of Default) no Event of Default has occurred which is subsisting;
- (f) (Laws) it has complied with all laws (except to the extent that the non-compliance is not material);
- (g) (Financial Liabilities) is not in default in the payment of any amount, or in the satisfaction of any obligation in respect of any Financial Liability and no event has occurred which with the giving of notice, lapse of time or other condition could constitute a default in respect of any such Financial Liability and require its prepayment;
- (h) (Solvency) no Event of Insolvency has occurred in relation to it;
 - (i) (No financial assistance or benefit to related party) the execution and delivery by it of any Finance Document to which it is a party or the participation by it in any transaction in connection with any Finance Document to which it is a party will not contravene Part 2J.3 or Chapter 2E of the Corporations Act;
 - (ii) (Environmental matters) all Authorisations required by it under any Environmental Law in order for it or to carry on business have been obtained and are valid and subsisting and it has at all times complied in all material respects with the terms and conditions of those Authorisations;
- (i) (Tax liabilities)
 - no claims are being or are reasonably likely to be asserted against it with respect to Taxes which are reasonably likely to be determined adversely to it;
 - (ii) is not overdue in the filing of any Tax returns or other information required to be filed by it with any relevant Tax authority to ensure compliance with any obligation to pay Tax; and

- (iii) has paid all Taxes due and payable by it by the due date (except to the extent that the payment of any Tax is being contested in good faith where the Borrower has sufficient financial resources to pay that Tax if a legally binding determination is made and that payment is required); and
- (j) (Immunity from suit) it does not enjoy immunity from suit or execution in relation to its obligations under any Finance Document.

15.2 Information representations and warranties

The Borrower and the Responsible Entity in its personal capacity represents and warrants to the Financier that:

- (a) (No misrepresentation) all information relating to the Borrower and the Finance Documents provided by it (or on its behalf) to the Financier (or any advisor of the Financier in connection with any Finance Document) was at the time it was provided or as at the date (if any) at which it was stated true in all material respects and was not, by omission or otherwise, misleading in any material respect at that time;
- (b) (Accounts) the Borrower's most recent Accounts delivered to the Financier under clause 16.2(a):
 - (i) have been prepared in accordance with applicable laws and Accepted Accounting Practices consistently applied except to the extent disclosed in the Accounts;
 - (ii) if audited, give a true and fair view of or, if unaudited, fairly represent the financial condition of the Borrower for the period to which those Accounts relate and of the results of the operations of the Borrower for that period and since the last date of that accounting period there has been no material adverse change in the financial condition of the Borrower as shown in those Accounts; and
 - (iii) all material Financial Liabilities and other material contingent liabilities of the Borrower are disclosed in those Accounts; and
- (c) (Documents) the documents delivered to the Financier under clause 3.1(a) were genuine or, in the case of copy documents, were true, complete and accurate copies of originals which, in each case, have not been amended, varied, supplemented or superseded in any way.

15.3 Corporate representations and warranties

The Responsible Entity in its personal capacity represents and warrants to the Financier that:

- (a) (Existence and power) it:
 - (i) is duly registered and remains in existence; and
 - (ii) has the power to act as responsible entity of the LM Mortgage Trust;
- (b) (No violation) its execution, delivery and performance of each Finance Document to which it is a party does not (and the transactions contemplated by those Finance Documents do not) violate the LM Mortgage Trust Constitution;

- (c) (Power) it has the corporate power to enter into, deliver and perform each Finance
 Document to which it is a party and to carry out the transactions contemplated by
 those Finance Documents;
- (d) (Due authority) it has taken all corporate action required to enter into, deliver and perform each Finance Document to which it is a party and to carry out the transactions contemplated by those Finance Documents; and
- (e) (Corporate benefit) the entry into and the performance of its obligations under each Finance Document to which it is a party is for its corporate benefit.

15.4 Responsible Entity representations and warranties

The Responsible Entity in its personal capacity represents and warrants to the Financier that:

- (a) (General representations & warranties) in relation to the LM Mortgage Trust:
 - (i) (Responsible entity) it has power to enter into the Finance Documents in its capacity as responsible entity of the LM Mortgage Trust;
 - (ii) (LM Mortgage Trust validly created) the LM Mortgage Trust has been validly created and is in existence at the date of this agreement;
 - (iii) (Responsible entity and Custodian validly appointed)
 - A. it has been validly appointed as responsible entity of the LM Mortgage Trust and is presently the sole responsible entity of the LM Mortgage Trust; and
 - B. the Custodian has been validly appointed as custodian of the LM Mortgage Trust and is the sole custodian of the LM Mortgage Trust;
 - (iv) (Trust Documents)
 - A. no Trust Document is void, voidable or otherwise unenforceable;
 - B. the LM Mortgage Trust is solely constituted by the LM Mortgage Trust Constitution, a true copy of which was provided to the Financier before the date of this agreement;
 - C. the Custody Agreement provided to the Financier before the date of this agreement discloses all of the terms of the custody arrangements between it and the Custodian in respect of the LM Mortgage Trust (except to the extent those arrangements are affected by the Securities Custody Deed); and
 - D. the Services Agreement provided to the Financier before the date of this agreement discloses all of the terms of the services arrangements between it and the Service Provider in respect of the LM Mortgage Trust (except to the extent those arrangements are affected by the Service Provider Side Deed);
 - (v) (Rights of indemnity and exoneration against Trust property) as responsible entity of the LM Mortgage Trust, it has valid Rights of Indemnity against the Trust Property, which rights are available for

- satisfaction of all liabilities and other obligations incurred under the Finance Documents to which it is a party;
- (vi) (Due constitution of Trust) the LM Mortgage Trust has been duly constituted and the LM Mortgage Trust Constitution is not void, voidable or otherwise unenforceable, and the LM Mortgage Trust Constitution (and the Compliance Plan if any) complies in all material respects with applicable requirements of the Corporations Act;
- (vii) (Legal owner of Trust Property) subject to the LM Mortgage Trust, it or the Custodian is the legal owner of all of the Trust Property and all undertakings in relation to the LM Mortgage Trust carried on by it, in each case, free from any Encumbrance other than a Permitted Encumbrance; and
- (viii) (No action for removal) no action has been taken or proposed to:
 - remove it as responsible entity of the LM Mortgage Trust of which it is aware; or
 - B. remove the Custodian as custodian for the LM Mortgage Trust:
- (b) (Commercial benefit) it has carefully considered the purpose of the Finance
 Documents to which it is a party and considers that its entry into the Finance
 Documents (and the grant of the Securities) is for the benefit and in the interests of
 the LM Mortgage Trust;
- (c) (Licences) it holds and will maintain all licences and authorisations as are necessary to lawfully perform its obligations under the LM Mortgage Trust Constitution and any applicable law relating to its obligations as responsible entity of the LM Mortgage Trust;
- (d) (AFS licence) it has complied and will continue to comply with the conditions of its Australian Financial Services licence;
- (e) (Insolvency) there is no reasonable grounds to suspect that the LM Mortgage Trust will not be able to pay its debts as and when they become due and payable;
- (f) (Financial Liabilities) other than Financial Liabilities (i) in respect of guarantees given by the Responsible Entity in its personal capacity in respect of certain audit and accounting liabilities of each of LM FZE, LM Investment Management New Zealand Ltd, LM Investment Management International Ltd and LM Investment Management UK Ltd or (ii) owed or owing to the Service Provider for an amount not exceeding \$500,000, it has no Financial Liabilities in its personal capacity (and for the avoidance of doubt, this does not extend to any Financial Liabilities that LM Investment Management Limited may have as trustee or responsible entity of any other trust scheme).

15.5 Specific Portfolio Representations

The Borrower and the Responsible Entity in its personal capacity represents and warrants that:

(a) (LM Documentation Assumptions): each LM Documentation Assumption is true and correct and has been satisfied and complied with;

- (b) (priority arrangements): where there is a second or other mortgage in existence over Land the subject of an LM Mortgage, the Borrower has (other than as disclosed in Schedule 7) ensured (by way of a priority agreement with the subsequent mortgagee or otherwise) that the LM Mortgage will rank in priority to the second or other mortgage on enforcement for an amount not less than the principal amount (plus accrued but unpaid interest) outstanding on the LM Loan;
- (c) (Customer not insolvent): at the time that the LM Loan was approved, the Borrower had not received any notice of the insolvency of the Customer or that the Customer did not have the legal capacity to enter into the LM Mortgage relating to the LM Loan;
- (d) (Custodian and Borrower sole legal and beneficial owner): the Custodian and the Borrower are the sole legal and beneficial owners of the LM Loan Documents and (other than as disclosed in Schedule 7) no prior ranking Encumbrance exists in relation to their right, title and interest in any LM Loan Document
- (e) (LM Loan not discharged): the LM Loan has not been satisfied, cancelled, discharged or rescinded (other than where Net Proceeds are applied in accordance with this agreement or except as disclosed in Schedule 7) and the Land relating to each relevant LM Mortgage has not been released from the security of that LM Mortgage except for the Released LM Mortgaged Property;
- (f) (hold all documents necessary to enforce): except as disclosed (in Schedule 7) or as released in accordance with the Securities Custody Deed, the Custodian holds all LM Loan Documents and documents it should hold to enforce the provisions of any security created by the LM Loan Documents;
- (g) (no notice of security interests): other than in respect of priorities granted by statute or as disclosed in Schedule 7 (or as otherwise notified by the Borrower to the Financier following Financial Close), the Borrower has not received and the Custodian has not received notice from any person that it claims to have an Encumbrance ranking in priority to or equal with the Encumbrance held by the Borrower or Custodian and constituted by any LM Mortgage;
- (h) (all licences and consents): the Borrower and Custodian holds all consents, licences, approvals, authorisations and exemptions from any Governmental agency required for, or in connection with, performance and enforceability in respect of any LM Loan Document;
- (i) (Loan Files complete): the LM Loan Documents in the Loan Files are either originals or, if originals are unavailable, true copies of the documents or materials that they purport to be, all Customer information included in the Loan Files is correct at the time of its inclusion, and neither the Borrower nor the Custodian has knowingly omitted to include in the Loan Files any material agreement, document or other material in its possession or control;
- (i) (no LM Loan Document): neither the Borrower nor the Custodian has breached any of its representations, warranties, obligations, or covenants under any LM Loan Document;
- (k) (value of mortgaged property): except as disclosed in Schedule 7:
 - (i) neither the Borrower nor Custodian has received any notice of any pending or threatened compulsory acquisition, execution, attachment, levy or similar proceedings affecting any LM Mortgaged Property that

may have a material adverse effect upon the value of that LM Mortgaged Property;

- (ii) no material portion of any LM Mortgaged Property has been damaged or destroyed;
- (I) (no litigation): except as disclosed in Schedule 7 (or as otherwise notified by the Borrower to the Financier following Financial Close), no litigation, administration proceedings or arbitration is pending against the Borrower or Custodian with respect to any LM Loan Document which would adversely affect the value or marketability of the LM Loan or the related LM Mortgaged Property;
- (m) (loan schedule information accurate): the data relating to the LM Loans and LM Loan Documents as set out in the LM Loan Schedule is accurate in all material respects;
- (n) (no notice of invalidity): except as disclosed in Schedule 7 (or as otherwise notified by the Borrower to the Financier following Financial Close) the Borrower has not received and is not aware of any notice, assertion, claim or demand, and is not otherwise aware, that:
 - (i) any LM Loan Document or any part thereof is subject to any impairment or is invalid or void; or
 - (ii) LM Collateral Security is or might be invalid or is void or voidable whether as a preference or fraudulent transfer or otherwise;
- (o) (no rights of set-off or defence): except as disclosed in Schedule 7 any amounts payable under any LM Loan Document will be payable without set-off or counterclaim and no Customer has any valid defence or counterclaim to payment in full of the LM Loan and all amounts shown in the LM Loan Schedule are due and owing with respect to that LM Loan;
- (p) (no consumer credit loan): no LM Loan is regulated by the Credit Legislation and nothing in an LM Loan Document shall be construed as securing or purporting to secure the payment of any moneys or the performance of any obligation by a Customer (whether as principal debtor, guarantor, surety or otherwise) under or in connection with any credit contract or mortgage to which any Credit Legislation applies; and
- (q) (use): from Financial Close (except in respect of a LM Loan funded by cash available to the Borrower under clause 6.2(a)(vii)) any amount drawdown by a Customer under a LM Loan will not be used by the Customer to discharge any liability owed to an Affiliate.

15.6 Representations and warranties repeated

Each Repeating Representation will be repeated on the date of each Utilisation Notice, each Utilisation Date and the last day of each Interest Period, in each case, with reference to the facts and circumstances then subsisting.

16. Undertakings

16.1 Duration and benefit

The undertakings in this agreement are given for the benefit of the Financier and will remain in force on and from the date of this agreement for the period any amount is or may be outstanding under any Finance Document or any Commitment is in force.

16.2 Information and Custody Audits

- (a) (Financial information) The Borrower will provide to the Financier.
 - (i) as soon as practicable and in any event no later than 90 days after the end
 of each Financial Year, a copy of the audited Accounts for the LM
 Mortgage Trust for that Financial Year, certified as correct by an auditor
 approved by the Financier;
 - (ii) as soon as practicable and in any event no later than 45 days after the end of the first half of each Financial Year, a copy of the unaudited Accounts for the LM Mortgage Trust for that half year and, if certified as correct by an Authorised Officer of the Borrower; and
 - (iii) promptly, any further information regarding its financial condition and business operations or the financial condition and business operations of the Borrower as the Financier, acting reasonably, requires from time to time.
- (b) (Compliance with Accepted Accounting Practices) The Borrower will ensure that the Accounts delivered to the Financier under clause 16.2(a):
 - (i) are prepared in accordance with applicable laws and Accepted
 Accounting Practices consistently applied except to the extent disclosed in them;
 - (ii) if audited, give a true and fair view of or, if unaudited, fairly represent the financial condition of the LM Mortgage Trust for the period to which those Accounts relate and of the results of the operations of the LM Mortgage Trust for that accounting period; and
 - (iii) disclose all material Financial Liabilities and other material contingent liabilities of the LM Mortgage Trust.
- (c) (Custody Information) The Borrower will promptly upon receipt deliver to the Financier:
 - (i) a copy of any report given by the SCO to the Borrower pursuant to the Custody Agreement;
 - (ii) a copy of the Auditor's report given to the Borrower pursuant to the Custody Agreement including any assessment of compliance as required by section 601HG(3)(c) of the Corporations Act.
- (d) (LM Loan Schedule) The Borrower will provide to the Financier within 5 Business

 Days of the end of each month an updated and current LM Loan Schedule in respect
 of the preceding month.

- (e) (Know your client procedures) The Borrower will promptly provide to the Financier any documentation and other evidence which is from time to time:
 - required by the Financier to comply with its know your client procedures; and
 - which has not already been made available to the Financier by the Borrower.
- (f) (Provision of further information) The Borrower will provide to the Financier:
 - (i) (Special resolutions) promptly after the date on which the notice is issued, a copy of any notice calling an extraordinary general meeting of it or, in the case of the Responsible Entity, the members of the LM Mortgage Trust or proposing any resolutions of persons entitled to vote, in any case, for consideration by its shareholders or, in the case of the Responsible Entity, the members of the LM Mortgage Trust. The Borrower undertakes that the Financier or a representative of the Financier will be entitled to attend a meeting of members of the LM Mortgage Trust;
 - (ii) (Reports to members) in the case of the Responsible Entity only, promptly after the date on which the report or other item is issued, a copy of all reports, accounts, notices and circulars issued by it to any of its shareholders of the Responsible Entity or, in the case of the Responsible Entity, the beneficiaries of the LM Mortgage Trust; and
 - (iii) (Evidence of compliance) as and when required by the Financier, evidence to the reasonable satisfaction of the Financier, acting reasonably, proving that the Borrower is complying with the Obligations and, if applicable, taking steps to remedy any subsisting Event of Default or Potential Event of Default.
- (g) (Notification of certain events) The Borrower will immediately notify the Financier in writing if it becomes aware of the occurrence of:
 - (i) (Event of Default) any Event of Default or Potential Event of Default,
 - (ii) (Litigation) any litigation, arbitration, criminal or administrative proceedings which are current, pending or to its knowledge threatened relating to it or any Trust Property and which involve, in aggregate, claims against the Borrower in excess of \$500,000 (or the equivalent in any other currency) or which, if determined adversely, could have a Material Adverse Effect;
 - (iii) (Authorised persons) any change in its Authorised Officers, giving evidence of the authority of any newly appointed Authorised Officer as required under paragraph (a) of the definition of "Authorised Officer" in clause 1.1:
 - (iv) (LM Loans Litigation) any litigation, administrative proceedings or arbitration against the Borrower or Custodian with respect to any LM Loan Document which could adversely affect the value or marketability of the LM Loan or the related LM Mortgaged Property;
 - (v) (LM Loan Document Void) any notice, assertion, claim or demand that any LM Loan Document or any part thereof is invalid or void;

- (vi) (Competing Interests) any notice from any person that it claims to have an Encumbrance ranking in priority to or equal with the Encumbrance held by the Borrower or Custodian and constituted by any LM Mortgage;
- (h) (Portfolio information) The Borrower must no later than the last day of each month provide to the Financier an electronic listing containing:
 - a list of all of the offices at which the LM Loan Documents are retained, showing the street address and telephone number of the relevant office;
 - (ii) the name and address of the Customer under each LM Loan;
 - (iii) the street address of the Land which is subject of the LM Mortgage relating to each LM Loan;
 - (iv) a brief description of the LM Collateral Security; and
 - (v) such other information in respect of the LM Loans which is reasonably requested by the Financier.
- (i) (Document custody audit report) The Borrower must retain the Auditor to conduct periodic reviews (at the times reasonably requested by the Financier in respect of the Custodian's role as Custodian of the LM Loan documents) and, in doing so:
 - (i) the Auditor must review:
 - A. the Custodian's procedures adopted by the Custodian; and
 - B. the accuracy of information in respect of the LM Loan obtained on electronic listing provided to the Financier pursuant to clause 16.2(h) and the Borrower must instruct the Auditor that its review should consist of reporting whether in respect of the LM Loan Documents:
 - the LM Loan Documents are capable of identification and are distinguishable from the other assets of the Custodian or Borrower;
 - controls exist such that those LM Loan Documents may not be removed or tampered with except with appropriate authorisation; and
 - an appropriate tracking system is in place such that the location of those LM Loan Documents can be detected at any time;
 - (ii) the Borrower must instruct the Auditor to confirm, after having conducted the reviews referred to in clause 16.2(h) and this clause 16.2(h), the accuracy of information in respect of the electronic listing provided to the financier pursuant to clause; and
 - (iii) the Borrower must instruct the Auditor to prepare a document custody audit report immediately after delivery of the electronic listing referred to in clause 16.2(h) and on 31 March and 30 September of each year thereafter.

16.3 Restrictions on Disposals

- (a) (Restriction on Encumbrances) The Borrower will not create, purport or attempt to create or permit to exist, any Encumbrance however ranking over any of its assets.
- (b) (Transactions similar to security) The Borrower will not:
 - sell or otherwise Dispose of any of its assets on terms which permit or require those assets to be leased to or re-acquired by the Borrower;
 - sell or otherwise Dispose of any of its receivables on recourse terms or securities its receivables; or
 - (iii) purchase any asset on terms providing for a retention of title by the vendor or on conditional sale terms or on terms having a similar substantive effect.
- (c) (Disposals) The Borrower will not enter into or make any Disposal of any other part of its assets or undertaking, unless the Disposal is a Permitted Disposal. The Financier will release from the Securities any asset Disposed of as permitted by this clause 16.3(c).
- (d) (Permitted Encumbrances) clauses 16.3(a), 16.3(b) and 16.3(c) do not apply to Permitted Encumbrances.

16.4 Liabilities

- (a) (Financial Liabilities of the Borrower) The Borrower will not incur any Financial Liabilities other than the Permitted Financial Liabilities.
- (b) (Restriction on guarantees) The Borrower will not enter into any bond, guarantee or indemnity in favour of any person other than:
 - (i) under the Finance Documents; or
 - (ii) as permitted under clause 16.4(a).
- (c) (Treasury Transactions) The Borrower will ensure and procure that:
 - (i) any loss or liability in respect of any Treasury Transactions is and will be subordinated and subject to right of payment and repayment to all monies payable by the Borrower to the Financier under the Finance Documents; and
 - (ii) the maximum principal amount of all Treasury Transactions does not exceed \$10,000,000.
- (d) (Financial Liabilities of the Responsible Entity) The Responsible Entity in its personal capacity will not incur any Financial Liabilities other than the Permitted Responsible Entity Financial Liabilities (but for the avoidance of doubt, this does not extend to any Financial Liabilities that LM Investment Management Limited may have as trustee or responsible entity of any other trust scheme).

16.5 Financial Accommodation

· From Financial Close the Borrower shall not provide any financial accommodation to any person unless:

42

- (a) no Default or Potential Event of Default is subsisting; and
- (b) such financial accommodation is:
 - (i) provided:
 - A. from moneys available to the Borrower pursuant to clause 6.2(a)(vii); or
 - B. in respect of moneys available to the Borrower from Net
 Proceeds because the Trust Bank Accounts are not less than
 the Alternative Minimum Cash Balance and provided the
 Financier is satisfied the New Lending Criteria is or will be
 satisfied (or the Financier otherwise consents);
 - (ii) provided in the ordinary course of the Borrower's business and on arm's length and on commercial terms;
 - (iii) not provided to any employee, director, shareholder of the Responsible Entity or to any Related Body Corporate of the Responsible Entity; and
 - (iv) secured by a first ranking mortgage over Land which is the only mortgage over the relevant Land.

The Financier acknowledges that this clause does not apply to any Existing Financial Accommodation.

16.6 Environmental matters

- (a) (Environmental Law) The Borrower will:
 - (i) obtain all Authorisations required under any Environmental Law in order for it to carry on business;
 - (ii) ensure that each of those Authorisations is valid and subsisting;
 - (iii) comply in all material respects with the terms and conditions of those Authorisations;
 - (iv) ensure that no act or omission occurs in connection with its business which would:
 - A. entitle a Government Authority to make a declaration, or to issue a notice, direction or order, under any Environmental Law;
 - B. give rise to a nuisance; or
 - C. give rise to any Contamination the investigation and remediation of which would require any material expenditure; and
 - (v) ensure that it:
 - A. has in place an appropriate environmental management system for the operation of its business; and

- B. monitors and audits compliance with all Environmental Laws which are relevant to its business, whether as part of its environmental management system or otherwise.
- (b) (Notice) The Borrower will, promptly on receipt, notify the Financier of any claim, notice or other communication received by it in respect of any breach or alleged breach of, or any liability under, any Environmental Law.
- (c) (Indemnity) The Borrower will indemnify the Financier and the Delegates (together the Indemnified Parties) against any Cost, loss, damage, claim, demand or action (including any legal costs on a full indemnity basis) suffered or incurred by an Indemnified Party (except, in the case of any Indemnified Party, if caused by its own gross negligence or wilful default) which arises directly or indirectly from:
 - (i) any actual or alleged breach of any Environmental Law; or
 - (ii) any Contamination associated with the past or present facilities or operations.

16.7 Insurance

- (a) (Insurances) The Borrower will insure its business and assets, and keep its business and assets insured, with insurance companies approved by the Financier against any risks and liabilities to which the business of the Borrower is exposed and to the extent as is commercially reasonable/appropriate having regard to those risks and liabilities. Without limiting this obligation, the Borrower must effect and maintain the following insurances:
 - (i) insurance of the Security Property against loss, theft, damage or destruction;
 - (ii) insurance against business interruption, loss of profits, consequential and advanced consequential loss, product liability, professional indemnity (or, if appropriate, civil liability), pollution, Contamination, loss of key executives and public liability;
 - (iii) insurance required by any law; and
 - (iv) any other specific insurance.
- (b) (Terms) No later than 30 days following Financial Close, the Borrower will unless it is not commercially practicable to obtain insurance on such terms:
 - (i) ensure that all Insurances (other than employers' liability, professional indemnity and workers' compensation insurance policies) will have the Financier named as an insured and as the sole loss payee;
 - (ii) ensure that all Insurances (other than employers' liability, professional indemnity and workers' compensation insurance policies) include clauses in which the insurer agrees:
 - to waive all rights of subrogation or action that it may have or acquire against all or any of the persons comprising the insured:
 - B. that the term "insured" applies to each of the persons comprising the insured as if a separate policy of insurance had

been issued to each of them (subject always to the overall sum insured not being increased as a result); and

- that any non-disclosure, breach of any duty or act or omission by one insured under any Insurance does not prejudice the right of any other insured to claim under that Insurance;
- (iii) ensure that all Insurances will where applicable, insure the Financier's insurable interests; and
- (iv) ensure that all Insurances will be for the amounts, cover the risks and contain the terms and conditions as the Financier, acting reasonably, requires unless it is not commercially practicable to obtain insurance on such terms in respect of a particular loss.
- (c) (Full replacement value) All Insurances against loss, theft, damage or destruction of the Security Property will be for the full replacement value of the Security Property from time to time.
- (d) (Policies) The Borrower must deliver to the Financier:
 - (i) certificates of currency and receipts for the payment of each premium immediately on renewal of any Insurance; and
 - (ii) any other documents relating to the Insurances (including the policies, endorsement slips and all other money payable in respect of any policy) immediately upon request by the Financier.
- (e) (Maintenance of Insurances etc) The Borrower will:
 - (i) duly and punctually pay or cause to be paid all premiums and other money payable under, and perform and observe the terms of, all Insurances; and
 - (ii) ensure that every policy of Insurance contains an agreement by the insurer that, notwithstanding the lapse of that policy (except because of expiration in accordance with its terms) or any right of cancellation of the insurer or any cancellation by the Borrower (whether voluntary or involuntary), that policy will continue in force for the benefit of the Financier for at least 30 days after notice of cancellation has been sent by certified mail to the Financier and that no reduction in limits or coverage in that policy in whole or part or other material change to the terms of that policy will be effected except with the Financier's prior consent.
- (f) (No liability) The Financier will not incur any liability to any person arising out of a failure by any insurer to meet any claim, for any reason, under any Insurance.
- (g) (Option as to payments) Except as provided in clause 6.3, each sum received under any Insurance will be applied towards the replacement or repair of the property insured or to discharge the relevant liability or to make good the relevant loss covered by the sum received.
- (h) (LM Mortgaged Properties) The Borrower must:
 - ensure and procure that the LM Mortgaged Property is insured in accordance with the provisions of each LM Loan Agreement; and

(ii) deliver to the Financier upon request any certificates of currency and other documents relating to any insurance in respect of a LM Mortgaged Property.

16.8 Authorisations

The Borrower will:

- (a) (Obtain) obtain all Authorisations required:
 - (i) in connection with the execution, delivery and performance of each Finance Document to which it is a party, the transactions contemplated by those Finance Documents and the legality, validity and enforcement of each of those Finance Documents:
 - (ii) for it to carry on business and which are material; and
 - (iii) in the case of the Responsible Entity, for the LM Mortgage Trust to be established;
- (b) (Renew) on or before the time and in the manner prescribed by any law, procure the renewal of each of its Material Authorisations and pay or cause to be paid the fees and other sums required in respect of each of its Material Authorisations (including in connection with any renewal) within the time allowed and in the manner prescribed by any applicable law;
- (c) (Production) on request by the Financier, provide to the Financier a copy of each
 of its Material Authorisations;
- (d) (No cancellation) not do, fail to do or consent to any act, matter or thing as a result of which any of its Material Authorisations is or may be suspended, forfeited, withdrawn, cancelled, refused or rendered void or by which it is disqualified permanently or temporarily from receiving or continuing to hold any of its Material Authorisations; and
- (e) (No transfer) not surrender or, without limiting clause 16.3(c), consent to the transfer of any of its Material Authorisations.

16.9 Security Property

The Borrower will:

- (a) (Good repair) maintain the Security Property in good repair and condition and in good working order;
- (b) (Operation) diligently operate its assets in accordance with Good Industry Practice;
- (c) (Not to prejudice) not do, fail to do or consent to any act, omission or thing as a result of which any part of the Security Property becomes or could become liable to surrender, forfeiture or cancellation or becomes or could become prejudiced in any manner or the value of any Security as a security to the Financier becomes or could become lessened;
- (d) (Protection of Security Property) at the request of the Financier take or defend all legal proceedings that the Financier, acting reasonably, considers necessary or desirable for the preservation, protection or recovery of the Security Property; and

- (e) (Documents of title and other securities) upon request by the Financier, lodge with the Financier all:
 - (i) certificates, scrip and other indicia of title or interest in any Marketable Securities forming part of the Security Property;
 - (ii) negotiable instruments other than cheques forming part of the Security Property;
 - (iii) certificates of title to land forming part of the Security Property; and
 - (iv) other documents of title to the Security Property.

16.10 Cash management and application

- (a) The Borrower undertakes that it will ensure and procure that:
 - (i) within 30 days of receipt all Receivables (except for amounts payable by Customers under LM Loan Agreements which will be paid directly to the Managed Cash Account) will, where possible, be paid directly into the Managed Cash Account or upon receipt be paid into the Managed Cash Account;
 - (ii) Net Proceeds of the sale of any LM Mortgaged Property will be promptly paid directly into the Managed Cash Account and that such Net Proceeds will be applied in accordance with clause 6.2.
- (b) If the Responsible Entity is a responsible entity of another trust, it will not use or distribute any Trust Property whether by way of loan or otherwise to that trust or any Related Body Corporate or Affiliate of the Borrower.
- (c) Following Financial Close the Borrower and the Financier shall consult with each other in good faith to ascertain the most efficient manner in which to operate the Managed Cash Account.

16.11 Operation of Managed Cash Account

- (a) The Borrower must ensure and procure that the Financier has internet access to view the Managed Cash Account.
- (b) The Borrower must maintain the Minimum Cash Balance in the Managed Cash Account. If the balance in the Managed Cash Account is less than (or following any proposed withdrawal, will be less than) the Minimum Cash Balance, the Borrower must not withdraw funds from the Managed Cash Account unless such withdrawal is in accordance with clause 16.11(c) to (g) or with the prior written consent of the Financier.
- (c) Provided there is no subsisting Event of Default or Potential Event of Default or an Unpaid Sum, the Borrower may request the Financier to release an amount from the Managed Cash Account to be used in connection with Treasury Transactions, and the Financier will consent to that request provided:
 - (i) the Financier is satisfied that the amount to be withdrawn will be used by the Borrower to meet the Borrower's (and not any feeder funds) cash cover obligations to a Hedge Provider under a Treasury Transaction; and

- (ii) aggregate draws for this purpose do not exceed \$2,000,000 from the Managed Cash Account or the Alternative Minimum Cash Balance which is ever applicable; and
- (iii) the amount withdrawn or any greater amount is deposited to the Managed Cash Account if the Borrower is no longer required to maintain that cash cover.
- (d) Provided there is no subsisting Event of Default or Potential Event of Default or an Unpaid Sum, the Financier will consent to withdrawals from the Managed Cash Account provided such amounts are required in respect of and are applied for one of the purposes specified in clause 6.2(a)(vi) (Permitted Drawings) and:
 - (i) subject to clause 16.11(e), the Permitted Drawings do not exceed \$500,000 in total (Maximum Drawings); and
 - (ii) as soon as practicable after the Maximum Drawings have been made, the Borrower will provide to the Financier a report showing details of the amounts withdrawn and how they were applied by the Borrower.
- (e) If:
 - (i) the Financier has been provided with the report referred to in clause 16.11(d)(ii), together with any other information reasonably requested by the Financier; and
 - (ii) the Financier has notified the Borrower that it has no objections to the report and the disbursement of the Permitted Drawings.
 - (iii) then the current level of Permitted Drawings shall be reduced to zero and the Borrower shall be able to make further withdrawals from the Managed Cash Account in accordance with the provisions of clause 16.11(d).
- (f) The Financier will consent to withdrawals from the Managed Cash Account even if the withdrawal would result in the credit balance of the Managed Cash Account being less than the Minimum Cash Balance provided that:
 - (i) the amount withdrawn is applied for any of the purposes specified in clause 6.2(a)(i) to 6.2(a)(vi) inclusive; and
 - (ii) within 90 days from the date the credit balance of the Managed Cash Account is less than the Minimum Cash Balance the Borrower will deposit such monies to ensure that the credit balance of the Managed Cash Account is not less than the Minimum Cash Balance.
- (g) Provided there is no subsisting Event of Default or Potential Event of Default or an Unpaid Sum, the Borrower may, with the Financier's consent, withdraw amounts from the Managed Cash Account provided that such amounts are required for and are applied in respect of those Qualifying Capital Improvements even if the withdrawal would result in the credit balance of the Managed Cash Account being less than the Minimum Cash Balance.
- (h) If the Financier agrees that the Minimum Cash Balance may in any month be less than \$6,000,000, the Borrower will ensure that within 90 days the Minimum Cash Balance will be restored.

- (i) Notwithstanding anything to the contrary in this clause 16.11, any withdrawal from the Managed Cash Account that is or is intended to be applied for the purpose or in the manner described in clause 16.11(d) to (f) and that is for a single item of expenditure in excess of \$250,000 shall require the prior written consent of the Financier.
- (j) Provided there is no subsisting Event of Default or Potential Event of Default or an Unpaid Sum, the Financier will consent to the Borrower withdrawing amounts from the Managed Cash Account, provided that the Minimum Cash Balance is maintained in the Managed Cash Account following such withdrawal and:
 - (i) the provisions of clause 6.2, 6.3 and 6.5 have been, and will at the next Monthly Interest Payment Date be, complied with; and
 - (ii) the withdrawal would not result in a breach of any other provision of this clause 16.11.

16.12 Receivership of Customers

(a) The Borrower will ensure that the terms of appointment of any Controller to a Customer will contain the following provisions:

"The Appointee may, without prejudice to any or all other rights and remedies it may have under this deed or applicable law, terminate this deed by giving 7 days written notice to the Controller. The Controller must formally retire by executing an appropriate deed of retirement prior to the end of the notice period given by the Appointor".

(b) The Borrower agrees that if the Financier in its absolute discretion is not satisfied with the identity, conduct or otherwise of any Receiver, Receiver and Manager or Controller appointed by the Borrower to a Customer, then the Borrower will, upon notice from the Financier, terminate the appointment of that Receiver and after consultation with the Financier appoint a Receiver which is acceptable to the Financier.

16.13 Custodian Arrangements

The Borrower shall:

- (a) monitor the performance of the Custodian in accordance with the methods and standards set out in Schedule 3 of the Custody Agreement;
- (b) promptly advise the Financier if it believes the Custodian is in breach of its obligations under the Custody Agreement; and
- (c) upon receipt provide the Financier with the reports and statements described in Schedule 4 of the Custody Agreement at the intervals mentioned in that Schedule.

16.14 General undertakings

- (a) (Corporate existence) The Responsible Entity will do all things necessary to maintain its corporate existence.
- (b) (Change of business) The Responsible Entity will not make or threaten to make any material change in the nature of its business as conducted at the date of this agreement or, if later, the date it became a party to this agreement.

- (c) (Mergers, etc) The Borrower will not enter into any merger or consolidation or make any acquisition of any other person or business or incorporate or any interest in a Joint Venture or acquire any shares or other Marketable Securities.
- (d) (Business of the Borrower) The Borrower will not carry on any business other than investing in a diversified portfolio of Australian registered first mortgages over commercial, residential, industrial, retail and vacant land, interest bearing cash investments and at call securities.
- (e) (Amendments to constitutional documents) Neither the Borrower nor the Responsible Entity will not agree to amend its Constitution or the LM Mortgage Trust Constitution in any way which would be likely to adversely affect the Financier's interests under the Finance Documents.
- (f) (Compliance with constitutional documents) The Borrower will comply with the LM Mortgage Trust Constitution and the Responsible Entity will comply with its Constitution.
- (g) (Comply with laws) The Borrower and the Responsible Entity will comply with all laws applicable to it (except to the extent that non-compliance is not material).
- (h) (Taxes) The Borrower and the Responsible Entity will:
 - file all Tax returns and other information required to be filed by it with any relevant Tax authority to ensure that it complies with all of its obligations to pay Tax;
 - (ii) provide to the Financier all Tax returns and information referred to in clause 16.14(h)(i) above immediately after such Tax returns and information has been filed; and
 - (iii) pay all Taxes due and payable by it by the due date (except to the extent that payment of any Taxes is being contested in good faith.
- (i) (Auditors) On request from the Financier, the Borrower will instruct its auditors to:
 - (i) discuss the Borrower's financial position and operations with the Financier; and
 - (ii) disclose to the Financier any information which the Financier, acting reasonably, requests regarding the financial condition and operations of the Borrower and the Customers' obligations under the LM Loan Documents including the status of any actual or threatened enforcement or recovery actions.
- (j) (Access) On reasonable notice being given by the Financier, the Borrower will procure that any Delegate or professional advisor of the Financier is allowed (at the Financier's expense) to enter any land or buildings subject to a LM Mortgage or owned or occupied by the Borrower and to have access during normal business hours to the assets, books and records of the Borrower or Customer and to inspect the same in each case for the purposes of monitoring compliance with the Finance Documents.
- (k) (Application of proceeds) The Borrower undertakes to use each Advance only in accordance with clause 2.2.

- (I) (Commercial dealings) The Borrower will not deal in any way with any person except at arm's length in the ordinary course of business for valuable commercial consideration.
- (m) (Bank accounts) The Borrower will not (either in its or the Customer's name) maintain any bank account other than the Managed Cash Accounts and the Trust Bank Accounts unless the prior written consent of the Financier is first obtained.
- (n) (Distribution) While an Event of Default or Potential Event of Default is subsisting or will result from the making of the Distribution or there is an Unpaid Sums, not declare or make any Distributions without the prior written consent of the Financier (which may be withheld in its absolute discretion).

16.15 LM Mortgage Trust undertakings

The Borrower undertakes that it will:

(a) (Trust Documents)

- (i) other than to the extent that it would cause an Event of Default or a
 Potential Event of Default, or be a breach of an undertaking under a
 Finance Document, comply with all of its obligations as responsible
 entity under each Trust Document and at law and take all action
 reasonably required to enforce and preserve its rights under each Trust
 Document;
- (ii) ensure that no waiver, revocation, amendment or variation of any term of any Trust Documents is made, whether formally or by conduct, without the prior written consent of the Financier other than:
 - A. an amendment or variation which is minor or immaterial or is a change of a technical nature which would not be likely to adversely affect the interests of the Finance Parties under the Finance Documents; or
 - B. in respect of the Compliance Plan, any modification required to be made by ASIC under section 601HE(2) of the Corporations Act);
- (iii) not take nor fail to take and will procure that the Custodian as custodian of the LM Mortgage Trust does not take nor fails to take any action which would or could result in the termination, repudiation, rescission or revocation of any Trust Document to which it is a party; and
- (iv) comply with its Compliance Plan;

(b) (Custodian)

- (i) procure that the Custodian complies with (and performs) all of its obligations as custodian of the LM Mortgage Trust under each Trust Document to which it is a party and at law;
- (ii) not to instruct, or consent to, the Custodian doing anything which would constitute or result in an Event of Default or Potential Event of Default; and

- (iii) not to do anything (or permit anything to be done) which could (except with the prior written consent of the Financier) effect or facilitate the resignation, retirement, removal or replacement of the Custodian as custodian of the LM Mortgage Trust or the appointment of an additional custodian of the LM Mortgage Trust;
- (c) (General) not to do anything (or permit anything to be done) which:
 - (i) results or may result in the deregistration of the LM Mortgage Trust under section 601PB or Part 5C.10 of the Corporations Act;
 - (ii) restricts or limits the Borrower's:
 - A. Right of Indemnity out of the Trust Property; or
 - B. power under the LM Mortgage Trust Constitution or ability to observe its obligations under the Finance Documents to which it is a party;
 - (iii) restricts or limits the Financier's right of subrogation in connection with the Responsible Entity's Right of Indemnity or lien over the Trust Property;
 - (iv) causes its lien over any Trust Property to be restricted or limited or not to have priority over the rights of the beneficiaries of the LM Mortgage Trust; and
 - (v) effects or facilitates:
 - A. the termination of the LM Mortgage Trust or the termination, rescission or revocation of any Trust Documents to which it is a party:
 - B. the resettlement of any Trust Property; or
 - C. (except with the prior written consent of the Financier) the resignation, retirement, removal or replacement of LM Investment Management Limited as responsible entity of the LM Mortgage Trust; or
 - (vi) determines, alters, shortens or fixes the vesting date of the LM Mortgage
 Trust under the LM Mortgage Trust Constitution;
- (d) (Guarantee) not provide or incur any liability or indebtedness for or in respect of any guarantee, indemnity, letter of credit or any other similar assurance in connection with any Financial Liability (other than Financial Liability arising under the Finance Documents);
- (e) (Determination of Trust) notify the Financier immediately in writing if:
 - (i) the LM Mortgage Trust is determined or for any other reason ceases to exist, or it is required or directed by any of the beneficiaries of the LM Mortgage Trust to do any act or thing in relation to the LM Mortgage Trust or the trust funds of the LM Mortgage Trust; and
 - it proposes to retire or it otherwise ceases to be the responsible entity or trustee of the LM Mortgage Trust;

(f) (Responsible Entity / Manager fees)

- (i) except with the prior written consent of the Financier, not receive, nor pay (as the context may require) any management fees or other amounts in the nature of management fees in respect of its role as the responsible entity of the LM Mortgage Trust, unless at the time that the payment is to be made or received, no Event of Default is subsisting and no fees, charges or interest or principal payable under or in connection with the Finance Documents are due and payable and remain to be paid and the Borrower undertakes to procure the recovery of all such sums paid in breach of this clause immediately upon request of the Financier, and to apply such sums in satisfaction of the relevant fees, charges and / or interest then payable under or in connection with the Finance Documents; and
- (ii) the Borrower, in its role as Responsible Entity agrees:
 - to waive any claims for management fees which may have accrued but are unpaid as at Financial Close;
 - B. that any claim for management fees is and will be subordinated and subject in right of payment and repayment to all monies payable by the Borrower to the Financier under the Finance Documents; and
 - C. that it will not pay any fees to any agent or Service Provider if an Event of Default is subsisting and any amount due and payable to the Financier under the Finance Documents is unpaid; and
- (g) (Hedging) not alter the hedging strategy disclosed to the Financier and will not amend or vary that strategy without the Financier's prior written consent.

16.16 LM Loan Document undertakings

The Borrower must ensure and procure, unless the Financier otherwise consents, that it or the Custodian (as applicable) will in respect of each LM Document:

- (a) punctually comply with and perform its obligations under each LM Loan Document:
- (b) subject to all applicable laws, licences, approvals and authorisations regulating the enforcement of the Borrower's and/or the Custodian's rights under the LM Loan Documents, take all commercially reasonable action (including as reasonably required by the Financier) to enforce, and in the event of such enforcement:
 - (i) while an Event of Default subsists, it must exercise those Powers in accordance with the directions (if any) of the Financier and, in relation to any dispute, it must act in accordance with those directions unless it believes in good faith that to do so will cause it to breach any law; and
 - (ii) at all other times it must consult with the Financier in relation to the exercise of those powers or any disputes below:
- (c) not amend, vary, replace or consent or agree to any change in, or not waive, vary or replace any requirement of;

- (d) not terminate, rescind or discharge (except by performance);
- (e) not assign, transfer or novate any interest in; or
- (f) not waive, (expressly or impliedly) or extend or grant time or indulgence in respect of, or do anything which would or would be likely to release another party from any obligation under, and
- (g) (Caveat) upon request by the Financier, at its own cost and expense, use its best endeavours to obtain a consent caveat from each Customer in respect of the Land subject to each LM Mortgage.

16.17 Financier's reputation

- (a) The Borrower will at all times conduct itself in a way which in no way damages or, in the Financier's reasonable opinion, may damage or adversely affect the Financier's reputation.
- (b) Without limiting clause 16.17(a) above, and other than in the case of legal proceedings or similar action, the Borrower undertakes to refrain at all times during and subsequent to the termination of this agreement from making or issuing any derogatory remark or statement regarding the Financier, its personnel or its business.
- (c) The Borrower will use its best endeavours to procure that the Service Provider and any of its agents and attorneys will conduct themselves in a manner which is consistent with clauses 16.17(a) and 16.17(b) above.

16.18 Customer Payments

The Borrower will ensure and procure that all payments made by a Customer under an LM Loan Agreement are paid directly by the Customer to the Managed Cash Account and shall as soon as practicable following Financial Close give notice to the Customers of such requirement and the specific details of the Managed Cash Account into which all payments are to be made.

16.19 LM Mortgaged Property Taxes

The Borrower undertakes:

- (a) promptly following Financial Close it will pay all arrear Taxes, land taxes, rates, body corporate fees, and other statutory charges (LM Mortgage Charges) payable in respect of any property the subject of a LM Mortgage; and
- (b) to ensure and procure that within:
 - (i) 30 days of Financial Close it will apply not less than \$1,600,000 of the Advance towards satisfying LM Mortgage Charges in arrears and
 - (ii) no later than 90 days after Financial Close all outstanding LM Mortgage Charges, which are not already the subject of a payment plan and that plan is being complied with, will be paid up to date and no LM Mortgage Charges will be in arrears or unpaid on the due date through to the Termination Date.

17. Default and termination

17.1 Events of Default

Each of the events set out in this clause 17.1 is an Event of Default, whether or not the cause is beyond the control of the Borrower or any other person:

- (a) (Failure to pay) the Borrower does not pay any amount payable by it at or before the due time on the due date and in the manner specified in any Finance Document unless that default is caused by a technical or administrative error by a bank or financial institution in the transmission of funds and is not remedied within 2 Business Days of its occurrence;
- (b) (Failure to comply) the Borrower, Custodian or Service Provider defaults in performing or observing any provision of any Finance Document (other than a provision requiring the payment of money as contemplated by clause 17.1(a) or any other provision specifically referred to in this clause 17.1), and if that default is capable of remedy it has not been remedied by the later of:
 - (i) 10 Business Days after the date of its occurrence; and
 - (ii) 10 Business Days after the date on which when the Borrower, Custodian or Service Provider becomes aware of the failure;
- (c) (Untrue warranty) any representation, warranty or statement made, repeated or taken to been made or repeated in any Finance Document or in any document, of any nature, issued under any Finance Document is proved to be untrue in any material respect when made or repeated or taken to be made or repeated, as the case may be (a Misrepresentation) and if the circumstances causing the Misrepresentation are capable of remedy those circumstances have not been remedied within the later of:
 - (i) 10 Business Days after the date of the occurrence of the Misrepresentation; and
 - (ii) 10 Business Days after the date of the Borrower, Custodian or Service Provider becomes aware of the occurrence.
- (d) (Breach of undertaking) the Borrower, Custodian or Service Provider breaches any undertaking given at any time to the Financier or fails to comply with any condition imposed by the Financier in agreeing to any matter (including any waiver);
- (e) (Default under other transactions)
 - (i) any Financial Liability of the Borrower exceeding \$500,000 becomes or becomes capable of being declared, prematurely due and payable as a result of a default, an event of default or a review event, however described;
 - (ii) any Financial Liability of the Borrower or any sum payable in respect of it is not paid when due or within any originally applicable grace period;
 and
 - (iii) any Encumbrance relating to any asset of the Borrower becomes capable of enforcement;

- (f) (Event of Insolvency) an Event of Insolvency occurs in relation to the Borrower, the Custodian or the Service Provider;
- (g) (Cessation of business) the Borrower, the Service Provider or the Custodian ceases, or threatens to cease, to carry on all or a material part of its business;
- (h) (Void or voidable) any Finance Document is, becomes or is claimed by the Borrower. Custodian or Service Provider to be invalid, void, voidable, unenforceable or of limited force and effect, either in whole or in part;
- (i) (Security) any Security does not create the Encumbrance it purports to create;
- (j) (Repudiation) the Borrower, the Custodian or Service Provider repudiates, or evidences an intention to repudiate, any Finance Document;
- (k) (Change in control) effective control of the Responsible Entity is altered from that subsisting at the date of this agreement. For the purpose of this clause 17.1(k) a person or persons has effective control of the Borrower if that person or those persons (acting jointly):
 - (i) control the composition of the board of directors of the Responsible Entity;
 - (ii) are in a position to cast, or control the casting of, more than one-half of the maximum number of votes that might be cast at a general meeting of the Responsible Entity; or
 - (iii) hold more than one-half of the issued share capital of the Responsible
 Entity excluding any part of that issued share capital which carries no
 right to participate beyond a specified amount in the distribution of either
 profits or capital;
- (l) (Illegality) at any time it is unlawful for the Borrower, the Custodian or Service Provider to perform any of its obligations under any Finance Document;
- (m) (Breach of LM Mortgage Trust provisions)
 - the Responsible Entity defaults in performing or observing any provision of clause 16.15;
 - (ii) the Responsible Entity ceases to be the sole trustee of the LM Mortgage Trust; and
 - (iii) the LM Mortgage Trust terminates or the capital the LM Mortgage Trust vests in any beneficiary of the LM Mortgage Trust or any other person (other than where it vests solely as a result of a person becoming a beneficiary of the LM Mortgage Trust);
- (n) (Material Adverse Effect) any event or series of events whether related or not occurs which has or could reasonably have a Material Adverse Effect;
- (o) (Auditors) the Auditor's appointment is terminated for whatever reason and is not replaced by an Approved Auditor;
- (p) (Borrower ceases to be responsible entity) the Responsible Entity retires or otherwise ceases to be the responsible entity or trustee of the LM Mortgage Trust or any resolution is passed at a meeting to replace LM Investment Management

Limited or other person as responsible entity or trustee of the LM Mortgage Trust, in any case without the Financier's prior written consent;

- (q) (Order for removal as responsible entity) an application or order is made in any court for:
 - (i) removal of the Responsible Entity as responsible entity or trustee of the LM Mortgage Trust; or
 - (ii) any Trust Property to be brought into court or administered by the court or under its control,

and the Financier is not satisfied with the replacement or additional or proposed replacement or additional responsible entity or trustee;

- (r) (Restrict right of indemnity) the Responsible Entity's Right of Indemnity from the Trust Property in respect of obligations incurred by the Responsible Entity under the Finance Documents is materially restricted or materially lost other than a restriction imposed under the Corporations Act;
- (s) (Termination of Trust)
 - (i) the LM Mortgage Trust terminates or the capital of the LM Mortgage Trust vests for any reason whatsoever; or
 - (ii) without the Financier's prior written consent, there occurs:
 - A. any resettlement of the Trust Property; or
 - B. any vesting of any Trust Property; or
- (t) (Deregistration of Trust) an application is made for deregistration of the LM Mortgage Trust or a notice is given under section 601PB(2) of the Corporations Act or the LM Mortgage Trust is or becomes deregistered under Part 5C.10 of the Corporations Act;
- (u) (Variation of Compliance Plan) any Compliance Plan is varied in a material respect without the prior written consent of the Financier and that variation has a Material Adverse Effect and that default is not remedied within 10 Business Days of notice from the Financier;
- (v) (Custody arrangements):
 - (i) the Custodian ceases to be the sole custodian of the LM Mortgage Trust unless replaced by a person and on terms acceptable to the Financier within 30 days:
 - (ii) the Custodian breaches the Custody Agreement and such breach is not remedied within 10 Business Days (or such other longer period as is consented to by the Financier, such consent not to be unreasonably withheld or delayed) of its occurrence.
- (w) (Meetings under 601FL and 601FM) in respect of the LM Mortgage Trust either:
 - (i) a meeting is called under section 601FL without the Financier's prior written consent; and/or

- (ii) a resolution is passed at a meeting called under section 601FM of the Corporations Act;
- (x) (Application for or appointment of temporary responsible entity) an application is made under sections 601FL(3) and/or 601FN for the appointment of a temporary responsible entity to the LM Mortgage Trust or a temporary responsible entity is appointed under section 601FP of the Corporations Act to the LM Mortgage Trust, except where such temporary responsible entity has been approved in writing by the Financier;
- (y) (Winding up of Trust) in respect of the LM Mortgage Trust:
 - a resolution of members is passed at a meeting is called under section 601NB of the Corporations Act to wind up the LM Mortgage Trust;
 - (ii) a notice is given under section 601NC of the Corporations Act; or
 - (iii) an application is made in any court under section 601FQ(5) and/or 601ND of the Corporations Act,

in any case without the Financier's prior written consent:

- (z) (Action under section 601MA) an action is commenced against the Responsible Entity under section 601MA of the Corporations Act and such action has, or could reasonably likely have, a Material Adverse Effect;
- (aa) (Minimum principal repayments) the Borrower fails to reduce the principal amount owing in respect of Advance:
 - (i) by an amount of \$7,000,000 on or before 31 December 2010;
 - (ii) by a further amount of \$11,000,000 on or before 30 June 2011;
 - (iii) by a further amount of \$9,000,000 on or before 31 December 2011; and
 - (iv) where the Termination Date is extended to 30 June 2013 by the Borrower validly exercising the option pursuant to clause 2.5 by a further amount of \$13,500,000 on or before 30 June 2012.
- (bb) (Key man) Peter Drake ceases any involvement in the conduct of the business of the Borrower, of the Service Provider and the Borrower's Affiliates;
- (cc) (Services Agreement):
 - the Services Agreement is terminated and not replaced by a services agreement acceptable to the Financier or the Borrower fails to satisfy the Financier that it can provide the services referred to in the Services Agreement;
 - (ii) the Service Provider ceases to be LM Administration Pty Ltd or such other person approved by the Financier;
 - (iii) the Service Agreement is amended, varied or supplemented without the Financier's prior written consent;
 - (iv) the Service Provider fails to provide the "services" (as defined in the Service Agreement); and

- (v) the Service Provider breaches the Service Agreement and such breach is not remedied with 10 Business Days of its occurrence.
- (dd) (New lending) the Alternative Minimum Cash Balance is not maintained while any new LM Loan provided under clause 16.5(b)(i)B remains outstanding.

(ee) (LM Loan Documents):

- any LM Loan Document or any part thereof is invalid, unenforceable or void:
- (ii) any LM Mortgage or LM Collateral Security is set aside as a preference, transaction at an under value or fraudulent transfer or otherwise:
- (iii) as a result of any litigation, administrative proceedings or arbitration made with respect to any LM Loan Document the value or marketability of the LM Loan or the related LM Mortgaged Property is adversely affected; or
- (iv) notice is received from a person that claims to have an Encumbrance ranking in priority to or equal with the Encumbrance held by the Borrower or Custodian and constituted by any LM Mortgage and such notice is not withdrawn within 10 Business Days of receipt (or the Financier is otherwise satisfied that the claim will not succeed).

where (in respect of any matter described in sub-clauses 17.1(ee)(i) to (ee)(iv)) the balance of the Adversely Affected LM Loans exceeds (in aggregate) \$1,000,000 at that time.

17.2 Financier's rights on Event of Default

If any Event of Default occurs then during the period it subsists the Financier may by notice to the Borrower take any one or more of the following actions:

- (a) declare that an Event of Default has occurred;
- (b) declare that all of the Commitment and any other obligations of the Financier to the Borrower will be cancelled immediately and all fees payable in relation to the Commitment will become immediately due and payable, and those amounts will become immediately due and payable;
- (c) declare the Advance, interest on the Advance and all other money the payment or repayment of which forms part of the Obligations immediately due and payable, and those amounts will become immediately due and payable; and
- (d) at the cost of the Borrower, appoint a firm of independent accountants or other experts to review and report to the Financier on the affairs, financial condition and business of the Borrower or Responsible Entity.

18. Indemnities

- (a) The Borrower will on demand by the Financier indemnify the Financier against any Cost, loss, damage or claim which the Financier may sustain or incur as a consequence of:
 - (i) any sum payable by the Borrower under any Finance Document not being paid when due (including as a result of any Advance not being

Legal\301465383.15

prepaid in accordance with a notice of prepayment given by the Borrower);

- (ii) the occurrence of any Event of Default or Potential Event of Default;
- (iii) an Advance requested in a Utilisation Notice not being provided for any reason including failure to fulfil any condition precedent but excluding any default by the Financier;
- (iv) any enquiry, investigation, subpoena (or similar order) or litigation with respect to the Borrower or with respect to any transaction contemplated by any Finance Document;
- (v) the Financier acting on any fax or other notice or on any direct oral or telephone instructions or request reasonably believed by the Financier to originate from the Borrower or an Authorised Officer of the Borrower; or
- (vi) the Financier receiving payments of principal other than on the last day of an Interest Period for any reason including prepayment in accordance with a Finance Document.

These Costs, losses, damages and claims will include the amount determined by the Financier as being any Break Costs suffered or incurred by that Financier by reason of the liquidation or re-employment of deposits or other funds acquired or contracted for by the Financier to fund or maintain any Advance.

(b) The Financier must, as soon as reasonably practicable after demand by the Borrower, provide a certificate to the Borrower verifying the amount of its Break Costs claimed under clause 18(a).

19. Set-off

- (a) The Borrower authorises the Financier, at any time after an Event of Default has occurred and is subsisting, to apply without prior notice any credit balance (whether or not then due) to which the Borrower is at any time entitled on any account and in any currency at any office of the Financier in or towards satisfaction of any sum then due and unpaid from the Borrower to the Financier under any Finance Document.
- (b) The Borrower further authorises the Financier, at any time after an Event of Default has occurred and is subsisting, to set-off without prior notice any amount owing (whether present or future, actual, contingent or prospective) by the Borrower to the Financier under any Finance Document against any liability (whether present, future, actual, contingent or prospective) of the Financier to the Borrower under any Finance Document or on any other account.
- (c) The Financier will not be obliged to exercise any of its rights under this clause 19, which will be without prejudice and in addition to any right of set-off, combination of accounts, lien or other right to which it is at any time otherwise entitled (whether by operation of law, contract or otherwise).

20. Costs, stamp duties and GST

20.1 Costs

- (a) The Borrower must reimburse the Financier on demand for, and indemnifies the Financier against, all Costs including legal fees, costs and disbursements (on a full indemnity basis and determined without taxation, assessment or similar process) incurred in connection with:
 - (i) (Preparation) negotiating, preparing and executing the Finance
 Documents and any subsequent consent, agreement, approval, waiver or
 amendment relating to the Finance Documents; and
 - (ii) (Enforcement) exercising, enforcing or preserving, or attempting to exercise, enforce or preserve, any Power under the Finance Documents, including any Costs incurred in the evaluation of any matter of material concern to the Financier or the investigation of any matter which the Financier determines, acting reasonably, may be an Event of Default or Potential Event of Default.

20.2 Stamp duties

The Borrower:

- (a) must promptly pay all stamp duty, transaction, registration and similar Taxes, including fines and penalties which may be payable to, or required to be paid by, any appropriate Government Authority or determined to be payable in connection with the execution, delivery, performance or enforcement of the Finance Documents or any payment, receipt or other transaction contemplated by them; and
- (b) indemnifies the Financier against any loss or liability incurred or suffered by it as a result of the delay or failure by the Borrower to pay any Taxes as required in accordance with clause 20.2(a).

20.3 GST

- (a) The consideration payable for any supply made by or through the Financier under or in connection with any Finance Document does not include GST.
- (b) If GST is payable in respect of any supply made by or through the Financier under or in connection with any Finance Document (GST Liability) then:
 - (i) where consideration is provided by the Borrower in relation to that supply, the Borrower will pay an additional amount to the Financier equal to the full amount of the GST Liability; and
 - (ii) except where clause 20.3(b)(i) applies, the Borrower will indemnify and keep the Financier indemnified for the full amount of the GST Liability.

If required by law, the Financier will provide to the Borrower a tax invoice complying with the relevant law relating to any payment made to it in accordance with this clause 20.3(b).

(c) Any payment or reimbursement required to be made to the Financier under any
Finance Document that is calculated by reference to a Cost or other amount paid or
incurred will be limited to the total Cost or other amount less the amount of any

input tax credit or other credit to which the Financier is entitled for the acquisition to which the Cost or other amount relates.

21. Assignments and confidentiality

21.1 Successors and assigns

This agreement is binding on and enures to the benefit of each party to it and its respective successors and permitted assigns.

21.2 Assignments by the Borrower

The Borrower cannot assign, novate or otherwise transfer any of its rights or obligations under any Finance Document without the prior written consent of the Financier.

21.3 Assignments by Financier

The Financier may assign all or any of its rights or transfer all or any of its rights or obligations under the Finance Documents to any person it determines at any time provided that with effect from such assignment clause 15 shall be deemed to have been amended so that references to the "Responsible Entity" in its personal capacity are deleted in regard to the representations in clause 15 without in any way limiting any breaches of such representations that may have occurred prior to the assignment

21.4 No deductions and withholdings or increased costs in certain circumstances:

- (a) If:
 - the Financier assigns or transfers any of its rights or obligations under the Finance Documents or changes the office through which it provides the Facility; and
 - (ii) the Borrower would be obliged to make a payment to the New Financier or Financier acting through its new office under clause 11 or clause 13,

then the New Financier or Financier acting through its new office is only entitled to receive payment under those clauses to the same extent as the existing Financier or Financier acting through its previous office would have been if the assignment, transfer or change had not occurred.

- (b) A Financier may not assign or transfer any of its rights or obligations under the Finance Documents or change the office through which it provides the Facility, if the New Financier or the Financier acting through its new office would be entitled to exercise any rights under clause 12 as a result of circumstances existing at the date the assignment, transfer or change is proposed to occur.
- (c) A Financier must bear its own costs and expenses (including legal fees) in connection with any such assignment or transfer.

21.5 Confidentiality

(a) Subject to clause 21.5(b), the Financier will not disclose any confidential or unpublished information or documents supplied by the Borrower in connection with the Finance Documents.

- (b) The Financier will be entitled to disclose any confidential information or documents referred to in clause 21.5(a) or otherwise relating to the Borrower or the Finance Documents:
 - in any proceeding arising out of or in connection with any Finance
 Document to the extent that the disclosure is deemed by the Financier necessary to protect its interests;
 - (ii) where the information is in the public domain other than as a result of a breach of this clause 21.5;
 - (iii) if required to do so under a binding order of any Government Authority or any procedure for discovery in any proceedings;
 - (iv) if required to do so under any Binding Requirement;
 - (v) to ratings agencies, to the extent required by them;
 - (vi) otherwise as required or permitted by any Finance Document;
 - (vii) to its Related Bodies Corporate, its legal advisors and its consultants as long as disclosure is made on a confidential basis;
 - (viii) to a proposed assignee or transferee of any rights or obligations under any Finance Document or to any sub-participant or other person with whom any other transaction may be entered into under which payments may be made by reference to any Finance Document or the Borrower so long as disclosure is made on a confidential basis; or
 - (ix) with the Borrower's prior written consent.
- (c) The Borrower will ensure and procure that it, its directors, employees or any Related Body Corporates (Related Persons) will treat the terms of this agreement and the involvement of the Financier as confidential information and no announcement or communications relating to the negotiations of the parties, to the Financier's role as lender or to the existence, subject matter or terms of the Finance Documents may be made or authorised by the Borrower or Related Persons unless:
 - in any proceeding arising out of or in connection with any Finance
 Document to the extent that the disclosure is deemed by the Borrower necessary to protect its interests;
 - (ii) where the information is in the public domain other than as a result of a breach of this clause 21.5;
 - (iii) if required to do so under a binding order of any Government Authority or any procedure for discovery in any proceedings;
 - (iv) to its Related Bodies Corporate, its legal advisors and its consultants as long as disclosure is made on a confidential basis;
 - (v) if required to do so under any Binding Requirement;
 - (vi) to inform investors in the LM Mortgage Trust, or any feeder fund of the LM Mortgage Trust, in relation to the affairs of the LM Mortgage Trust or the relevant feeder fund;

- (vii) in response to a request from any Government Authority, or to inform any Government Authority, in relation to the affairs of the LM Mortgage Trust, or any feeder fund of the LM Mortgage Trust; or
- (viii) with the Financier's prior written consent.
- (d) This clause 21.5 will survive only until the third anniversary of the date on which all Advance are fully and finally repaid.

22. Governing law

22.1 Governing law

This agreement is governed by the law applying in Queensland.

23. Arbitration

23.1 Notice of Dispute

If any dispute or difference arises between the parties, directly or indirectly based upon, arising out of, relating to or in connection with this agreement (including but not limited to any question relating to the existence, validity or termination of this agreement) ("Dispute"), either party may give a notice in writing to the other party, adequately identifying and providing details of the Dispute.

23.2 Negotiations

The parties agree that they will attempt to resolve the Dispute through good faith negotiations within ten (10) Business Days after the date of receipt of the Notice of Dispute, or such longer period as is agreed in writing between the parties ("Negotiation Period")

23.3 Reference to arbitration

- (a) If the Dispute is not resolved during the Negotiation Period, the Dispute shall be referred to and finally resolved by arbitration in accordance with the arbitration rules of the Australian Centre for International Commercial Arbitration (known as the ACICA Arbitration Rules).
- (b) The seat of the arbitration will be Sydney.
- (c) The number of arbitrators will be 1.
- (d) The language of the arbitration will be English.

23.4 General principles

The parties further agree to the following general principles relating to the procedure of the arbitration:

- (a) that they have chosen arbitration for the purposes of achieving a just, quick and cost-effective resolution of any dispute;
- (b) that any arbitration conducted pursuant to this clause shall not necessarily mimic court proceedings and the practices of those courts will not regulate the conduct of the proceedings before the arbitral tribunal;

- (c) that in conducting the arbitration, the arbitral tribunal must take into account the matters set out above, particularly in deciding issues such as:
 - (i) how many written submissions will be allowed;
 - (ii) where appropriate, the length of written submissions;
 - (iii) the extent of document discovery permitted, if any;
 - (iv) the consolidation of arbitration proceedings, when requested;
 - (v) the joinder of parties or the consolidation of proceedings, when requested;
 - (vi) the length of any hearing; and
 - (vii) the number of experts, if any, each party is allowed to appoint; and
- (d) that the arbitral tribunal has the power to grant all legal, equitable and statutory remedies, except punitive damages.

23.5 Expedited proceedings

- (a) The parties agree that the arbitral tribunal will conduct the arbitration as expeditiously as possible and no party will unnecessarily delay the arbitration proceedings.
- (b) All evidence in chief will be in writing, unless otherwise ordered by the arbitrator.
- (c) Each party may only rely upon one expert witness in respect of any recognised area of specialisation, unless otherwise ordered by the arbitral tribunal.
- (d) After consultation with the parties the arbitral tribunal will determine whether to conduct the proceedings on the basis of documents and other materials only or whether an oral hearing will be held. In doing so the arbitral tribunal shall have particular regard to the parties' request for an expedited procedure and the rules of natural justice.
- (e) If the arbitral tribunal determines that an oral hearing will be conducted, the following principles will apply in respect of the oral hearing:
 - (i) the duration of the oral hearings shall be fixed by the arbitral tribunal;
 - (ii) unless otherwise ordered by the arbitral tribunal, the oral hearing shall be conducted on a stop-clock basis with the effect that the time available to the parties will be split equally between the parties so that each party shall have the same time to conduct its case unless, in the opinion of the arbitral tribunal, such a split would breach the rules of natural justice or is unfair to one of the parties;
 - (iii) oral evidence in chief at the hearing shall be permitted only with the permission of the arbitral tribunal for good cause;
 - (iv) not less than 14 days prior to the date fixed for the oral hearing, or any other period of time specified by the arbitrator, each party shall give written notice of those witnesses (both factual and expert) of the other party that it wishes to attend the hearing for cross-examination;

(v) in exceptional circumstances the arbitrator may extend the time for the oral hearing set pursuant to sub-paragraph (i) above.

23.6 Consolidation

The parties agree that section 24 of the International Arbitration Act 1974 (Cth) will apply.

23.7 Joinder

The arbitral tribunal has the power, on the application of any party to this arbitration agreement, to allow a third party who the arbitrator considers has a sufficient interest in the outcome of the arbitration to be joined in the arbitration as a party. Each party to this agreement hereby consents to such joinder. In the event of such joinder of parties in the arbitration, the arbitrator has the power to make a single final award, or separate awards, in respect of all parties so joined in the arbitration.

23.8 Award final and binding

Any award will be final and binding upon the parties.

23.9 Effect on Powers

The Borrower agrees that notwithstanding this clause 23, the Financier will be entitled to exercise any Power under its Security if the Financier determines in its sole discretion that the exercise of the Power is necessary or desirable to protect the Financier's interests under the Finance Documents, and any exercise of the Power shall be unaffected by the existence of any Dispute, provided that the Financier agrees that its exercise or purported exercise of the Power will not preclude the Borrower seeking injunctive relief and/or damages arising from the Financier's exercise or purported exercise of the Power.

24. Miscellaneous

24.1 Certificate of Financier

A certificate or determination in writing signed by the Financier or an Authorised Officer of the Financier certifying:

- (a) an exchange rate, a rate of interest or an amount payable by the Borrower under any Finance Document is sufficient evidence of the matters to which it relates; and
- (b) any other act, matter or thing relating to any Finance Document is sufficient evidence of those matters unless the contrary is proved.

24.2 Notices

- (a) Except as otherwise specified in any other Finance Document, each communication (including each notice, consent, approval, request or demand) under or in connection with any Finance Document:
 - must be legible and in writing or, subject to clause 24.2(c), dispatched by electronic communication;
 - (ii) must be addressed or dispatched by electronic communication to the following address or facsimile number (or as otherwise notified, in the case of the Borrower, to the Financier and, in the case of the Financier, to the Borrower, from time to time or when it becomes a party to the Finance Documents):

66

Borrower

Address:

Level 4, RSL Centre, 9 Beach Road,

Surfers Paradise, Qld, 4217

Fax:

+61 7 5592 2505

Email:

stickner@lmaustralia.com

For the attention of:

Simon Tickner

Financier

Address:

126 Philip St., 14th Floor

Sydney, NSW

Australia

Fax: Email: + 61 2 8258 3632 lm1@db.com

For the attention of:

George Wong

with a copy to:

Address:

55/F Cheung Kong Center

2 Queen's Road Central

Hong Kong

Fax: Email: +852 2203 7224

For the attention of:

beaux.pontak@db.com
Beaux Pontak, Deutsche Bank AG, Hong

Kong Branch

- (iii) must be signed, or in the case of electronic communication dispatched, by the sender or by an Authorised Officer of the sender;
- (iv) must be delivered by hand or posted by prepaid post to the address, dispatched by electronic communication or sent by fax to the facsimile number, in each case, provided for in accordance with clause 24.2(a)(ii);
- (v) is taken to be received by the addressee:
 - A. (in the case of prepaid post) on the second Business Day after the date of posting;
 - B. (in the case of fax) at the time in the place to which it is sent equivalent to the time shown on the transmission confirmation report produced by the fax machine from which it was sent;
 - C. (in the case of delivery by hand) on delivery; and
 - D. (in the case of electronic communication) on the electronic communication entering a system outside the control of the sender, or the first of those systems where the electronic communication enters successively 2 or more systems.
- (b) Subject to clause 24.2(c), if under any provision of any Finance Document any notice or other communication is required to be in writing that requirement is taken to have been satisfied if the person giving the notice or other communication gives that notice or other communication by electronic communication in accordance with this clause 24.2.

- (c) Despite any other provision of any Finance Document:
 - (i) each Utilisation Notice and Selection Notice; and
 - (ii) any notice or other communication which must be given, served or made under or in connection with clause 17.1(bb) or if not given, served or made under a Finance Document may give rise to an Event of Default,

must be in writing in order to be valid.

- (d) Each communication sent in accordance with this clause 24.2 may be relied on by the recipient if the recipient, acting reasonably, believes the communication to be genuine and if it appears to be executed by or on behalf of the sender in accordance with clause 24.2(a)(iii) (or as otherwise required in accordance with any other Finance Document) or to have been sent from the email address of an Authorised Officer of the sender (without the need for further enquiry or confirmation).
- (e) Any communication sent or received by the Borrower in accordance with this clause 24.2 will be taken to have been sent or received by the Borrower.

24.3 Continuing obligation

Each Finance Document constitutes a continuing obligation regardless of any settlement of account, intervening payment, express or implied revocation or any other matter or thing until a final discharge has been given to the Borrower.

24.4 Settlement conditional

If:

- (a) (i) the Financier has at any time released or discharged:
 - A. the Borrower from its obligations under any Finance Document; or
 - B. any assets of the Borrower from a Security,

in either case in reliance on a payment, receipt or other transaction to or in favour of the Financier; or

- (ii) any payment, receipt or other transaction to or in favour of the Financier has the effect of releasing or discharging:
 - A. the Borrower from its obligations under any Finance Document; or
 - B. any assets of the Borrower from a Security; and
- (b) that payment, receipt or other transaction is subsequently claimed by any person to be void, voidable or capable of being set aside for any reason, including under an Insolvency Provision or under the general law; and

(c) that claim is upheld or is conceded or compromised by the Financier,

then:

- (i) the Financier will immediately become entitled against the Borrower to all rights (including under any Finance Document) as it had immediately before that release or discharge;
- (ii) the Borrower must, to the extent permitted by law:
 - A. immediately do all things and execute all documents as the Financier may, acting reasonably, require to restore to the Financier all those rights; and
 - B. indemnify the Financier against all Costs and losses suffered or incurred by it in or in connection with any negotiations or proceedings relating to the claim or as a result of the upholding, concession or compromise of the claim.

24.5 Further acts and documents

The Borrower must promptly do all further acts and execute and deliver all further documents (in form and content satisfactory to the Financier, acting reasonably, and at the entire Cost of the Borrower) required by law or requested by the Financier, acting reasonably, to give effect to each Finance Document or to perfect or improve the Powers afforded or created, or intended to be afforded or created, by any Finance Document.

24.6 Attorney

The Borrower irrevocably appoints:

- (a) the Financier; and
- (b) each Authorised Officer of the Financier,

jointly and severally the attorney of the Borrower, in the Borrower's name and on the Borrower's behalf, at any time and in any manner as the Financier in its absolute discretion thinks fit to do all acts necessary or proper to:

- (i) further or fully assure any Finance Document to the Financier; and
- (ii) perfect the Powers afforded or created, or intended to be afforded or created, by any Finance Document.

Each attorney may exercise its powers under this clause 24.6 notwithstanding that this may result in a conflict between the duty of that attorney to the Borrower and either the interests of that attorney or any Related Body Corporate or other related person of that attorney or another duty of that attorney.

24.7 Severance

If at any time any provision of any Finance Document is or becomes illegal, invalid, void or unenforceable in any respect under the law of any jurisdiction, that will not affect or impair the legality, validity or enforceability:

(a) in that jurisdiction of any other provision of that Finance Document; or

Legal\301465383.15

 under the law of any other jurisdiction of that or any other provision of that Finance Document.

24.8 Powers cumulative

The Powers conferred by each Finance Document on the Financier are cumulative and in addition to all other Powers available to the Financier by law or under any other Finance Document.

24.9 Waiver

- (a) Failure to exercise or enforce, or a delay in exercising or enforcing, or the partial exercise or enforcement of, any Power provided by law or under any Finance Document by the Financier does not preclude, or operate as a waiver of, the exercise or enforcement, or further exercise or enforcement, of that or any other Power provided by law or under any Finance Document.
- (b) A waiver or consent given by the Financier under any Finance Document is only effective and binding on the Financier if it is given or confirmed in writing by the Financier.
- (c) No waiver of a breach of a term of a Finance Document operates as a waiver of another breach of that term or of a breach of any other term of any Finance Document.

24.10 Consents

A consent required under any Finance Document from the Financier may be given or withheld, or may be given subject to any conditions, as the Financier (in its absolute discretion) thinks fit, unless the Finance Document expressly provides otherwise.

24.11 Indemnities

- (a) Each indemnity in each Finance Document is a continuing obligation, separate and independent from the other obligations of the parties, and survives termination, completion or expiration of the Finance Documents.
- (b) It is not necessary for the Financier to incur any expense or to make any payment before enforcing a Right of Indemnity conferred by any Finance Document.

24.12 Time of the essence

Time is of the essence in respect of the Borrower's obligations under the Finance Documents.

24.13 Moratorium legislation

To the fullest extent permitted by law, all laws which at any time operate directly or indirectly to:

- (a) lessen or affect in favour of the Borrower any obligation under any Finance Document; or
- (b) delay or otherwise prevent or prejudicially affect the exercise by the Financier of any Power under any Finance Document or otherwise,

are expressly waived.

24.14 Binding on each signatory

Each Finance Document binds and is enforceable against each party despite:

- (a) any other person not executing a Finance Document or its execution being defective in any way; or
- (b) any obligation or liability of any other party under a Finance Document not being binding or enforceable against that party for any reason.

24.15 Counterparts

- (a) Any Finance Document may be executed in any number of counterparts and by the parties on separate counterparts. Each counterpart constitutes an original of that Finance Document, and all together constitute one document.
- (b) A party who has executed a counterpart of any Finance Document may exchange that counterpart with another party by faxing it to that other party and, if that other party requests it, promptly delivering that executed counterpart by hand or post to the other party. However, the validity of that Finance Document is not affected if the party who has faxed the counterpart delays in delivering or does not deliver it by hand or by post.

24.16 Entire agreement

To the extent permitted by law, in relation to their subject matter, the Finance Documents:

- (a) embody the entire understanding of the parties, and constitute the entire terms agreed by the parties; and
- (b) supersede any prior written or other agreement of the parties.

25. No representation by or reliance

The Borrower acknowledges that:

- the Financier does not have any duty to supply it with information in relation to or affecting the Financier before the date of this agreement or during the currency of any Finance Document;
- (b) it has relied on its own inquiries as to the nature and extent of its entire relationship with the Financier (whether or not recorded in the Finance Documents) and the nature and effect of the Finance Documents; and
- (c) it does not enter into any Finance Document in reliance on any representation, promise, statement, conduct or inducement by or on behalf of the Financier except for any inducement expressly set out in the Finance Documents.

Schedule 1 - Documentary conditions precedent

A. Conditions precedent to be delivered before the first Advance

- A verification certificate substantially in the form of Schedule 8 from each of the Borrower, the Responsible Entity and the Service Provider.
- 2. A certified extract of the resolutions of the directors of the Responsible Entity:
 - (a) approving the Facilities;
 - (b) confirming that it is in the Borrower's best interests to enter into those Finance Documents;
 - authorising the execution by the Borrower of those Finance Documents and the performance by it of its obligations under those Finance Documents; and
 - (d) appointing an Authorised Officer or Authorised Officers.
- 3. If any Finance Document will be executed by any party under power of attorney, the original or a duly certified copy of that power of attorney or those powers of attorney, duly executed and, in respect of the Custodian, appropriate evidence of the Custodian having been instructed by the Borrower to execute the Transaction Documents to which it is party.
- 4. The Finance Documents duly executed by all parties and, if necessary, stamped and in registrable form (or if not stamped the Financier is satisfied it has received from the Borrower sufficient funds to pay the estimated stamp duty).
- Replies to all requisitions of the Financier and its solicitors relating to the Facilities and the Finance Documents.
- A copy of all Authorisations required for the entry into or performance of the Finance
 Documents by the Borrower or the Responsible Entity including for the avoidance of doubt
 each Material Authorisations.
- 7. An opinion, addressed to the Financier, of the legal advisors to the Financier as to all matters relating to the Borrower, and the Finance Documents as the Financier may require.
- 8. Evidence that any Encumbrances in respect of any of the assets of the Trust Property which are not permitted by the Finance Documents have been released by the person in whose favour they were granted and in particular the Encumbrances granted in favour of the Commonwealth Bank of Australia.
- 9. Without limiting the following obligation, a settlement statement issued by the Commonwealth Bank of Australia setting out details of all amounts that are or will be due to it from the Borrower as at Financial Close and evidence that all Financial Liabilities of the Borrower which are not permitted by the Finance Documents have been fully and finally repaid.
- Due Diligence Report.
- 11. All documentation and other evidence requested by the Financier to comply with its know your client procedures in respect of the Borrower.
- 12. The Financier being satisfied that no instruction can be given to the Custodian without the Financier being a co-signatory to that instruction.

For the purposes of this Schedule 1, "certified" means certified by a director of the Borrower in the form required by the Financier.

Schedule 2 - Form of Utilisation Notice and Selection Notice

A. Form of Utilisation Notice To: Deutsche Bank AG, Sydney Branch A (Financier) From: LM Investment Management Limited ABN 68 077 208 461 in its capacity as Responsible Entity for the LM First Mortgage Income Fund ARSN 089 343 288 (Borrower) Date: [] **UTILISATION NOTICE** Facility agreement dated [] (Facility Agreement) Dear Sirs Unless otherwise defined, expressions used in this notice have the meaning given to them in the Facility Agreement. This is a Utilisation Notice and is irrevocable. We give you notice under clause 4.1 of the Facility Agreement that the Borrower requires an Advance as follows: **Utilisation Date:** date Amount: \$[amount] Purpose: [purpose] Interest Period: [one, month] Payment Instructions: [insert] We confirm that: 1. each Repeating Representation is true and correct on the date of this Utilisation Notice and will be true and correct on the proposed Utilisation Date (both before and after the Advance is made) in each case with reference to the facts and circumstances existing at that time; and no Event of Default or Potential Event of Default is subsisting or will result from the making 2. of the Advance1. Yours faithfully [Insert name of Authorised Officer]] For and on behalf of [1 Delete as appropriate

Schedule 3 - LM Loan Schedule

Schedule 4 - Release Prices

Description of LM Mortgaged Property Release Price

1687,333 1,687,333 2,050,824 2,050,824 2,117,647 2,117,647 2,050,824 2,050,829		20 20 20 20 20 20 20 20 20 20 20 20 20 2
Weight Construction (Fig.) State of the Construction of the Const	MARGE LINE OF THE PROPERTY OF	Automotic Linited Automotic Li
ACOSTACO MATRICA PARTICAL MATRICA PARTICAL MATRICA PARTICAL MATRICA PARTICAL MATRICA PARTICAL MATRICA PARTICAL MATRICAL PARTICAL MATRICAL MATRICAL	Abenment interest (Tuttes of Permanent (Tuttes of P	Formanie (Fortis Periode (Fort
Marita de Marcono, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864, 1864,	1000000 1 1000000 1 1000000 1 1000000 1 1000000	
100% 100% 100% 100% 100% 100% 100% 100%		£\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$
(1) 224 FOLING IN REAL BENEAR (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)		
Provide Bulk Types B The transport of the transport of t	A Alexandra Alex	Translesson Transl
	######################################	2 2 2 5 6 5 4 5 8 6 5 2 2 2 3 8 5 5 5 5 6 5 2 2 8 8 5 5 5
2008. Standing Billing Billing State		######################################
Stewarzen (11) Antonio (12) Ant	United Control	Cheek
MATRICA, MAIL SAN, PETRA LOSS PLESS AND SAN AN	And the control of th	The state of the s
A Northelle A Nort		PROPERTY OF THE PROPERTY OF TH
3 TILE RESPONDE RECEDENTS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHOOLS SCHO	20110000000000000000000000000000000000	10 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17 00 17
THE PROPERTY OF THE REAL PROPERTY OF THE PROPE		57 国外连续扩张的政治
A CHARLES AND A		
RACENIA MINE BESTATIONS OF SECOND MINE BESTATION OF SECOND MINE BESTATION OF SECOND MINE BESTATION OF SECOND MINE BY LESS OF SECOND MINE	Manifestina Springs 20 PV 15 in the control of the	Profession That Control Transcription That Control Transcription The Control Transcription The Control Transcription Transcripti
CONTROL (ASSESSED IN A SECOND	MAN STREET, WEST S	THE STATE OF THE S
Schedule No : See Front Market Color Res Res Front Market Color 2 2 3 4 4 6 6 6 6 6 6 7 7 7 7 7 7 7 7 7 7 7 7	6 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	8 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2

				
	IN PRICE CONTROL OF THE PRICE	19.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9.7731 9	22 22 22 22 22 22 22 22 22 22 22 22 22	25 25 25 25 25 25 25 25 25 25 25 25 25 2
	Secretary Relations in the least of the leas	Annual de la companya del companya de la companya de la companya del companya de la companya de	A CONTROL OF THE PROPERTY OF T	Emiled
	The Australia The Au	the Actual is th	A CONTROLLED ON THE CONTROLLED	Since Australia Since Australia Since Australia Since Australia Since Australia Since Australia Since Australia Since Australia Since Australia Since Australia
	ample of the state	The state of the s	The state of the s	Portagnosti in Pormatshi In Pommatshi In
	TOTAL SERVICE		100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 -	704/10/21/10/21/21/21/21/21/21/21/21/21/21/21/21/21/
	600% A DECEMBER 1000% A	(100%)		100% 100% 1100% 1100% 1100% 1100% 1100% 1100% 1100% 1100% 1100% 1100% 1100% 1100% 1100% 1100%
	Applied to the control of the contro	Puri Puri Puri Puri Puri Puri Puri Puri	Record (1908) Re	
	######################################	20,100 100 100 100 100 100 100 100 100 10	្ត្រី ខេត្តក្តីក្តីក្តីក្នុងខេត្តក្នុងនិក្សាក្នុងខាងនៅអ្វីក្តីក្នុងខុស្តិក្ខុងខាងខាងក្នុងខ្ពស់ និក្សាក្នុងខាងខ ខេត្តក្តីក្តីក្តីក្តីក្តីក្នុងខាងខាងខាងក្តីក្តីក្តីក្នុងខាងខាងខាងខាងខាងខាងខាងខាងខាងខាងខាងខាងខាង	1,286 1,286 1,286 1,286 1,286 1,286 1,128 1,128
		M_ X	· · · · · · · · · · · · · · · · · · ·	1,123 1,123 1,123 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234 1,234
	Unes Unes Unes Unes Unes Unes Unes Unes			# 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5
		MARCHAN STANDARD STAN		in the state of th
	Activities of the control of the con	the state of the s		in the control of the
Control Cont	Annual State	And Price are and the control of the		Tercombal (Y)
Control Cont	Secretary States and Secretary Secre	Interpretation of the control of the	INTERPRETATION OF THE PROPERTY	Tanky Read. Tanky Read. Tanky Read. Tanky Road. Tanky Road. Tanky Road. Tonky Road. Tonky Road. Tonky Road. Tonky Road. Tonky Road. Tanky Road. Tanky Road. Tanky Road. Tanky Road.
Company Comp	100 March 1986 March 1	Control of	######################################	Actions Act
Control Cont	LE RESCUENTE LE RE	BENTRA BE		MEMORIAN CONTROL OF CO
Company Comp	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	11 (10 600) His man with the country of the country		501
Control Cont	A CONTROL OF THE CONT			<u> </u>
Mail of the part Mail of the				100 CO
THE STATE OF THE S		Single-		Para de la companya d
THE STATE OF THE S				Face (1) This is a part of the
March 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971 1971		TEXAL PROPERTY OF THE PROPERTY	######################################	SECOND STATE STREET STATE STREET STATE THOUSEN (Ve)
THE STATE OF THE S	TOTALITY TARKING STATES THE STATES TO THE STATES TO THE STATES THE		Company of the part of the par	TELEGRAPH POR CONTROL OF THE CONTROL
	100 100 100 100 100 100 100 100 100 100			828888888888888888888888888888888888888
	0010694 Deep 1100 1100 1100 1100 1100 1100 1100 1	MATERIAL STATES OF THE STATES	12 12 12 12 12 12 12 12 12 12 12 12 12 1	MARCHEN MARCHE
横立て工工工の内になるないはならなるとなるないというとなっている。 大きな 大きな ちゃん ちゃん おおり はんしょう カー・ファイト・ファイト・ファイト・ファイト・ファイト・ファイト・ファイト・ファイト	44			200
	50 50 50 50 50 50 50 50 50 50 50 50 50 5	18 6 18 18 18 18 18 18 18 18 18 18 18 18 18	######################################	200

4 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		765,889 865,090 872,272 1,319,644 885,800
month filtrack Acades In Info mental filtrack Acades Linked mental filtrack Acades Linked	mental Transport United Adults Linguistics Control Adults Control	essanani Trustee Australia Limber ensanani Trustee Australia Limbed ensanani Trustee Australia Limbed ensanani Trustee Australia Limbed ensanani Trustee Australia Limbed
PA 201505.	### 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 1995-1997 19	1309289 6 4309290 PP 1309289 PP 1
	1000 1000 1000 1000 1000 1000 1000 100	* 200 * 200
Land Office of the Control of the Co	March 1997 Mar	Lizury Aparlments Lizury Aparlments Lizury Aparlments Lizury Aparlments
1948 1,484 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1		150 150 150 150 150 150 150 150 150 150
Present to the presen	Character Charac	Uncold Uncold Uncold Uncold
Escotisted Secretarion Secreta		
peter Tingk (seaming) (repeat)	The state of the s	J Mill Form recent, cerein, r cerein, sec. 3 Mil Poler Road, South Perfe, WA. 3 Mill Poler Road, South Perfe, WA.
852 8871100 871100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 8711100 871	19 19 19 19 19 19 19 19	G UNK NO AVESSA FALLS G UNE NO SV2884 F421 G UNE NO SV2884 F425 C UNE NO SV2884 F428 C UNE NO SV2884 F428 T T T T T T T T T T T T T T T T T T T
		Toward To
		-
Levindi muq çabançı üçükböna) Levindi muq çabançı üçükböna) Levindi müq çabançı üçükböna) Levindi müq çabançı üçükböna Lev	Bennichen 79 (19 cm)	Eden Agustments Ply Ltd Eden Apartments Ply Ltd Eden Agartments Ply Ltd Eden Apartments Ply Ltd
110,000 TOTAL ALONG	THE ORDER AND TH	LM-01-00T-AL-1008 LM-01-00T-AL-1008 LM-01-00T-AL-1008 LM-01-00T-AL-1008
2		55 55 55
ត្តក្រុង		

	het het	
184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184 184		446,947 278,478 278,478 542,059 447,928 230,705
Permanent Interpretation of the Management of th	Formurant Theres Asking Links	Permanent Trotte Australia Lumina Permanent Trotte Australia Lumina
2017/01/16/16/16/16/16/16/16/16/16/16/16/16/16		AC983407 AC993407 AC993407 AC993407 AC993407 AC993407
2.00 (100 kg) (100 kg		2003 2003 2004 2001 2001 2001 2001
Table Tabl		
Library Agatom Librar	Office O	Warehouse Warehouse Warehouse Warehouse Warehouse
		22224
100 100 100 100 100 100 100 100 100 100	1982 1982 1982 1982 1982 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983 1983	
United Control of Cont	Chrose to throse	Unsold Unsold Unsold Unsold Unsold
TO MAN TO STATE THE THE THE THE THE THE THE THE THE T	20 Nicholani (My. Bioteca, MSN. 20 Nicholani (My. Bioteca, MSN. 20 Nicholani (My. Bioteca, MSN. 21 Nicholani (My. Bioteca, MSN. 22 Nicholani (My. Bioteca, MSN. 23 Nicholani (My. Bioteca, MSN. 23 Nicholani (My. Bioteca, MSN. 24 Nicholani (My. Bioteca, MSN. 25 Nicholani (My. Bioteca, MSN	CIT Chem and Protein City City Chem and Protein City Chem and Chem and Protein Chem and Ch
TOTAL STATE OF THE PROPERTY OF		SECRECIA (20) SERIORATOR (20)
Appl Lamb Broad and Appl L		Condension Development Pry 110
MAINTENNESS (MAINTENNESS (MAINT	Mail of the color of the colo	

NI E				-		
200,706 200,706 200,706 200,706 201,671 221,671 221,671 221,571 221,571 221,571 221,771 221,771 221,771	22,088 22,088 22,086 22,088 24,088 24,288 28,288 28,288 28,288 28,288 28,288 28,288 28,288 28,288	### 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999 1999	164,810 165,800 165,800 165,800 165,800 164,311 164,311 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,800 164,80	4 5 55 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	24 217 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	155,554 17,250 17,150 15,170 15,170 15,590 15,500 15,500 15,500 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 15,700 1
TOTAL THE PROPERTY OF THE PROP	Atha is Umbed Adhair Chilad Adhair Chilad	and the property of the proper	intonia Luntonia Intonia Luntonia Luntonia Luntonia Luntonia Intonia In	and a limbod and limbo	state in Liceles General Limited Gener	THE SECRET OF TH
CONTRACTOR IN SHARE IN THE SHARE IN THE SHARE IN THE SHARE SHARE IN THE SHARE SHARE IN THE SHARE	about Tustee At about Tustee At annu Tustee At annu Tustee At annu Tustee At wasen Tustee At annum Tustee At annum Tustee At annum Tustee At	annon I Lusboo As annon I Lusboo As	ampari Trustes Araman Intologo Araman Trustes Araman	want if the factors of the factors o	traction of the control of the contr	There is the control of the control
March 2025 Polymer 2 Permanent					Parent Date	1320009 Puri
AC9346 AC9346 AC9340 AC9340 AC9340 AC9340 AC9340 AC9340 AC9340 AC9340 AC9340 AC9340	AVBIEGO AV39166	AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE AASIBE	AA2998 AA2998 AA2998 AA2998 AA2998 AA2998 AA2998 AA2998 AA2998 AA2998	AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS AGDIS	AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818 AA2818	10593741 6.7 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.4 (1993) 1.
#001 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 10	100% 100% 100% 100% 100% 100% 100% 100%	200 200 200 200 200 200 200 200 200 200	1005 1005 1005 1005 1005 1005 1005 1005	\$ 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	100% 100% 100% 100% 100% 100% 100% 100%	200 200 200 200 200 200 200 200 200 200
AND ALL MACHES	****	******				
by Salat Tybe Fit along the state of the sta	aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation aformation	attential attent	and friends attronic att	advincing advinc	aptiments suppressible substances	Semilarias semilarias semilarias semilarias semilarias semilarias semilarias semilarias semilarias semilarias semilarias semilarias semilarias
158 W. W. 159 W. W. W. 159 W.	**************************************	**************************************	**************************************	1 <u>28</u> 45845845888888888888		4442882222222
134 Harris Harris 134	表 你 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	88578987877895557789	22222222222222222222222222222222222222	1 26 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	***************************************	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
COSTO CHAIR A					- EB LU L L	
ERMENTO	Lineald Chroad Chroad Uncode Uncode Uncode Uncode Uncode Uncode	Unradio Unradi	Unseld Un	2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 2004 - 20	SOG: 1995 Varies 664. DAIR Vales 665. DAIR VAL	Sold - DNF Va Sold - DNF Va Sold - DNF Va Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Unsold Un
				200 200 200 200 200 200 200 200 200 200	Paint South State South State South State South State South State South	
FEGURAL STATES OF THE ASSET OF	RROOK, Banois F Rook, Banois F Rook, Banois F Rook, Banois B Rook, Banois B	ROOK, BINDS BY CASA, BY	a Read, Sartonic Landon and Read and Sartonic Landon and Read and Read and Read and Read and Read Bonors at Rea	* Hatel, Benton * Hatel, Benton * * House, Benton * * The Benton *	nn Reder, British an Reder, British an Reder, British an Reder, British and Reder, Britis	
MARKETANTELL SIRINII, SANDAN SIRINII, SANDAN SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII SIRINII S	ith, 2-4 Terranor old, 2-4 Terranor Mai, 2-4 Terranor	ilit, 2.4. Torrano ilit, 3.4. Torrano ilit,	old, 24 Terrano 1016, 24 Terrano	ishin, 2-4 Tomano deli,	ichm, Z.4. Torrano ichm,	CONTROLLER OF THE PROPERTY OF
March Stricts of Warner of Warner of Warner of Power of Power of Power of Power of Warner of Warn	N. View Barbin P. N. View Barbi	In View Barrier Pro- In View B	n View Basters Po 10 View	A WANT SERVICE A WAY WAY SERVICE A WAY SERVI	In View Benorin House Ho	REGISTAL SERVICES GENERAL DIVENTION GENERAL DIVEN
2000 S S S S S S S S S S S S S S S S S S	997 Ocean 997 Ocean 907 Ocean 907 Ocean 907 Ocean 907 Ocean 907 Ocean 907 Ocean 907 Ocean 907 Ocean 907 Oc	100 00000 00000 00000 00000 00000 00000 0000	CO C	22 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	417 000 000 000 000 000 000 000 000 000 0	100 100 100 100 100 100 100 100 100 100
Eleca 1173/se back 105/pc 25 055/pc 25	39 39/5P/77 39 39/5P/77 39 39/5P/77 39 39/5P/77 39 39/5P/77 40 40/5P/77 40 40/	22952600 22972600 259 22972600 259 22972600 259 22972600 259 25972600 25972600 25972600 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 2597260 259726	10-10-10-10-10-10-10-10-10-10-10-10-10-1	11 305974 12 205974 13 205974 14 205974 27 2116776 27 2116776 28 505974 28 505974 28 505974 29 605974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974 405974	5.5 5.5 5.5 5.5 5.5 5.5 5.5 5.5 5.5 5.5	<u> </u>
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Communication	- 40		
E STATE CONTINUE CONT	and the control of th					
प्राप्तका व्यक्त			<u></u>	· 		
######################################						
specials of the control of the contr		- 	FEEEEEEEEEE	<u> </u>		Manual Ma
Action of the control	OVST PY CONST PY CONS		0.041 PM 0.0451 PM	00517 PC 00517 PC		Section 1999 (Section 1999) Section 1999 Section 1999 (Section 1999) Section 1999 (Sec
COM COMMENT INC.	400 FALL 1013 600 FALL 1013	1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-007-44-103 1-	100 140 100 100 100 100 100 100 100 100	4 BODT 440-013 14 BOT 440-013 14 BOT 440-013 14 BOT 440-013 14 BOT 440-013 14 BOT 440-013 14 BOT 440-013 15 BOT 440-013	### OT *** I TO *** I	MACHINE STATE OF THE STATE OF T
199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 199-00 19	LINES		20000000000000000000000000000000000000		NAME OF THE PROPERTY OF THE PR	WILLIAM
Burness (STRE)	SEATON BROWN					in the second se
200 Property C	475 475 476 477 477 478 479 480 481 481 482 483	**************************************	3 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		2%	

	191	
	1010 1010 1010 1010 1010 1010 1010 101	0,557 6,258 6,258 6,258 10,244 10,244 10,244 10,245 12,535 12,536 12,536 12,536 12,536 12,536 12,536 12,536 12,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536 13,536
		Links Christian
AND	100 May 1,	ide Lake Estel de Lake Estel d
Permanent in Perma	Permanent Perman	Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe Bridgewe
17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.120.00 17.12	12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (1992) 12 (19	wood wood wood wood wood wood wood wood
100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990) 100 (1990)	7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	
100 10 10 10 10 10 10 10 10 10 10 10 10		100% 100% 100% 100% 100% 100% 100% 100%
4 de		
500 1991 1991 1991 1991 1991 1991 1991 1		
Management of the property of		
		28
65 5 6 7 7 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	9000000000000000000000000000000000000	######################################
Di	Control of the contro	Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Value Va Value Va Value Va Value Va Va Va Va Va Va Va Va Va Va Va Va Va
Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managarah Managa	NOT SEED TO SEE SEE SEE SEE SEE SEE SEE SEE SEE SE	Sade - Designation of the control of
NO THE REPORT OF THE PROPERTY		
		2008 2008 2008 2008 2008 2008 2008 2008
	A STATE OF THE STA	wigh Park, VI wigh P
Control of the contro	D D D D D D D D D D D D D D D D D D D	And Read, Res See Seed from See Seed from See Seed from See Seed from Seed from Se
10 Addition	1 15 Accesses 1 1 15 Accesses 1 1 15 Accesses 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	9 Wedgaws 9 Wedg
10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		IDPACE F571 IDPACE F572 IDPACE F573 IDPACE F573 IDPACE F574 IDPACE F574 IDPACE F574 IDPACE F576 IDPACE
#1=	2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-	54545454545454545454545454545454545454
STATE OF THE PROPERTY OF THE P		
		
The state of the s		Francisco de la composito della composito de la composito de la composito de l
AND	Calmington Names (1912) Calmington Names (191	Agenetic Life Separation Legisland Separation Legisland Separation Legisland Separation Legisland Separation Legisland Separation Legisland Separation Legisland Separation Legisland Separation Legisland Separation Legisland
<u> </u>	22	2
100 100 100 100 100 100 100 100 100 100	MARKET CONTROLLED AND AND AND AND AND AND AND AND AND AN	MICHOLINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA MICHOCINA
當	.000000	
	33333888888888888888888888888888888888	2
CL.	<u></u>	

10.00000000000000000000000000000000000	
	10 20 20 20 20 20 20 20 20 20 20 20 20 20
AND	AND
CHAINER AND	THE PARTY OF THE P
With the second	AND CONTROL OF THE CO
100 (100 (100 (100 (100 (100 (100 (100	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
#H	
100 100 100 100 100 100 100 100 100 100	
Heiner He	
(a) (a) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c)	
를 다 있다. 한 한 한 분석	**************************************
Notes that the second s	Change of the control
19 19 19 19 19 19 19 19 19 19 19 19 19 1	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
PRINTER STATE OF THE PRINTER STATE OF THE PRINTER STATE STAT	PO 10 10 10 10 10 10 10 10 10 10 10 10 10
Reforming the benging the beng	And the second s
Section 1997 Sectio	troop to some control of the control
MARCON CONTROL OF THE PROPERTY	
1755 1755 1755 1755 1755 1755 1755 1755	
10001010 100010101010101010101010101010	
2 22 22 22 22 22 22 22 22 22 22 22 22 2	
्याको विकास सम्बद्धाः	
THE CONTRACT OF THE CONTRACT O	
A THE SECOND STATE OF THE	
A Company of the Comp	Management Laboratory and an application of the property of th
Management of the control of the con	MAINTENENCE OF THE STATE OF THE
18 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	<u> </u>
100	aaba
N ISSUERIE	
	程 在程度已成在 更更受受 \$ 2 更是 更更 的 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$
<u> </u>	

81 a l	· · · · · · · · · · · · · · · · · · ·	
74.545 74.545 74.545 74.545 74.545 74.545 74.645 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87.745 87	200 (191) 200 (191)	
TANANCE BELIAMENTO BELLANDE BE		The property of the property o
Table Automated	THE OF ALL PROPERTY OF THE OFFICE OFT THE OFT THE OFFICE OFT THE OFFICE OFT THE OFT THE OFT THE OFT THE OF	The second secon
CONSTRUCTOR Pointment II Pointm	Portrainant III Portrainant II	Personal District of the Control of
AD HIGH AD HIG	100 100 100 100 100 100 100 100 100 100	
10 MARTIN	20	
Men walke		
a February III		
IN SUPPLY STAND		Land Land Land Land Land Land Land Land
100 00 00 00 00 00 00 00 00 00 00 00 00	ELSSELESSEEES	
1	PC580004860003333330000044618888	
100 (100 (100 (100 (100 (100 (100 (100	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	过程的 化过程 化过程 化对射性 经工程 化聚物 化二甲基乙二甲基甲基 化二甲基甲基 化二甲基甲基甲基 化二甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基甲基
Hand Section of the s	University of the control of the con	Property of the property of th
100 Per 100 Pe	The control of the co	Change of the ch
SEGMENT OF THE SEGMEN		Part of the control o
Seculation of American Company of American Com	And the second of the second o	mental significant control of the co
interpretation of the property	Miller March 1997 (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994	
9 Web	Maria	
TILIGA MACASA VIOSAGE FEB VIOSAGE FEB VIOSAGE VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE FEB VIOSAGE VIOSAGE FEB VIOSAGE		30 10 10 10 10 10 10 10 10 10 10 10 10 10
2.5 2.5 2.5 2.5 2.5 2.5 2.5 2.5 2.5 2.5	1	
THE THE PERSON OF THE PERSON O		
ह्यासम्बद्धाः स्थापनाम्		
1971 (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (197	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	######################################
AMERICAN CONTROL OF THE PROPERTY OF THE PROPER	Marking promised by the program of the promised by the promise	Problems of the problems of th
Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridgen Bridge	Secured States of States o	Source Development No. 1 Source Source Source Conference Conference No. 1 Source Source Source Conference Conference No. 1 Source Source Source Conference
Tradition International Intern	######################################	8.01 00 00 00 00 00 00 00 00 00 00 00 00 0
MACHON MA	2000 000 000 000 000 000 000 000 000 00	10 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -
ETTER COUNTY		
		2
200 Production 100 Pr	2 2 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	

100	200 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 -	657,44 657,44 657,44 657,44 657,44 657,44 657,44 657,44 657,44	11 11 11 11 11 11 11 11 11 11 11 11 11
The Manual Totals Author Lands Personal Total	Formment of varies believed by the control of the c	The PARTY AND THE CONTROL AND THE COMMENTARY OF	Commonent Dis Batte of Admands
THOUSE TO THE TOTAL TO THE TOTA	On Hopey On Hop	201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 201520 20	100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,000 100,
100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 100% 10	4601 4601 4601 4601 4601 4601 4601 4601	2007 2007 2007 2007 2007 2007 2007 2007	
NEW TOTAL PROPERTY OF THE PROP			0 K M N N N N N N N N N N N N N N N N N N
ILLES AND REGISTED POSITION PO		CONTRACTOR	PROPERTY OF THE PROPERTY OF TH
Appared Appare	Worker Wo	Apath	
単 2 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	200 200 200 200 200 200 200 200 200 200	2 2 2 2 2 3 3 4 4 5 5 5 5 5 5 6 7 7 7 7 8 7 8 7 8 7 8 7 8 7 8 7 8 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 7 8 7 7 8 7 8 7 8 7 8 7 7 8 7 7 8 7 7 8 7 7 7 7 7 8 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	
Toes Alei (1)			######################################
			CARDO
The state of the s	Many Many Many Many Many Many Many Many		22 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
SE SOMEON WAS	通	E CONTRACTOR AND A CONT	
**************************************	就我们是一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个	हा प्रस्ति है। इ.स. १६ वर्षा	
Managements of the common of t	THE STATE OF THE S		Section 19 19 19 19 19 19 19 19
The properties of the properti	1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (1990 (Mary Mary Mary Mary Mary Mary Mary Mary	Fedibi i diputation del production d
20	18 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Travisor Harrison Har	
10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000 (10.000	100 000 000 000 000 000 000 000 000 000	100 M	1
CLE A CLE	- 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	Link No. 2 vi. 1 v	
Control of the contro		Pilot Hattaned	
NI SYSTAM COL		110 A A A A A A A A A A A A A A A A A A	
Account of the control of the contro		7. de la	
n-communication of the communication of the communi			
		A Record Place	The Transit of March 1995 of M
The Student I Marker 8 (Student I Language 1	100 100 100 100 100 100 100 100 100 100	Marie Dev. Ph. Julio Dev. Ph.	The state of the s
100 C C C C C C C C C C C C C C C C C C	MINICIPATION OF THE PROPERTY O	######################################	100 100 100 100 100 100 100 100 100 100
0-7			
400 Per 1	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	29 29 29 29 29 29 29 29 29 29 29 29 29 2	
			•

	The state of the s
	Bening of the property of the
when the property of the prope	The control of the co
Common Co	Permanent Common
	10 10 10 10 10 10 10 10
######################################	
	,
The state of the s	THE STATE OF THE S
\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$	日本の大学を表示を表示を示するとのできるとのできます。 日本の大学を表示を表示を示するとのできます。 日本の大学を表示を表示を示するとのできます。 日本の大学を表示を表示を表示を示する。
28 27 5 5 5 5 5 5 5 5 5 5 5 5 5 5 6 5 6 5 6	### ##################################
ないなのととというないないのはいないのののはないないないないないないでは、これには、これにあるとは、これには、	3 8 8 E E E 8 C C 3 5 8 E E E B 8 B C E E 2 D E E B B B B E E E E E E E E E E E E E
THE STATE OF THE S	A
18	Frequity (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997)
99999999999999999999999999999999999999	
A STATE OF THE STA	STATE OF STA
Rockets Rocket	Andrews and the second
in the second se	be Avonsis of the Avo
The state of the s	Seminary of the control of the contr
9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	(RESS 8 6 6 6 6 6 6 6 6 6 7 6 7 6 7 6 7 7 7 7
60 Section 9. Section	A STATE OF THE STA
<u> </u>	
网络林沙拉尔 化二甲基甲基甲基甲基甲基甲基甲基甲基甲基甲甲基甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲	
	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
	联和场
15 15 15 15 15 15 15 15	
	Turns (Nation) Turns
	SECTION OF THE PROPERTY OF THE
Transh Marsi Tr	An Albert March Ma
	15 The Times With Means 16 The Times With Mea
MARCH DOT ALLORS MARCH SOTT ALL	
	###
000000000000000000000000000000000000000	

			(40 00000000000000000000000000000000000		2
8 5 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20,000 20	22.22.22.22.22.22.22.22.22.22.22.22.22.	######################################	### ### ##############################
Notes Audioid blidge and the state of the st	The state of the s	A CONTROL CONT	There Accessed Links Three Accessed Links	The Act of	and Troste Auroria Linkola and Troste Auroria Linkola
Porter	Permanent Permanent Permanent Permanen	Potes Pores Pores Pores Pores Pores Pores Pores	Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma Perma	Portra	Perm Perm Perm Point Perm Perm Perm Perm Perm Perm Perm
ADVISERED AND SERVISERED AND SERVISE	12 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	AESTROOM	AESTRESOU RESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRESOU AESTRE	AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU AEGYBEOOU	AESTREAM LA PERTREAM LA PERTRE
\$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000 \$5000		%00 %00 %00 %00 %00 %00 %00 %00 %00 %00	100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 % 100 %	00% 00% 00% 00% 00% 00% 00% 00% 00% 00%	######################################
Constitution of the property o	Townships the Townships the Townships to Tow	Wheelease	What a house Water house house Water house	Wherehouse	Whiteholds Whenholds Whenholds Whenholds Whenholds Whenholds Whenholds Whenholds Whiteholds Whiteholds Whiteholds Whiteholds Whiteholds Whiteholds Whiteholds
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		8222222222	**************************************	1月2日の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本	2024
2,55		8222222222	******************	·	***************************************
	The second secon				
United the state of the state o	Sud-Core Value Sud-Core Value Sud-Co	99999999999999999999999999999999999999	Umand	The state of the s	Unsale Unsale Unsale Unsale Unsale Unsale Unsale Unsale Unsale Unsale
			<u></u>		
Woode Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victoria Victor	SERVICE AND TO THE SERVICE AND	Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne Molbourne	Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbourns Melbou	Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia Melboussia	Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn Melbourn
Read Single, Head	15. Montany 15. Montany 16. Montany 16. Montany 17. Montany 18. M	tion 9. South 1 from 1 from 1 south 1 from 1 from 1 south 1 from 1 from 1 south 1 from	filmer St. South filmer filmer filmer st. South filmer filmer st. South filmer filmer filmer St. South filmer filmer filmer st. South filmer s	dianer 18, South dianer 19, South dianer 29, South dianer 29, South dianer 29, South dianer 29, South dianer 21, South dianer 21, South dianer 29, South dianer 20, South dianer	where is south where is south where is south interest in south interest int
16 11 Hogens (16	be a particular of the particu	310-214 Lea 310-214 Lea 310-214 Lea 310-214 Lea 310-314 Lea 310-314 Lea 310-314 Lea 310-314 Lea 310-314 Lea	310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16; 310-34-16;	3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 3003446 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 30034 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300346 300	20034 to 200
157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/261 157/26	152649 FM 152649	10677702 10677704 10677704 10677705 10677705 1067707 1067707 1067707 1067707 1067707	1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/182 1607/1	1190271 1190271 1190272 1190274 1190227 1190227 1190227 1190227 1190227 1190227 1190227 1190227 1190227 1190227 1190227 1190227	(199024) (199024) (199024) (199024) (199024) (199025) (199025)
		588828286	34444444444444444444444444444444444444	######################################	222222222222222222222222222222222222222
					Ş
		1			
		######################################		**************************************	3444444 344444444
be beginned in the properties of the properties	Management of the control of the con	o (Southbank)	(Southeam)	(Southbank)	igo (Sauthbank)
This is the continued of the continued o	Common State Commo	Bungs Won lease won	Constitution of Statistical Part of Local Statistics of Local Stat	MINIS WOOTH MINIS	TOWN SENT THE COMMENT OF THE COMMENT
1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-14-168 1907-1	191	251)		
MARKET DE LA CONTROLLA DEL C					
119 119 119 119 119 119 119 119 119 119		1208 1208 1210 1211 1212 1214 1216 1216 1216 1216		200	255 255 275 275 275 275 275 275 275 275
10143					

ACHES ACCULANTING AND IN	340 347	\$ 10	264,516	420,000	347.858	340,000	32,685	338 344	ENGLISH STREET	Apr Par	301.215	306,073	512,551	140,491	я.	465.321		229,087	172 PM	300 000			230,202,562																													
CHEST COLOR CONTROL CO	Pormonere Treatme Andread Limited	Parmenan Trucioo Australia Limbed	Pomencal Trustee Australia United Pomencal Trustee Australia Lighted	Permanani Trumbe Australia Umited	Peded of minutes and the personal	Powerish Trysles Audralia Limited	Permanent Anales Australia Linded	Parmonnel Toucies Australia Limited	THE RESERVE OF THE PARTY OF THE	Permanent Tondres Andreits I (miles)	Permanent Trustice Auditalia Umaad	Permanent Tursines Australia Limited	Pormanoni Trustree Australia Elmited	Permanent Transfees Audmille Limited		Permanent Trustee Australia Limbed	RECORDER TO THE STORY OF	Pennation Trusted Australia Limited	Vernansen Impres Audralia Imped	Permangal Trustae Authorite Limited	Company of the Party of the Par	Permanent Trucheo Audralia Limited			=					_			-							-				_		_						_
HEAVEL THE WASHINGTON	AC343515	ACTIGNOSS	AC104065		AASO4549			AA904549				_	_	710227227	244220000	711770393		CE62104	AUTOMORPHICA NAMED IN THE	16508597	The second second second second	AB22/1568								Ī																						
20 mm 10 mm	%D03	100%	1985	1 100%	100%	100%	198	100%			-	100%	100%	100%	1	100%		1001	THE RESERVE OF THE PARTY OF THE	100%	和 医解析 的复数经验的	1009																							2		months			-	· ·	
Probatical States Foliate Balancia	Anariments	Shaps	Shaps	Shaps	Apericalism	Apartments	Apariments	Appelments	经发展,是是一种企业的	3 6		200	Diffee	Annual Contract Contr	Street, Square, Square	3		Appetimentle			THE RESERVE OF THE PARTY OF THE	Townstaugn																	-								Marina Parado, Kingschif Pa					**************************************
Hart & Hall Bland	(37	_	322	Contract Annual	The state of the s	_	8 5		ANGEL HEREKUZ	_	122	_	251 251	58 58	0.004	1,012		P	30	4.851	STATE OF THE PARTY	437 437																									If Truck Insking on the 32-34					
hin and the Charles of the State of the Charles	L	_	Chron	SOMEON AND ADDRESS OF THE PERSON ADDRESS OF THE PERSON ADDRESS OF THE PERSON ADDRESS OF THE PERS	The second little and	_	Unick			_	Plosun	_	_	United	WHICH PRODUCED	Uncode	金属 医动物		Supply and object where the supply and the supply a	,	400																										nest and the state of the ATE than 201 Those Expendion ATE the Reterrord 20-22 Months Percent Press Inch and the 32-34 Marine Percent Press (Annual Percent Pe					
STREET, STREET				A STATE OF THE PERSON NAMED IN COLUMN NAMED IN	Walter of the factor of the factor				ENGLISH BATTER THE SECTION AS					s Peradica	Street being the street of the street		HEREN SERVICES AND SERVICES				建设在建筑的建筑的								ant Partnership (Loun 2)																		Omounties ATE the Princes					
Carticles (Constitution of Afficiants	Marine Pd. Managall, NSW	ring Pd, Kingsdill NSW	Shap 2/32-34 Marine Pd, Kingscoff NSW Shap 3/32-34 Marine Pd, Kingscoff, NSW	Sarina Pid, Kingscoff, NSW	NOT ROOK Walnering NSW	and Road, Whitteenga NBW	Unit 14, 32 Cisezia Rosa, Warthoriga Navy Unit 15, 32 Citerata Rest. Walthornes MSW	and Road, Wehrsongs HSW	NAME OF TAXABLE PARTY.	Level 5, 48 Careff Avenue, Surface Paradisa	II AVERTUR, SURFEIS PRINCIPO III Averto Surfeir Demolica	Lavel S. 46 Cavill Avenue, Surfers Paredia	O Avenue, Surfers Paradise	I Avenue, Surfers Persolice		Lot 20 Livision a Close, Bushing Beach, Gir	CONTRACTOR OF THE PROPERTY OF	4 St Springy	Cat 100, 433 Keri Si Sydney	Jewella NSW	THE RESERVE OF THE PARTY OF THE	25 Palemen Read, Bohyana NSW					, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		283 IOR & ATF the Redund Bay Labouta Life Development Partnership (Lesse 2)								Total Total						rinentip				ALCOHOLD OF THE SAME	Lini Trust		ring louisiano poscoj		
A STATE OF THE PROPERTY OF THE PARTY OF THE	Ltn# 10202-34			1				2 22							2				S SEASON SECTION		59 SE UKO						Tried confeed No. 1 Intl' Tried	1) and	U2 383 IOR & ATF the Red		EN.						aw South Tweed Printering	TION OF THE PRINCIPLE O	Holel Trust		MI TRUE	- TO THE SAN DOWN	r fine Tianah Mareh Village Portnership	102 201 658		N. 058 1655 208	Tour Con Developments			it Lie Meneg bd Menormanas ir und Journana Boeso		
A PARTY CONTRACTOR OF THE PARTY	154 day 501		Shop 2 2/8P83332 Shop 3 2/8P83332		ALM 14 SD 75 CHEST	Unii 13 5P 78 12/5P78	Unit 14 14/5/P/1	BE-02-20	计三角形式和图像型数	Lot 43 BUP 1055011947	Lai 45 8110 41150110479		Lot 47 BUP 10/5071046	Loi 48 BUP 10(5011048	룏	Let 20 SP132650350202		551 20 41 376/5137	Let 100 SP 81,100/813	2 200520	AREA BEST MANAGEMENTS	Lot 7 nn Neighbourtood F/7/286108		of octoor releases			R. A. S. F. The Materials of the Control of the Con	24 6 C N 109 477 916 0 con	opmont Pty 154 A.C.N. 112 C		R ATF the St. Chipping Prop	G. C.	L 113 612 41B	Carigal Trast	GN. 119 783 470	/ Limited A.C.N. (19218 174	71 IOR & ATF ING Ocean W	DALIN, GOS BAS DAG DAG DAK S	IOR and ATF The Townsulb	UdA.CM. 129 099 871	OR & ATP Source States I	S Storole Piv Lid	. 119 755 672 as nominea for	well Services Ply Lid A.C.N.	P 114 A C N. 074 173 443	re (Launceston) Phy Ltd A.C.	V Ltd A.C.N. 005 753 391	V LA A.C.N. 108 145 624 IC	dA.C.N. to? 989 686 to R. &	miled A.C.N. 077 206 451 a	LIM A C.N. 104 485 074	558
Service of the Servic	COMPRESSOR STREET				WASSELLING TO SEE SERVING TO				的数据的 (1985年) [1985年] 1985年 [1985] [1985] [1985] [1985] [1985] [1985] [1985] [1985]				_		NEWS CONTROL OF THE BUILDING SHIP		是是一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个一个一			STATE OF THE PARTY	SAN GENERAL MICH. PRINCE OF STREET	7 [9]		will be afternooms of the		STOWER Name	Bezzing Developers Pty Lid IOR & ATF fire Jipósopyne Lhit Triss Stanfor	CHIMOTO DISTORATION OF PART LINE	Canada Day Lebums Life Days	Beliped Ply Lid	Si Criscia's Property Pty Lie ICR ATP the St. Crisch's Proper	Mag Load Corporation Fry La	Eden Appriserie Pty Us A.C.N. 113 612 418	Lot 111 Phy Lid MDR & ATF the Campat Train	eystenes Projects Ply 1.1d A.	andenning Developments Ph	4ST Ph LM A.C.N. 103 210	magnot wastiganess by Li	winsylle Commonthi Ply Lid	states Developments No 1 P.N.	urca Student Lodge Pry Lid	Mar Development Day 147	II Trops Topsh Mersh A.C.N.	shallon international trumster	infly Broken Developments P	Comes Edules Lifethyla Villages (Laumondon) Pby Lbd A.C.N. 058 1655 286	Own Slongge (Southbank) Pr	Magnetic Properties by Lin A.P. Und A.C.N. 108 145 524 ION	Young Lead Corporation Pty Ltd A.C.N., 102 959 696 10 R & A	A broadment Management L.	month of Mestale Respite Physics	HIII Pry LIGACA, 104 B38 5
	Section Colors		1		National Parkets				F	_	avdi Park)		And Park		THE PERSON NAMED IN	d Greech)	ă		Apper Hoddings					and an adversarial and an in-	a, mocalion in noisean parent	Bo	Be	1.00 1.00					100									5		2			ĺ			f Bergith		
	Appropriate the second	Mademy Properties (Resort Corp.)	Madmrs Proposition (Resort Corp.)	Madrors Proparises (Resort	A PARTICIPATION OF THE STREET OF THE STREET OF THE STREET, THE STREET OF THE STREET, THE STREET OF THE STREET,	Magnatile Grove Umestrate	Megnada Grave Investments Pty Ltd	Magnalla Grove Investment	PATRICIA NATIONAL PROPERTY OF THE PARTY OF T	Yearng Land Corporation (Cavill Pierk)	Young Land Corporation (C	Young Land Copporation (Casta Part	Young Lind Companies (Cavil Park	Young Land Corporation (Cavill Part)		LANKS BRILLIAN PF (Sushband Beesch)		Anjor Holdings.	Anpey Holdings	STATE OF THE PERSON NAMED IN	COLUMN SAND LITERATURE NAMED IN	Florid Pir Ltd		the state of the state of the state of	d part of the drawn me is con smontland below	Account Name	Bezzine Developers Ply Lic	Nadaghor Sayukry STPW	Redland Bay Lebsure Life	Reinac Pty Lid	St Crispin's Property Pty Ltd	Young Land Corporation (Eden Bradments Ptu I td	Lot 111 Plu Lid	Grentlanes Projects Ptv Ll	Glandoming Developments Pay Lid	CVSTPyCld	Cachigion Memorament Pry Lic	Transfer Land extra Line	Squees Developments No.1	Stures Student Lodge Pty Litt	Grada Square Prop Day. D	Tall Trees Tanah Marah	AIS PA.	Kingoonn Pri	Canno Erieles Lifertyle	U-Own Skings (Southban	Medicin Properties (Resort Corp.) Medicin Bross Investments Ptv Li	Young Land Corporation (Cavil) Park)	LIMINS OIL LIMINDE (Bunkland Bestel)	Anper Hakings	Devil Phy Ltd
	-	LALCT-DOT-GUARZ?	LA-01-001-AU-1027	LM:-01-DDTAU-4027	CHARLES AND	LM241-DOT-AU-1028	LMI-01-DDT-ALL-1028	LM-01-DDT-AU-1025	TEMPERATURE TRANSPORTED TO THE PROPERTY OF THE	LM-01-001-AU-1029	LM-d1-DDT-AtJ-tD29		TAILOR COLL-AU-1028	IM1-01-0057-AU-1028		LM-01-00T-AU-1030		LAK-01-DOT-AN-1031	1 UM-05-DDT-AU-5034	STREET, STATE OF STATE OF STATE OF STREET, STATE OF	LAKOT-DOT-AU-1035		The second second	3	When safets price is for multiple files, and it don't due a dual interest and under the modern of the safets of th																											
	HEATER CONTROL				STATE OF STA		• •	_	SOMETH OF SHARE STATES	-	-	-			化多氯甲基苯基苯基苯	-	ACTOR OF COMPANY AND ADDRESS.	1	-	ALTERNATION OF THE PROPERTY OF THE PERSON OF	1	CONTRACTOR OF THE PERSON OF TH	189	•	Where release																											
	abity Colonia	1270	1212	1273	PER PROPERTY.	1276	121	1278	1278		1281	1282	1202	1285	V-100513	1288	1287	1200	1285	district the	1290	to the same	102			•																										

Schedule 5 - New Lending Criteria

For clarity, the restrictions below only apply to release price allocations for new lending and not to lending from unrestricted cash.

Acceptable Loan Types:

- Subject: Real Property
- Property Type: Office, Retail, Apartment
- Specific prohibitions: land / construction loans, manufacturing
- · Seniority: Senior, sole first charge on subject property
- Syndications: None

Acceptable Loan Size:

- Minimum size: A\$10m
- Maximum size: A\$20m
- Loan-to-value: <=60% Loan-to-Value
- Loan-to-Income: <=6.5x last 12 mos net income
- Loan-to-Income: <=6.5x passing and stabilized net income
- DSCR: Passing and stabilized net income >=150%

3rd Party Reporting Standards:

- Appraisal, Engineering, Environmental
- Appraisal Firms: CBRE, Savills, Jones Lang, Colliers
- Engineering & Enviro Firms: >50 subject professionals

Acceptable Terms & Conditions:

- Minimum interest rate: 10%, paid monthly
- Maximum Term: 3 years
- Australian market standard documents, terms & conditions

Independent Nature:

- Must be 3rd party, arm's length in nature
- · Unrelated to existing borrowers of any LM associated party
- Unrelated to current or former LM associated parties
- Fees and expenses to LM associated parties must be approved

All receipts under the new loan must be applied to the settlement of outstanding deferred release prices utilized for new lending under this clause.

Financier has option to veto new loans within 5 business days of receiving an agreed upon FMIF lender pack (including key information on the underlying collateral, description of material terms of the loan, applicable 3rd party reports and certification of Independent Nature of the facility from LM).

Schedule 6 Financier Authorised Officers

George Wong

Beaux Pontak

Schedule 7 - Exceptions to LM Documentation Assumptions

Note: Headings are for convenience only and do not affect interpretation

LM Documentation Assumption (d)(i); clause 15.5(b)

Commonwealth Bank of Australia has a first ranking mortgage on the LM Mortgaged Property for the LM Loan known as Tall Trees.

LM Documentation Assumption (h); clause 15.5(d)

A solicitor's lien may apply where LM Loan Documents are held by solicitors. See schedule below for amounts owed to solicitors where a lien may apply.

LM Documentation Assumption (k); clause 15.5(f)

Certain LM Loan Documents are held by solicitors as follows:

Solicitors	LM Loan Documents	Amounts owed to solicitors
Allens Arthur Robinson, Brisbane	Bellpac	Approx \$180K
Norton Rose, Sydney	Green Square, Lot 111	Nominal amounts only
Holman Webb, Sydney	Beetil, Glendenning, Greenleaf, OVST, Anpor	Nominal amounts only

All of the above documents are held by the solicitors for the benefit of the Borrower.

LM Documentation Assumption (p)

The Borrower may be obliged to provide further financial accommodation on the following LM Loans:

- Tall Trees the Borrower considers it is not obliged to provide funding, however funding of two months interest plus operational costs is likely to be provided (approx \$50K/fortnight);
- Coulter the Borrower considers it is not obliged to provide funding, however funding of \$400K - \$450K is likely to be provided; and
- Trinity \$400K \$500K is required to be provided, less amount of bonds recovered from council.

Clause 15.5(i)

The Disclosed Litigation, relating to the LM Loan known as Bellpac, is disclosed for clause 15.5(I).

Clause 15.5(n)

The following assertion is disclosed for clause 15.5(n):

It was also resolved at that meeting that entering into the Finance Documents is in the Borrower's best interests and for its benefit. The benefit being obtained by the Borrower is [TBI].

The resolutions referred to in paragraphs (a) and (b) were duly passed at a properly convened and properly held meeting of duly appointed directors of the Borrower at which a quorum of directors entitled to vote was present and acting.

All provisions in the Corporations Act 2001 (Cth) and the constitution of the Borrower relating to the declaration of directors' interests and the powers of interested directors to vote were duly observed at or before the meeting.

4. Solvency

The Borrower is able to pay all its debts as and when they become due and payable and there are no grounds for suspecting that it will not continue to be able to do so after entering into the Finance Documents.

5. Related party transactions

Chapters 2E and 2J.3 of the Corporations Act 2001 (Cth) do not apply to the transactions contemplated by the Finance Documents or, if either applies, it has not been breached.

Dated:

·	
Signature of Director	Signature of Director
Title:	Title:

Schedule 9 - Trust Bank Accounts

				Number	Currency	Opening Balance 15/06/10	N
PTAL acf LM FMIF - Beetil	Premium Business Account	СВА	604000	12593774.00	AUD	- 693	Controller a/c
PTAL acf LM FMIF - Greenl	Premium Business Account	CBA	604000	12593782.00	AUD		Controller a/c
PTAL acf LM FMIF - Harves	Premium Business Account	CBA	604000	12593790.00	AUD	÷	Controller a/c
PTAL acf LM FMIF - Vitali	Premium Business Account	СВА	604000	12593803.00	AUD	-	Controller a/c
PTAL acf LM FMIF - YlandCav1	Premium Business Account	CBA	604000	12639517.00	AUD	\$3,371.66	Controller a/c
PTAL acf LM FMIF - YlandCav2	Premium Business Account	CBA	604000	12639533.00	AUD	- 69	Controller a/c
PTAL acf LM FMIF - Bridgewat	Premium Business Account	CBA	604000	12639568.00	AUD	\$281,003.98	Controller a/c
PTAL acf FMIF - Edens Controller A/C	Business Investment Acct	Suncorp	484799	03470475	AUD	\$1,865.66	Controller a/c
PTAL acf FMIF - Bezzina Controller A/C	Business Investment Acct	Suncorp	484799	034843355	AUD	\$64.12	Controller a/c
PTAL acf FMIF - Source Devel Controller A/C	Business Investment Acct	Suncorp	484799	034843398	AUD	\$65.14	Controller a/c
PTAL acf FMIF - U-Own Storage	Business Investment Acct	Suncorp	484799	034848802	AUD	\$10.10	Controller a/c
PTAL acf FMIF - Brambleton Controller A/C	Business Investment Acct	Suncorp	484799	034850572	AUD	\$113.12	Controller a/c
PTAL acf FMIF - Resort Controller Acc	Business Investment Acct	Suncorp	484799	034857909	AUD	\$877.79	Controller a/c
PTAL acf FMIF - Magnolia Controller	Business Investment Acct	Suncorp	484799	034857895	AUD	\$71.25	Controller a/c

Schedule 10 - Managed Cash Accounts

Bank	BSB	Account Number	Currency
SUNCORP	484799	49726640	AUD
SUNCORP	484799	34701216	AUD
CBA	602000	13564303	AUD
СВА	602000	13564346	AUD
СВА	N/A	100642778CAD115601	CAD
СВА	N/A	100642778EUR115601	EUR
СВА	N/A	100642778GBP115601	GBP
СВА	N/A	100642778HKD115601	HKD
CBA	N/A	100642778SGD115601	SGD
СВА	N/A	100642778THB115601	тнв
СВА	N/A	100642778USD115601	USD
	SUNCORP SUNCORP CBA CBA CBA CBA CBA	SUNCORP 484799 SUNCORP 484799 CBA 602000 CBA 602000 CBA N/A CBA N/A CBA N/A CBA N/A CBA N/A CBA N/A	SUNCORP 484799 49726640 SUNCORP 484799 34701216 CBA 602000 13564303 CBA 602000 13564346 CBA N/A 100642778CAD115601 CBA N/A 100642778EUR115601 CBA N/A 100642778HKD115601 CBA N/A 100642778SGD115601 CBA N/A 100642778THB115601

Signed as an agreement.

Borrower

Executed by LM Investment Management Limited ABN 68 077 208 461 in its capacity as Responsible Entity for the LM First Mortgage Income Fund in accordance with section 127 of the Corporations Act 2001 (Cth):

Signature of company secretary/director

Financier

Signed for and on behalf of Deutsche Bank AG, Sydney Branch by its attorneys under a sub-power of attorney dated 31 July 2009 registered in NSW Book No. 4577 No. 122 in the presence of:

Signature of attorney who declares that the attorney has not received any notice of the revocation of the power of attorney

Full name of attorney

Signature of witness

Signature of attorney who declares that the attorney has not received any notice of the revocation of the power of attorney

Full name of witness

Full name of attorney

EXECUTION VERSION

Override Deed

LM investment Management Limited in its capacity as responsible entity for the LM First Mortgage Income Fund

Borrower

Deutsche Bank AG, Sydney Branch Financier

Clayton Utz Lawyers Level 18 333 Collins Street Melbourne VIC 3000 Australia DX 38451 333 Collins VIC T + 61 3 9286 6000 F + 61 3 9629 8488

www.claytonutz.com

Our reference 17306/18128/80107994

Contents

1.	Defini	itions and Interpretation	
	1.1 1.2 1.3 1.4	Definitions	4 4
2.		itions	
	2.1 2.2	Conditions Precedent	4
3.	Ackno	owledgment of indebtedness	5
4.	Overr	ide of Facility	5
	4.1 4.2 4.3 4.4 4.5 4.6	Existing Facility Agreement	
5.	Defau	lk ,2000.000.000.000.000.000.000.000.0000.0000	
6.	Furthe	er Assurances	
7.		g,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
8.	Gener	rat confirmations and ratifications	8
	8.1 8.2	Continuing obligationsPreservation of existing rights	8
9.	Misce	llansous	,,,,, <u>,</u>
	9.1 9.2	Governing LawCounterparts	9 9
Sched	ule 1 - Ve	erification Certificate	10
School	ufa 2 _ 55.	anagad Cach Accounts	19

Override Deed made at

on 21 December 2012

Parties

LM Investment Management Limited ACN 077 208 461 in its capacity as Responsible Entity for the LM First Mortgage Income Fund (Romaner)

Deutsche Bank AG, Sydney Branch (Financier)

Recitals

- A. The Facility has been extended and currently expires on 30 June 2013.
- B. The Borrower has requested that the Financier further extend the Facility to the Override Facility Termination Date.
- C. The Parties have agreed to enter into this deed in order to, inter alia, extend the Facility to the Override Facility Termination Date on the terms set out in this deed.

Operative Provisions

1. Definitions and Interpretation

1.1 Definitions

In this deed:

Capital Works shall have the meaning as defined in clause 4.2(0).

Capital Works Account means the account listed as the "Capital Works Account" in Schedule 2 maintained by PTAL.

Capital Works Plan shall have the meaning as defined in clause 4.2(o).

Conditions Precedent means those conditions precedent set out at clause 2.1.

Current Amount shall have the meaning as defined in clause 3(a).

Debt means the total of all of the amounts for which the Borrower is liable to the Financier from time to time pursuant to the Facility Agreement, together with all accrued interest, fees and costs and other amounts for which the Borrower is liable to the Financier on any account whatsoever.

Default Amount has the meaning as described in clause 4.5(b).

Effective Date means the later of the date of this deed and the date by which all of the Conditions Precedent are satisfied.

Extended Term means the period beginning on 1 July 2012 and ending on the Override Facility Termination Date.

Facility Agreement means the facility agreement dated 1 July 2010 between the Borrower and the Financier as amended from time to time including pursuant to the letter from the Financier to the Borrower dated 21 September 2010.

Full Margin means the total amount of all interest to be paid by the Borrower to the Financier pursuant to the Facility Agreement at the relevant interest rate as determined by the Financier in accordance with the Facility Agreement during the Extended Term.

Fund means the LM First Mortgage Income Fund ABN 13 089 343 288.

Fund Security means the Deed of Charge insofar as it relates to the Fund.

Initial Amortisation Payment means such amount as is necessary to bring the Debt to an amount not exceeding \$28,000,000.00 to be paid by the Borrower to the Financier as a Condition Precedent pursuant to clause 2.1(e).

Initial Extended Term shall have the meaning as defined in clause 4.4(b).

Option Term means, provided that the option is exercised pursuant to Clause 4.3, the period beginning on 1 July 2014 and ending on 30 June 2015.

Override Facility Termination Date means, subject to Clause 4.3, 30 June 2014.

Party means a party to this deed.

PPSA means the Personal Property Securities Act 2009 (Cth).

PPSR means the Personal Property Securities Register established under the PPSA.

Preparation Fee means \$25,477.00 plus GST being the agreed legal costs of the Financier in respect of the preparation, negotiation and execution of this Deed.

PTAL means The Trust Company (PTAL) Limited ACN 008 412 913 formerly named Permanent Trustee Australia Limited as trustee for the LM First Mortgage Income Fund ABN 13 089 343 288.

Restricted Accounts means those accounts listed as "Restricted Accounts" in Schedule 2 maintained by PTAL and such other accounts as are approved by the Financier from time to time excluding any such accounts as are permanently closed by PTAL including any Unrestricted Accounts that are deemed to be Restricted Accounts pursuant to the defined term "Unrestricted Accounts".

Restructuring Fee means an amount equal to 1% of the Current Amount.

Revoked Qualifying Capital Improvements shall have the meaning as defined in clause 4.2(f).

Scheduled Amortisation Payment means the minimum monthly payment of \$500,000.00 to be paid by the Borrower to the Financier at the times and in the manner set out in clause 4.2(c).

Side Letter means the capital works account side letter from the Financier to the Borrower dated 24 June 2011.

Suncorp means Suncorp-Metway Ltd ABN 66 010 831 722.

Unrestricted Accounts means those accounts listed as "Unrestricted Accounts" in Schedule 2 maintained by PTAL and such other accounts as are approved by the Financier from time to time excluding any such accounts as are permanently closed by PTAL unless and until an Event of Default or Potential Event of Default has occurred and is subsisting whereupon the Unrestricted Accounts will be deemed to be Restricted Accounts.

Verification Certificate means a certificate substantially in the form of Schedule 1.

Voluntary Prepayment means any prepayment made by the Borrower to the Financier after the Effective Date other than a payment made:

- (a) pursuant to clause 4.2(g); or
- (b) during such time as an Event of Default or Potential Event of Default has occurred and is subsisting.

Voluntary Prepayment Amount means, in respect of any Voluntary Prepayment, the amount of that Voluntary Prepayment.

1.2 Defined terms

- (a) Unless otherwise defined in this deed, expressions used in this deed have the meanings given to them in or for the purposes of the Facility Agreement.
- (b) For the avoidance of doubt the Parties agree that terms defined in this deed are to apply to this deed only.

1.3 Finance Document

This deed is a Finance Document for the purposes of the Facility Agreement.

1.4 Interpretation

In this document:

- (a) a reference to a Party includes that Party's executors, administrators, successors and permitted assigns, including persons taking by way of novation;
- (b) a word importing the singular includes the plural (and vice versa), and a word indicating a gender includes every other gender;
- (c) where the day on which or by which any act, matter or thing is to be done under this document is not a Business Day, that act, matter or thing will be done on the immediately preceding Business Day;
- (d) "Includes" in any form is not a word of limitation; and
- (e) a reference to "5" or "dollar" is to Australian currency.

2. Conditions

2.1 Conditions Precedent

The override of the Facility Agreement as contemplated by this deed is of no force and effect until the following conditions precedent are satisfied:

- (a) (Registration): the Fund Security has been registered on the PPSR;
- (b) (Verification Certificate): the Financier has received a Verification Certificate from the Botrower;
- (c) (Restructuring Fee): the Restructuring Fee has been paid by the Borrower to the Financier;
- (d) (Preparation Fee): the Preparation Fee has been paid by the Borrower to the Financier; and
- (e) (Initial Amortisation Payment) the Initial Amortisation Payment has been made in full.

2.2 Waiver of conditions precedent

(a) The Financier may waive any one or more of the Conditions Precedent or determine that any such condition has been satisfied.

(b) Any waiver of a Condition Precedent by the Financier under clause 2.2(a) is subject to such conditions as the Financier determines, and such conditions may include making a Condition Precedent a condition subsequent to be satisfied by a later date.

3. Acknowledgment of Indebtedness

The Borrower agrees and acknowledges by executing this deed that:

- (a) it is indebted to the Financier for the full amount of the Debt, which, as at 18 December 2012, was \$29,409,489.00 (Current Amount);
- (b) it has no dispute, counterclaim, set-off or any other claim which would otherwise reduce the Current Amount of the Debt or negatively affect or impact upon its enforceability in any way:
- (c) the Security is and continues to be valid and fully enforceable by the Financier in accordance with its terms; and
- (d) all amounts payable by the Borrower under or pursuant to this deed shall form part of the secured obligations secured by the Security.

4. Override of Facility

4.1 Existing Facility Agreement

On and from the Effective Date, the Parties agree and acknowledge that:

- (a) the Facility Agreement is, to the extent necessary, overridden and deemed varied, so as to be read consistently with this deed; and
- (b) to the extent of any inconsistency between this deed and the Facility Agreement, this deed prevails.

4.2 Override of Facility Agreement

On and from the Effective Date, the Parties agree and acknowledge that:

- (a) the Facility shall be extended to, and expire on, the Override Facility Termination Date;
- (b) no further financial accommodation (including by way of redraw) shall be provided by the Financier to the Borrower;
- (c) the Borrower shall make six Scheduled Amortisation Payments on a monthly basis on the Monthly Interest Payment Date following the Initial Amortisation Payment;
- (d) clause 6.2(a)(v) of the Facility Agreement shall be deemed to be deleted and shall have no further force and effect;
- (e) the following wording shall be added to the start of clause 6.2(b) of the Facility Agreement: "Unless otherwise agreed by the Financier in writing,";
- (f) the Financier may at any time, and in its absolute discretion, revoke its approval with respect to any one or more Qualifying Capital Improvements that have previously been approved by the Financier pursuant to the Side Letter (Revoked Qualifying Capital Improvements), by notice to the Borrower and, without limiting clause 4.2(h), require all or any credit balances in the Capital Works Account that relate to the Revoked Qualifying Capital Improvements to be

immediately transferred to the relevant Restricted Account. However, nothing in this clause 4.2(f) will:

- (i) require the Borrower to reimburse the Capital Works Account for any amounts previously agreed to by the Financier and applied by the Borrower for the purpose of any Qualifying Capital Improvements; or
- (ii) limit the Financier's rights provided for in the Side Letter, including the right of the Financier to terminate the Capital Works Account in its absolute discretion, pursuant to the Side Letter,
- (g) provided no Byent of Default or Potential Event of Default has occurred and is subsisting, the Financier shall be authorised at all times and in its absolute discretion to apply all or any part of any credit balance of the Restricted Accounts (but not including the Capital Works Account) towards the reduction of the Debt;
- (h) if an Event of Default or Potential Event of Default has occurred and is subsisting, the Financier shall be authorised at all times and in its absolute discretion to apply all or any part of any credit balance of the Managed Cash Accounts towards the reduction of the Debt;
- (i) despite any Voluntary Prepayment, the Borrower will remain liable to pay to the Financier the Full Margin in respect of the Voluntary Prepayment Amount:
 - (i) at the time and in the manner described in the Facility Agreement and on the basis that the Voluntary Prepaid Amount had been drawn for the duration of the Extended Term; or
 - (ii) where all amounts outstanding under the Facility are prepaid prior to the expiry of the Extended Term, on the date that all amounts (other than the Full Margin) under the Facilities are discharged and satisfied in full;
- (j) any reference to \$6,000,000, with respect to the Minimum Cash Balance, in the definition of "Minimum Cash Balance" and in clause 16.11(h) of the Facility Agreement, shall be deemed to be a reference to \$3,000,000.00;
- (k) if, in any month, the Minimum Cash Balance falls below \$3,000,000.00, the Financier shall be authorised, in its absolute discretion, to apply all or any part of any credit balance of the Unrestricted Accounts towards recapitalising the Restricted Accounts;
- (1) notwithstanding the Side Letter, the Capital Works Account must be capitalised with a minimum balance of \$2,000,000.00 on the Effective Date and increased to \$3,000,000.00 by way of equal six equal monthly instalments of \$166,667.00 commencing on the first Interest Payment Date following the Effective Date and thereafter to be made monthly on each Monthly Interest Payment Date;
- (m) without limiting the Borrower's obligations under clause 4.2(1), to the extent that any funds standing to the credit of the Capital Works Account are applied by the Financier in accordance with clause 4.2(h), the Borrower must ensure that the amount applied is reinbursed to the Capital Works Account within 60 days of such application; and
- (n) any repayment of any amount under any provision of this deed will be made together with interest and fees accrued on the amount repaid or prepaid and any amount required to be paid in accordance with the Facility Agreement.
- (o) notwithstanding the Side Letter, the Borrower will provide the Financier with a capital works plan (the Capital Works Plan and the proposed works being the

LegaR308959550.10

Capital Works that are referred to in the Side Letter) in respect of each project to which the Capital Works Plan applies setting out in full detail the proposed utilisation of the funds that are to be applied in respect of the Capital Works in sufficient detail necessary to allow the Financier to determine whether to approve the relevant Capital Works.

4.3 Option to Extend Override Facility Termination Date

Provided no Event of Default or Potential Event of Default has occurred and is subsisting, the Botrower will be entitled, upon written notice to the Financier given not later than 120 days and not before 180 days prior to 30 June 2014, require the Financier to agree to an extension of the Override Facility Termination Date to 30 June 2015.

4.4 Interest Rate payable during the Initial Extended Term

- (a) Subject to clause 4.4(b) and clause 4.6, interest during the Initial Extended Term will be payable at the rate of 15% per annum.
- (b) If an Event of Default has occurred and is subsisting during the Extended Term but before 30 June 2014 (Initial Extended Term), in addition to the obligation of the Borrower to pay interest for the period of the Initial Extended Term as described in clause 4.4(a), additional interest calculated at an amount equal to the difference between the amount of interest payable by the Borrower under the Facility during the Initial Extended Term as described in clause 4.4(a), and the amount payable by the Borrower during the Initial Extended Term from the date of the Event of Default to:
 - (i) the date that the Event of Default is remedied to the satisfaction of the Financier; or
 - (ii) if the Event of Default is not remedied to the satisfaction of the Financier, 30 June 2014,

at the Default Margin will accrue pursuant to the Facility and shall be immediately due and payable.

4.5 Interest rate payable during the Option Term

- (a) Subject to clause 4.5(d) and clause 4.6, interest during the Option Term will be payable at the rate of 15% per annum.
- (b) The Borrower acknowledges that an amount equal to the difference between the amount of interest payable by the Borrower under the Facility during the Option Term at the relevant interest rate and the amount payable by the Borrower during the Option Term at the Default Margin will accrue pursuant to the Facility (Default Amount).
- (c) Provided there is no Event of Default during the Option Term, and the Facility is repaid in full or refinanced on or prior to the 30 June 2015, the Default Amount will not be payable by the Borrower at any time.
- (d) However, if an Event of Default has occurred and is subsisting during the Option Term, in addition to the obligation of the Borrower to pay interest for the period of the Option Term as described in clause 4.5(a), additional interest calculated at an amount equal to the difference between the amount of interest payable by the Borrower under the Pacility during the Option Term as described in clause 4.5(a), and the amount payable by the Borrower during the Option term from the date of the Event of Default to:

- the date that the Event of Default is remedied to the satisfaction of the Financier; or
- if the Event of Default is not remedied to the satisfaction of the Financier, 30 June 2015,

at the Default Margin (being part or all of the Default Amount) will accrue pursuant to the Facility and shall be immediately due and payable.

4.6 Default Margin

Nothing in this clause 4 shall prohibit the Financier from applying a Default Margin of 5% to any outstanding Debt pursuant to the Facility in circumstances where the Borrower has been in default of any payment obligations under the Facility or this deed for six consecutive months. If applicable, pursuant to the Facility, the increased rate shall apply from the Interest Period commencing immediately after such 6th month.

5. Default

The Financier may terminate and demand repayment of the Pacifity (including any amounts payable pursuant to clause 4.4(b) and (provided that the Borrower has exercised the option pursuant to clause 4.3), any amounts payable pursuant to clause 4.5(d) immediately upon (or at any time after) an Event of Default in any Finance Document has occurred.

6. Further Assurances

The Borrower will, on demand by the Financier and at the cost of the Borrower, perform or procure the performance of all such acts and execute all such agreements, assurances and other documents and instruments as the Financier requires to document and/or perfect the rights, arrangements and powers contemplated by this deed.

7. Timing

Time is of the essence with respect to the performance of the Borrower's obligations under this deed

8. General confirmations and ratifications

8.1 Continuing obligations

The Borrower unconditionally and irrevocably:

- (a) ratifies and confirms to the Financier its continuing obligations under the Facility
 Agreement (as amended by this deed) and the other Finance Documents; and
- (b) acknowledges and agrees that except to the extent of the variations effected by the provisions of this deed, the Facility Agreement and Finance Documents shall in all respects continue to apply.

8.2 Preservation of existing rights

Except to the extent expressly contemplated by this deed, this deed does not in any way limit, reduce or affect any right, entitlement or remedy of the Financier under the Finance Documents (or any of them) arising prior to, or after, the Effective Date which rights, entitlements and remedies are hereby expressly reserved.

9. Miscellaneous

9.1 Governing Law

This deed is governed by and will be construed according to the laws of Queensland.

9.2 Counterparts

This document may be executed in any number of counterparts, all of which taken together will be deemed to constitute one and the same document.

Schedule 1 - Verification Certificate

To: Deutsche Bank AG, Sydney Branch

Override deed (the "deed")

Capitalised terms used in this Certificate and not otherwise defined have the same meaning as that given to them in the deed.

I, Eghard Van Der Hoven of Level 4, 9 Beach Road, Surfers Paradise in the State of Queensland hereby certify that:

Director

 I am a director of LM Investment Management Limited (the "Company") and am duly authorised to sign this Certificate on behalf of the Company;

Constitution

 the copy of the Company's Constitution previously provided to the Financier remains a true, complete and up to date copy of the Company's Constitution;

Resolutions

- attached hereto as Annexure B is a true copy of an extract of the minutes of a meeting of the Board of Directors of the Company duly called, convened and held on 20th December 2012 resolving, among other things, to enter into and execute the deed.
- 4. the resolutions of the Company set out in the minutes marked as Annexure B were duly passed at a properly convened meeting of duly appointed Directors of the Company at which a quorum was present and acting throughout;
- all provisions contained in the Corporations Act and any other applicable equivalent law, and the Constitution of the Company, relating the declaration of directors' interests and the powers of interested directors to vote at that meeting were duly observed;
- 6. the resolutions set out in those minutes have not been amended, modified or rescinded and remain in full force and effect;

Date: this 20 day of December 2012

Signature

Name

Director of Company

Annexure A Constitution [if applicable]	
Annexure B Extract of Minutes of Meeting	

LegoR308959550.10

I)

RESOLUTION OF THE BOARD OF DIRECTORS

LM Investment Management Ltd ABN 68 077 208 461

Dated: 20/12/2012

The following resolution has been proposed:

To agree a restructure and extension of the existing LM FMIF debt facility with Deutsche Bank AG

The LM First Mortgage Income Fund has an existing finance facility with Deutsche Bank AG (DB). The term of the current facility expired on 30th June 2013.

Negotiations with DB have been finalised to provide for an extension of the facility until 30th June 2014. At that time LM has an exclusive option to extend the loan term by a further 12 months to 30th June 2015.

Re-documentation has been achieved by way of an Override Deed revising the terms of the original agreement where required. Whereas certain commercial terms of the agreement have been revised the overall intent of the original agreement has been retained.

In summary the material revised terms are as follows:-

- Interest rate to remain at 15% (+3% Default rate for first 6 months of subsisting default and 5% thereafter)) with a 1% Re-structuring Fee.
- Should LM exercise the option to extend the loan facility by 12 months then a 15% interest rate remains payable. However should an event of default occur during the 12 month extended term the higher rate (18%) will be deemed due and payable.
- The current requirement for facility amortisation to occur in a pre-determined manner linked to asset sales ("Release Price" mechanism) will become redundant.
- The facility will be reduced from its current balance of c. \$29.4m to \$25m by the end of May 2014 with a first reduction to \$28m as a condition precedent and subsequent minimum monthly repayments of \$500k. From that time there will be no fixed repayment requirements.
- Any balances repaid by LM to DB prior to 30th June 2014 will require a "make good" payment
 equivalent to the interest applicable to those amounts during the term. This will naturally exclude
 the payment noted above and also any "Sweep" by DB as noted below.
- DB will be entitled to "Sweep" any cash proceeds created during each month's sales activity at
 their discretion up to the balance of the Restricted Cash Accounts. Cash from previous
 settlements achieved in prior months and transferred to Unrestricted Cash or Capital Works
 Account will not be able to be swept.
 - (In discussion with DB we are advised this is not likely to be actively applied unless in circumstances where for example the balance of security within the Fund changes significantly during the term of the facility).
- The Fund's minimum cash balance will be reduced to \$3m. Any balances above \$3m in restricted cash accounts following each month end may be transferred to unrestricted cash or to Operational Works Account at LM's discretion and upon instruction to DB. The Operational Works account will have an initial minimum cash balance of \$2m increasing progressively over a 6 month period to \$3m as development activity increases. Both accounts will have replenishment requirements should the balances drop below agreed minimums.
- Works to be undertaken and funded from the Capital Works Account will be treated as per the current terms of the loan agreement relating to such works.
- LM intends to continue with the strategy that all assets are sold at or above their nominated Release Prices as defined in the original loan agreement where possible.

THE BOARD RESOLVED to accept the terms of the revised loan agreement as it was determined to be in the best interests of scheme members as it provides for both an acceleration of return of capital to investors and additionally the ability to accelerate further capital works expenditure on those Fund assets identified for further works prior to sale and help achieve the optimum return of capital to investors

Each of the Directors accepts the resolution as signed below.

We, the undersigned, being the Directors of LM Investment Management Ltd, state that we are in favour of the resolution.

Peter Drake

Date:

Katy Phillips Date:

RESOLUTION OF THE BOARD OF DIRECTORS

Lill investment Management Ltd ABN 68 077 208 461

Dated: 20/12/2012

The following resolution has been proposed:

To agree a restructure and extension of the existing LM FMF debt facility with Deutsche Bank AG

The LM First Mortgage Income Fund has an existing finance facility with Deutsche Bank AG (DB). The term of the current facility expired on 30th June 2013.

Negotiations with DB have been finalised to provide for an extension of the facility until 30th June 2014. At that time LM has an exclusive option to extend the loan term by a further 12 months to 30th June 2015.

Re-documentation has been achieved by way of an Override Dead revising the terms of the original agreement where required. Whereas certain commercial terms of the agreement have been revised the overall intent of the original agreement has been retained.

In summary the material revised terms are as follows:-

- Interest rate to remain at 15% (+3% Default rate for first 6 months of subsisting default and 5% thereafter)) with a 1% Re-structuring Fee.
- Should LM exercise the option to extend the loan facility by 12 months then a 15% interest rate
 remains payable. However should an event of default occur during the 12 month extended term
 the higher rate (18%) will be deemed due and payable.
- The current requirement for facility amortisation to occur in a pre-determined manner linked to asset sales ("Release Price" mechanism) will become redundant.
- The facility will be reduced from its current balance of o. \$29.4m to \$25m by the end of May 2014
 with a first reduction to \$28m as a condition precedent and subsequent minimum monthly
 repayments of \$500k. From that time there will be no fixed repayment requirements.
- Any balances repaid by LM to DB prior to 30th June 2014 will require a "make good" payment
 equivalent to the interest applicable to those amounts during the term. This will naturally exclude
 the payment noted above and also any "Sweep" by DB as noted below.
- DB will be entitled to "Sweep" any cash proceeds created during each month's sales activity at
 their discretion up to the balance of the Restricted Cash Accounts. Cash from previous
 settlements achieved in prior months and transferred to Unrestricted Cash or Capital Works
 Account will not be able to be swept.
 - (In discussion with DB we are advised this is not likely to be actively applied unless in circumstances where for example the balance of security within the Fund changes significantly during the term of the facility).
- The Fund's minimum cash balance will be reduced to \$3m. Any balances above \$3m in restricted cash accounts following each month end may be transferred to unrestricted cash or to Operational Works Account at LM's discretion and upon instruction to DB. The Operational Works account will have an initial minimum cash balance of \$2m Increasing progressively over a 6 month period to \$3m as development activity increases. Both accounts will have replenishment requirements should the balances drop below agreed minimums.
- Works to be undertaken and funded from the Capital Works Account will be treated as per the current terms of the loan agreement relating to such works.
- LM intends to continue with the strategy that all assets are sold at or above their nominated Release Prices as defined in the original loan agreement where possible.

THE BOARD RESOLVED to accept the terms of the revised loan agreement as it was determined to be in the best interests of scheme members as it provides for both an acceleration of return of capital to investors and additionally the ability to accelerate further capital works expenditure on those Fund assets identified for further works prior to sale and help achieve the optimum return of capital to investors

Each of the Directors accepts the resolution as signed below.

We, the undersigned, being the Directors of LM Investment Management Ltd, state that we are in favour of the resolution.

Peter Drake	Eghard van der Hoven
Date:	Date:
Francene Mulder Date:	Katy Phillips 12/12

Schedule 2 - Managed Cash Accounts

Account Name	Bank	BSB	Account Number
Restricted Accounts	<u>.</u>		
PTAL - Restricted Working Account	Suncorp	484 799	451081276
PTAL - Restricted Deposit Account	Suncorp	484 799	451081420
PTAL - Capital Works Account	Suncorp	484 799	451085980
Unrestricted Accounts			
PTAL - FMIM Deposit Account Unrestricted	Suncorp	484 799	602037866
PTAL - FMIM Working Account Unrestricted	Suncorp	484 799	049726640
Other Accounts	<u> </u>	<u>.L</u>	
PTAL - New Zealand Distribution Account	Suncorp	484 799	600810410
PTAL - Working Account (NZ dollar account)	Australia and New Zealand Banking Group	408 547	01-0102-0856678-000

Execution Page

Executed as a deed

Each of the attorneys executing this deed states that the attorney has no notice of the revocation of the power of attorney appointing them.

Borrower

Executed by LM Investment Management Limited ABN 68 077 208 461 in its capacity as Responsible Entity for the LM First Mortgage Income Fund in accordance with section 127 of the Corporations Act 2001 (Cth);

Signature of director

Full name of director

Full name of company secretary/director

Financier

Signed for and on behalf of Deutsche Bank AG, Sydney Branch by its attorneys under a sub-power of attorney dated 31 July 2009 registered in NSW Book No. 4577 No. 122 in the presence of:

Signature of attorney who declares that the attorney has not received any notice of the revocation of the power of attorney

DIME BAPOSIO Full name of attorney

Signature of witness

Signature of attorney who declares that the attorney has not received any notice of the revocation of the power of attorney

Mother Manual

Full name of witness

DAVID INVING

Fuil name of attorney